THE RELATIONSHIP OF NONCOGNITIVE VARIABLES AND THEIR CONTRIBUTION TO ATTRITION AMONG HEALTH CARE SPECIALISTS AT FORT SAM HOUSTON, TX

A Dissertation

by

YVETTE WOODS

Submitted to the Office of Graduate Studies of Texas A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

August 2007

Major Subject: Educational Human Resource Development

THE RELATIONSHIP OF NONCOGNITIVE VARIABLES AND THEIR CONTRIBUTION TO ATTRITION AMONG HEALTH CARE SPECIALISTS AT FORT SAM HOUSTON, TX

A Dissertation

by

YVETTE WOODS

Submitted to the Office of Graduate Studies of Texas A&M University in partial fulfillment of the requirements for the degree of

DOCTOR OF PHILOSOPHY

Approved by:

Chair of Committee,	M. Carolyn Clark
Committee Members,	Jennifer A. Sandlin
	Toby M. Egan
	Michael J. Ash
	Rebecca Hooper
Head of Department,	Jim Scheurich

August 2007

Major Subject: Educational Human Resource Development

ABSTRACT

The Relationship of Noncognitive Variables and Their Contribution to Attrition among Health Care Specialists at Fort Sam Houston, TX. (August 2007) Yvette Woods, B.S., Chicago State University;

M. S., Central Michigan University

Chair of Advisory Committee: Dr. M. Carolyn Clark

The Health Care Specialist Course trains Active Army, Army Reserves, Army National Guard and various international students in basic medical care, culminating in the possession of the EMT-B certification. The course is conducted in a stressful environment where students are required to be successful in both academic and nonacademic domains. Within the last decade, course administrators have noticed a higher rate of attrition and requested assistance with understanding why one-fifth of students fail to graduate with their original unit. A high rate of attrition results in an increased use of resources and it decreases the Army's ability to provide qualified Health Care Specialists to forward units.

The purpose of this study was to understand how noncognitive factors contribute to attirition in the Health Care Specialist Program with students who were within their first six months of training. This study specifically focuses on the experiences of the recyled student. The Modified Noncognitive Questionnarie (NCQ) and the Military Environment Noncognitive Adjustment Scale (MENAS), which focused on measuring noncognitive variables, were used with both passing and recycled students. In addition, an interview was used for recycled students to allow them the opportunity to elaborate on their personal experiences.

This mixed methods explanatory research study revealed quantitatively, using the t-test, that a significant difference exists between the passing and recycled groups in their: level of motivation, realistic self-appraisal, battle buddy support, unit support, preference for long-term goals, ability to successfully handle racism, and their level of stress. Logistic regression revealed the following to be predictive of attrition for students participating in this course: low ST score, unrealistic self-appraisal, preference for short-term goals, low perception of battle buddy support and unit support, a high level of stress and low motivation to complete the course. Qualitative results were consistent with quantitative results and added a deeper understanding of how students negotiated the academic and military environment. The results of this study will contribute to course administrators understanding of the challenges that student's encounter while matriculating through this course.

ACKNOWLEDGEMENTS

The completion of this chapter in my life would not have been as meaningful without the valuable support I received from family and friends throughout this process. First and foremost, I thank God for giving me the energy and motivation to remain dedicated to successfully complete this degree.

Thank you to my husband, Kevin for your unwavering understanding and support during the past three years. Thanks to my children, Stephanie and Michael for the many sacrifices endured when I could not physically attend family functions. Although you may not have always understood, I am extremely grateful for your patience.

A very special thank you goes to the US Army and leaders in the Army Medical Specialist Corps for having the confidence in my abilities, and for allowing me the opportunity to excel in yet another endeavor.

A special appreciation goes to my committee members: Dr. M. Carolyn Clark, Dr. Jennifer Sandlin, Dr. Toby Egan, Dr. Michael Ash and Dr. Rebecca Hooper for the guidance, support, and encouragement they provided throughout this journey.

Finally, thank you to the participants in this study, for without their valuable feedback, this research would not have been possible.

TABLE OF CONTENTS

81

ABSTRACT	۲ 	iii
ACKNOWL	EDGEMENTS	v
TABLE OF	CONTENTS	vi
LIST OF TA	ABLES	viii
LIST OF FIG	GURES	X
CHAPTER		
Ι	INTRODUCTION	1
	Health Care Specialists Matriculation Process Statement of the Problem	2 22
	Research Questions	24 24
	Definitions	24
	Assumptions	27
	Limitations	27
	Significance of the Study	27
Π	METHODOLOGY	29
	Research Design	30
	Sample	31
	Data Collection	35
	Data Analysis	42
III	COURSE STRUCTURE	61
	Pedagogy	62
	Military Environment	74
	Discussion	79

Conclusion.....

CHAPT	ER		Page
]	IV	MOTIVATION, EXPECTATIONS AND	0.4
		SELF-APPRAISAL	84
		Method	87
		Results	90
		Discussion	103
•	V	PERSONAL RELATIONSHIPS	104
		Influence of Relationships	105
		Method	112
		Results	114
		Discussion	123
		Conclusion	125
	VI	SUMMARY AND DISCUSSION	126
		Findings	127
		Discussion of Findings	143
		Implications and Recommendations	151
		Conclusions	156
REFER	ENCES	5	159
APPEN	DIX A		169
APPEN	DIX B		171
APPEN	DIX C		188
APPEN	DIX D		191
APPEN	DIX E.		194
APPEN	DIX F.		195
VITA			215

LIST OF TABLES

TABLE		Page
2.1	Gender and Ethnicity	33
2.2	Highest Level of Education	34
2.3	Age and Aptitude Scores	34
2.4	Modified NCQ Total Variance Explained	47
2.5	Modified NCQ Rotated Factor Matrix	48
2.6	Total Variance Explained	51
2.7	Rotated Factor Matrix	53
2.8	MENAS Total Variance Explained	54
2.9	MENAS Rotated Factor Matrix	55
4.1	Motivation Items, Means, Standard Deviations, t Values and Significance Levels	91
4.2	Self-Appraisal Items, Means, Standard Deviations, t Values and Significance Levels	92
4.3	Expectation Items, Means, Standard Deviations, t Values and Significance Levels	92
5.1	Battle Buddy Support Items, Means, Standard Deviations, t Values and Significance Levels	115
5.2	Family Support Items, Means, Standard Deviations, t Values and Significance Levels	116
5.3	Unit Support Items, Means, Standard Deviations, t Values and Significance Levels	116
6.1	Modified NCQ Descriptive Statistics	128

TABLE		Page
6.2	MENAS Descriptive Statistics	128
6.3	Modified NCQ Items, Means, Standard Deviations, t Values and Significance Levels	130
6.4	MENAS Items, Means, Standard Deviations, t Values and Significance Levels	131
6.5	Modified NCQ Logistic Regression Variables in the Equation	134
6.6	Modified NCQ Logistic Regression	134
6.7	MENAS Logistic Regression Variables in the Equation	134
6.8	MENAS Logistic Regression	136
6.9	Cognitive Variables in the Equation	137
6.10	Cognitive Logistic Regression	137

ix

LIST OF FIGURES

FIGURE		Page
2.1	Schematic showing the levels of approval required for the data collection process	41
2.2	Qualitative themes showing the interaction between the student, the course structure and noncognitive factors	60
6.1	Health Care Specialist Attrition Model demonstrating student's negotiation of the course utilizing both cognitive and noncognitive factors which influences their attrition decision	n 150

CHAPTER I

INTRODUCTION

The relationship of noncognitive factors and attrition in the military setting has not been well publicized. For many years researchers have focused specifically on the impact of cognitive factors upon academic performance. The results from their research have contributed to the creation of standardized testing. Recently, noncognitive factors have been deemed important contributors to attrition and have begun to receive more attention. While this is true at the collegiate level, few studies have been specifically geared toward studying its impact on attrition with first-term military recruits.

The order of presentation within this dissertation is divided into six chapters. Each chapter addresses specific noncognitive factors and discusses how they interact to explain attrition in this environment. Chapter I provides a brief background of the process of transitioning to the Health Care Specialist course and some of the challenges students must encounter during the course. Transforming from civilian to soldier requires individuals to utilize the strengths of both their cognitive and noncognitive skills in order to successfully navigate through the course. Chapter II discusses the methodology used throughout this study. Chapter III focuses specifically on the qualitative responses provided by recycled students and discusses the impact of the course structure on their attrition. Chapter IV addresses how motivation, expectations and self-appraisal explain attrition, whereas Chapter V explains the impact of personal

This dissertation follows the style of Research in Higher Education.

relationships on performance. Chapter VI provides a summary and discussion of the findings as well as recommendations and implications that administrators can use to improve the course.

Trainee, student, and soldier is used interchangeably in this study as they all refer to a soldier with less than six months of military training and medic will be used interchangeably with Health Care Specialist.

Health Care Specialists Matriculation Process

Recruiting

The first major step toward the transformation is the recruitment process. Once applicants demonstrate an interest in the military they first contact a local recruiter. Upon contacting all of the services that they are interested in, applicants determine which Service provides the best incentive package geared toward their needs. The recruiter's job is to promote the Army by describing all benefits and incentives packages available. Once applicants decide they want to join, the recruiter conducts a background check to verify educational history and ensure there is no criminal history. Once their background is found to be clear, they are scheduled to take the Armed Services Vocational Aptitude Battery (ASVAB) to determine in which jobs they would likely perform best. The results provide a list of jobs for applicants to choose from based on their qualifications. The recruiter notifies applicants to inform them if there are special bonuses for selecting a certain Military Occupational Specialty (MOS) or if there are additional schools that can be added to their contract. If applicants want a job that is not available, then they have to choose from the list of offered jobs or decide not to enlist. Once a MOS choice is made, the recruiter processes the paperwork and allows applicants to select starting dates for basic training and the course to train them in their MOS. Most applicants, if offered the Health Care Specialist course, typically choose it. The next step is to complete the physical exam to determine applicants' physical readiness. Upon receiving a clean bill of health, all final paperwork is processed and the contract is signed, indicating the specificity of the enlistment (bonuses, additional schools, terms of service, etc.). The final step prior to reporting to basic training is to take the oath of enlistment.

I, _____, do solemnly swear (or affirm) that I will support and defend the Constitution of the United States against all enemies, foreign and domestic; that I will bear true faith and allegiance to the same; and that I will obey the orders of the President of the United States and the orders of the officers appointed over me, according to regulations and the Uniform Code of Military Justice. So help me God." (Title 10, US Code)

Basic Combat Training

The second major step to completing the transformation is to complete basic combat training (BCT). BCT is conducted for nine weeks and is located on one of five Army posts: Ft. Leonard Wood, MO; Ft. Jackson, SC; Ft. Sill, OK; Ft. Benning, GA; or Ft. Knox, KY. Several BCT locations are designed for co-ed training whereas a few are geared specifically for males. BCT is designed to transform the "civilian" applicants into "soldiers" whose personal values are consistent with Army's values. These values consist of: loyalty, duty, respect, selfless service, honor, integrity and personal courage (LDRSHIP). Until this occurs, applicants are called "trainees." This is a major transition for the typical individual since life as they previously knew it will dramatically change. During basic training "trainees" must submit to a rigid routine and allow the Drill Sergeants to take total control over every aspect of their life.

For the first three to five days of BCT, soldiers are located in a reception area in which they are issued uniforms, provided with identification cards, given immunizations and haircuts, and provided instruction in the basic rules they must follow for the next 9 weeks. During this period of time, alcohol and tobacco products are not allowed and intensive physical conditioning and basic military education commences. The most important person a soldier sees after leaving the reception center is their Drill Sergeant. The Drill Sergeant's job is to ensure that trainees are taught the fundamentals of being a Soldier. This includes skills such weapon qualification, combat maneuvers or marching, and learning basic Army rules and regulations as well as military customs and courtesies. This is done in a highly stressful context. BCT conducted at Ft. Jackson, SC, for example, has three phases: Red, White, and Blue. The red phase consists of the first two weeks in which trainees:

Learn the Army values; work on physical fitness; learning about communications, basic first aid, map reading, and the military justice system. They also practice drill and ceremony and negotiate Victory Tower [56 foot rappel tower]. Before moving to the next phase, soldiers must successfully complete knowledge and skills test.

The White Phase consists of week 3-5 in which trainees:

Continue Army values and physical fitness. Much of this phase is spent learning, practicing and qualifying on the M16A2 rifle. They will also learn about other U.S. military weapons, chemical warfare and bayonet training. Soldiers will participate in the obstacle course, gas chamber and bayonet assault course and pass another knowledge and skills test.

The Blue phase consists of week 6-9:

In addition to Army values and physical fitness this phase includes individual tactical techniques, foot marches, confidence course, and obstacle course. The culmination of basic training is Victory Forge, a 7-day field training exercise combining all previously taught basic combat skills. Soldiers march ten kilometers to their designated training site to start the exercise, occupy the position and establish a defense perimeter. On subsequent days, Soldiers complete the Teamwork Reaction Course, execute tactical exercise lanes and a night tactical and live-fire exercises. The last night includes a return march to the unit area and a ceremony recognizing the successful completion of this challenging operation - and the final transformation as a Soldier in the world's finest Army (Fort Jackson, 2007, Training Phases section, ¶ 2-4).

Another type of phasing is introduced to trainees in terms of privileges granted during BCT. Trainees are placed into a phasing cycle from the beginning of BCT which continues throughout Advanced Individual Training (AIT) which will prepare them for their MOS. There are six phases of training which are provided as milestones for trainees to strive for. Each phase allows trainees to have more privileges. The rules are as follows:

Phase 1: Week 1-3

- No passes. Soldiers are restricted to the company area except with cadre member
- No driving or riding in automobiles
- No civilian clothes
- No alcoholic beverages or tobacco products
- Must maintain battle buddy system

Phase 2: Week 4-6

- Same as above except:
 - Passes allowed within the brigade area to use swimming pools,

theaters, etc. that are not in the brigade area.

Phase 3: Week 7-9

- Same as above except:
 - On-post, yet no overnight passes are allowed; day passes are allowed with battle buddy on family day and on graduation day.
 - With the brigade or battalion commander's discretion, soldiers can ride with family members in automobiles on family day and graduation day.

 Soldiers may ride in automobiles with family members directly to AIT. They may not drive or consume alcoholic beverages when being transported.

Phase 4: Week 10-13 (AIT)

- If soldier passes the Army Physical Fitness Test (APFT) with 50 points per event, pass all academic standards, they will:
 - Receive all previous privileges (on-post and off-post passes to include use of electronic devices)
 - Must remain within a 50 mile radius of post
 - All passes must end no later than 2200 hours (10 p.m.)
 - Must wear uniform as determined by the commander
 - No driving of automobiles, however, may ride with family
 - Must maintain the battle buddy system
- During the second week of AIT, the end of week 11, if soldiers score 60 points per event on the APFT, and pass all academic standards, the commander may allow them to wear civilian clothes at prescribed times and to use tobacco and alcohol (if of legal age)
 - No overnights at Ft. Sam Houston, TX
 - May be included in the random drug testing
- Phase 5: Week 14-20
- Off-post and weekend passes are authorized, however no overnights at Ft. Sam Houston, TX

- Passes must end no later than 8 hours prior to the next training day
- Must maintain the battle buddy system

Phase 5+: Week 21+

- Soldiers are billeted separately from Soldiers in the lower phases; when this can not occur, alcohol and tobacco use is restricted.
- Provided privileges similar to fully qualified Soldiers (those who have completed AIT)
- Soldiers may be included in the random drug testing screening

Soldiers can regain prior privileges, if lost during phasing, by receiving a positive evaluation from the new company, and until that time, they will remain in Phase 4.

Trainees are expected to train from approximately 0530 to 2100 hours each day.

A typical day in the life of a BCT trainee, per Ft. Jackson, SC training schedule, consists of the following:

0530- Wake up 0600-0700 Physical Training 0700- 0800 Breakfast 0830 – 1200 Training 1200 – 1300 Lunch 1300-1700 Training 1700-1800 Dinner 1800-2000 Drill Sergeant time 2000-2100 - Personal Time 2100 - Lights Out

Training and Doctrine Command (TRADOC) Regulation 350-6 (2005) indicates that all trainees will participate in a battle buddy program beginning in basic training. This program requires course administrators to place trainees in 2-3 person teams for the following rationale: "mutual support and assistance; teaching teamwork; developing a sense of responsibility and accountability for fellow soldiers; improving safety during training; and reducing the likelihood and opportunity for sexual harassment, misconduct and suicide gestures and attempts" (p.26). Because battle buddies consistently train together they can develop deep relationships and help each other through the program. Many of the battle buddy teams report to AIT together and are again placed into the same platoons, allowing a continued relationship. Other forms of support for trainees may also include the relationships that have been developed with other platoon members, as well as with the unit Chaplain or available mental health personnel. *Advanced Individual Training*

The third major and final step toward transformation into a fully qualified soldier is to complete Advanced Individual Training (AIT). AIT is the specific job training that trainees receive immediately following BCT; it requires trainees to cope effectively in the academic and military environment simultaneously. It is conducted at multiple locations within the United States. This study focuses on the Health Care Specialist Course conducted at Ft. Sam Houston, TX; this program currently trains all Army medics, and soon will be where medics of all the services will be trained. The mission of the Army Health Care Specialist Program is to: "provide the Army with highly motivated, disciplined, warrior spirit Health Care Specialists ...who are National Registry EMT-B certified, possessing the additional necessary medical skills to sustain the force, survive the battlefield and accomplish the mission" (Hastings and Maness, 2007).

Academic structure

This course is conducted over 16 weeks and trains Active Army, Army Reserve, Army National Guard, and military students from other nations. The majority of the students in this study were Active Army. The course description is as followings:

The curriculum provides a foundation in fundamental health care knowledge and skills involving the administration of emergency medical treatment; evacuation; force health protection; and routine patient care, on the battlefield and in military treatment facilities. The course begins with CPR and a Department of Transportation (DOT)-based Emergency Medical Technician-Basic (EMT-B) curriculum, culminating in the National Registry (NREMT-B) Certification Examination. The remaining 10 weeks curriculum features advanced medical and patient care modules including a clinical experience, a Situational Training Exercise (STX), and a Field Training Exercise (FTX). The training consists of both classroom and practical exercises (Hastings and Maness, 2007, p. 3).

Specifically, classes are broken down into seven sections which include seven modules of EMT-B training. Prior to beginning EMT-B module training, students must pass a written and practical Cardiac Pulmonary Resuscitation (CPR) exam during the first week. The first section consists of EMT-B classes that cover: Medical/Legal Issues, Patient Assessment, Vitals, Airway Management, General Pharmacology, Respiratory, Cardiovascular and Neurological Emergencies, Substance Abuse, Bleeding, Soft Tissue, Eye, Chest, Abdomen, and Head and Spine Injuries, Musculoskeletal Care, Pediatric Airway, Geriatric Assessment and Ambulance Operations. The remaining sections are geared toward acquiring combat medic skills and cover the following areas: Limited Primary Care (medication administration, skin disorders, orthopedics, respiratory disorders) and Chemical Biological Radiation Nuclear Explosives (biological warfare, blood agents, decontamination of chemical casualties); Invasive Procedures (initiate and manage intravenous infusion, assemble a needle and injection, control bleeding and shock); Force Health Protection (Humanitarian laws, waste disposal in the field, heat/cold weather injuries, stress management); Combat Trauma and Evacuation (assessment and management of the trauma patient, head injuries, burn injuries, and litter and manual evacuation); Clinical Rotation and Situational Training Exercise (STX) and a Field training exercises (FTX) which requires the student to utilize all learned medical skills within the field environment (see Appendix F for a complete description of the curriculum).

The course requires students to be evaluated in both academic and nonacademic domains. To earn the MOS of Health Care Specialist, academically, they must be CPR qualified, pass all written and practical course exams, and pass the NREMT practical and written exams. Passing the NREMT is critical because it establishes the criterion by which emergency medical care is evaluated. Nonacademically, they must: adhere to the standards of conduct (not engaging in plagiarism, cheating, fraternization, stealing, drug

use, etc.); be medically qualified; and pass the APFT. If students fail to meet any of the above standards, they are recommended for recycle, reclassification, or relief from the course. Recycling from the course means that the student failed a test twice and is transferred to another company to be given an additional opportunity to successfully complete the course. Relief from the course is recommended when the student fails a test twice after they have already been recycled. Finally, reclassification occurs when the student has been relieved from the course and is given another MOS. Chapter 11 discharges from military service may also occur with students within their first six months of training. These discharges are for entry level performance and conduct difficulties and occur when students demonstrate unsatisfactory performance as evident by: "inability, lack of reasonable effort, failure to adapt to the military environment or minor disciplinary infraction" (AR 635-200, p. 81).

Policies and procedures are strictly enforced when evaluating trainees and these are thoroughly explained to students during formal counseling. Counseling is conducted at least three times throughout the course: initial, midterm, and at the end. Written counseling is also provided as needed for: exam failures, disciplinary action, lack of motivation, etc. Exams issued during the course include written, computer, practical, and the APFT. During the EMT phase of the course, testing occurs once or twice per week, whereas testing during the combat medic phase is conducted approximately once per week. The last six weeks of training is mostly hands on and the last week requires the student to consolidate all of their training and utilize those skills effectively in a field environment.

The pedagogy that is used in the course is highly formalized. Class instructors typically provide a pre-exam review of all covered material. Instruction is by lecture, often with a PowerPoint presentation. Exams are administered after each module. Following the exam, a post-exam review is conducted in which students are given their grades and are told which questions were most often missed by the class. If a student fails the exam, they are immediately counseled. Students are then required to attend a reteach class that evening for retraining of the material. Retesting, using a different version of the test, typically occurs during scheduled PT time at 0515 the next morning, so as not to interfere with regularly scheduled classes. If the student fails the test for a second time, they are again counseled and recommended for recycle to a follow-on class with another company (see Military Structure below). Any student who fails a retest twice is recommended for relief from the course and reclassification into another, less prestigious, MOS. Students are typically given two times to pass a written and practical exam before being recommended for recycle or relief from the program. Students are given approximately three times to pass the NREMT exam. After the first two failures, the student is counseled and required to attend study hall, a mandatory 16-hour retraining course, and to wait at least seven days after the previous exam scores are issued before retesting. During the retesting phase of the NREMT, students remain in class with their original company and continue with the combat medic training. If they pass the NREMT by the third attempt, they continue with the course. If, however, they fail to pass the exam after three attempts they are then recommended for academic relief (See Appendix D).

Some students enter the course already holding a NREMT certification. Once verification of their status is complete and they meet all requirements, they are progressed to a company that is beginning the combat medic phase of training, thereby shortening their length of stay in AIT. Students, however, who eventually recycled to another company, extended their stay in AIT. They typically are required to move to the same barracks that houses their new unit. This move is necessary to ensure that all students training in the same company are kept together for accountability purposes. Along with this move, they are required to integrate into a new unit and are assigned a new battle buddy.

Students who meet all of the academic and nonacademic standards are eligible for graduation and earn the MOS of Health Care Specialist (68W). Once they graduate, they have completed their final step in the process of becoming soldiers and are no longer considered trainees. They have demonstrated that they are willing to live by the Army's values, and that they were able to learn their job and demonstrate their proficiency while in a highly stressful environment. They are expected to perform as entry level Health Care Specialists upon their arrival at their first duty station. The majority of these soldiers are on active duty and will take leave prior to reporting to their new units to perform their jobs. The remaining students are Reservist and National Guard members and will either return to their home units to work as Health Care Specialists, or they will return to their previous jobs or look for new employment utilizing their new skills. For those students who are recycled to new units, many successfully complete the course, whereas some are recommended for relief from the course. Unless there are extenuating circumstances, students who are recommended for relief from the course are reclassified into another MOS or are discharged from the Army; they leave Ft. Sam Houston shortly after their failure, typically prior to the graduation of their peers.

There are both military and civilian instructors in the course. Military personnel can volunteer or be involuntarily selected for instructor duty. Instructor assignments are based on the following qualifications. They must: be a graduate of the program in which they are going to teach; possess a clean, five year background check; have held a variety of assignments; be physically fit and meet height and weight standards; be a graduate of the appropriate advanced military school, with at least one year of experience afterwards; have no speech impediment; have recently held a leadership position; have demonstrated the ability to be an instructor; and have no questionable personal habits (alcoholism, gambling, financial problems, emotional instability, etc). Civilians instructors are typically retired military Health Care Specialists, but they can also be civilian EMT trained personnel. Drill Sergeants are also carefully selected when being assigned to training schools. They too can volunteer or be involuntarily selected. Drill Sergeants in this course are also fully qualified Health Care Specialists. They are the MOS role models and serve as assistant instructors as needed (TRADOC Regulation 350-6, 2005).

All instructors, whether military or civilian, are required to be highly qualified and current in all aspects of their subject, which includes passing of the APFT for military instructors (TRADOC Regulation 350-6, 2005). Although all instructors are competent in all aspects of being an EMT and combat medic, they are typically assigned to teach in their preferred specialty area. Instructors are required to maintain their professional credentials. Along with being fully qualified in their skills, instructors of the Health Care Specialist Course are required to take the following courses at least 30 days prior to teaching in the classroom: Instructor Training Course (ITC), Support Cadre Training Course (SCTC), and the Installation Staff and Contractors Training Course (ISCTC). The ITC is an 80 hour course designed to provide basic instruction on the duties, skills, and competencies required of an instructor. The SCTC is a course designed for military personnel to review policies and procedures, installation policies, ethical conduct, Army values, investigations and Inspector General procedures, and Reserve and National Guard liaison activities. This course is also a prerequisite for the ITC course. Finally, the ISCTC is a five hour course for civilian personnel designed to provide a basic orientation to the classroom covering the same topics as the SCTC.

Health Care Specialists Military Structure

Each company structure consists of five platoons with approximately 80 students per platoon, depending upon the training rotation. There are seven companies operating within the battalion, resulting in approximately 2,800 students in training at any given time. The classes, depending on the rotation schedule, have an instructor-to-student ratio of approximately 1:50, with a Drill Sergeant-to-student ratio of 1:30. Instructors are primarily responsible for conducting academic training and Drill Sergeants are primarily responsible for nonacademic training. Students live and train together in on-post facilities. The living quarters vary by AIT location, and the Health Care Specialist course has communal living quarters. This means that there are typically 80-person open bays (sleep areas) and open shower facilities for all trainees. This type of environment, for some trainees, is similar to their BCT quarters; however, for others, these quarters are worse in terms of personal privacy.

Trainees are placed into a phasing cycle during BCT which indicates the level of privileges they receive, as discussed in the BCT section above. Students enter AIT in Phase IV and remain in this status for approximately four weeks. They can, however, be kept in this phase for disciplinary reasons, if needed. After the initial four weeks, and if their performance is satisfactory, they are progressed into Phase V and provided with additional privileges. This phase allows them the opportunity to have on and off post passes, utilize electronics outside of the classroom and finally wear civilian clothes, which had not been worn since they began BCT.

A typical day in the life of a student in this program is extremely busy and tightly structured. Below is an example of a typical weekday training schedule:

0430- Wake up 0515-0630- Physical training (PT) 0630-0700- Personal Hygiene 0730-0820- Breakfast 0820-0830- Value Training (emphasis on Army values) 0830-1140- Class, with breaks every 50 min 1150-1240- Lunch 1250-1720- Class with breaks every 50 min

1745-1835- Dinner

1900-2000- Company time: includes activities such as mail call, reteach (review of previous test material for soldiers who failed their last exam), study hall (for soldiers with grades below 80%), company announcements, issuing equipment, counseling, etc.

2000-2100-Personal time

2100 Lights Out

A typical weekend schedule, depending on the phase, is less demanding. In Phase IV, on Saturdays, the trainees participate in company scheduled activities during regularly scheduled class times, whereas in Phase V, students are typically on pass. Sunday mornings are dedicated to Spiritual Fitness for those who wish to participate and the afternoons are again filled with either company activities or passes, depending on phasing level and the company training schedule.

During the beginning of the course, as observed above in the daily schedule, students are provided with multiple tasks to complete in an assigned day. Students, according to the schedule, are provided with approximately one hour of personal time. This time is usually taken up by: calling home, preparing for the next day, studying and just unwinding. Also as noted from the schedule, some students are required to participate in study hall, when offered, if their exam scores are below 80%, and reteach/retraining, if they failed the test that day. These students are in a more structured environment, whereas students, who are considered above average are allowed to manage their own personal time. Students who were designated as above average were also able to participate in the study hall sessions on a space available basis only; however, they are seldom able to do this. For students who are passing the course, there is no designated place for them to study; instead they must study in public areas with all their distractions. Some also study by flashlight after lights out. Some students compete for special recognition by maintaining high standards of conduct and maintaining their grades at a high level. The student with the highest course point average is designated as the Distinguished Honor Graduate. Students who are within the top 5% of the graduating class, have an average grade of at least a 90%, pass the initial APFT, meet height and weight standards, pass all written exams, pass all initial practical exams, and have no adverse actions in their records are designated as Honor Graduates. Students who maintain less than 80% in the course are counseled and are required to attend study hall until their grades reach 80% or better. These students are placed in the at-risk group. Because of their poor performance, some students also have their privileges revoked or modified.

Administrator Perspectives

Course administrators view the Health Care Specialist Course positively. It is the most basic of the prestigious courses for enlisted personnel that the AMEDD conducts. This course was revamped from the 91W, basic combat field medic to a more enhanced 68W. This change creates a more advanced medic who possesses greater medical competency and is able to perform advanced procedures when compared to the former. This greater capability also enables medics to utilize their advanced skills to assist other

19

medical personnel with clearing the battlefield. The success of this program contributes to the overall mission of Army readiness and is not only vital to the AMEDD community, but also to the Army as a whole. The mission of the AMEDD is to "provide the Army with highly motivated, disciplined, warrior spirit Health Care Specialists who are National Registry EMT-B certified, possess the additional necessary medical skills to sustain the force, survive the battlefield and accomplish the mission" (AMEDDC&S, 2007, Mission Statement, ¶ 1).

Administrators value this course and are most concerned with ensuring that quality Health Care Specialists are being produced. A constant concern in all academic programs is attrition. Although some attrition is acceptable and expected in any course, increases above a set level bring about additional challenges. This challenge contributes to other concerns that the administrator already has. These include, but are not limed to: limited resources in terms of staff, space allocations, and budget. The continued increase in attrition means that more students will be in the course at any given time due to the need to retrain them on deficient skills. This requires additional staff and space in which to conduct the retraining, and it creates a housing problem. All of these issues ultimately result in extra budgetary demands thereby creating a vicious cycle for course administrators. Because of the increased commitment of our soldiers in wartime, more of a demand is placed on the AMEDD to get medics trained and sent forward to support fighting units. When the cause of attrition is fully understood and administrators implement programs to address these issues, attrition will decrease, and more quality soldiers will be sent forward.

Personal Perspective

My positionality as an African America female officer in the United States Army Medical Department contributes to the lenses through which I view the world. My prior military and professional experiences as a minority in terms of ethnicity and gender, my diverse educational background and having the opportunity to view challenges as both an enlisted person and as an officer have shaped the way in which I define meaning to all aspects of life. Because of my race, gender, and profession, various challenges have been presented that were geared toward competence and credibility issues. Because of personal experiences dealing with these challenges, both inside and outside of the classroom, I view education as an equal opportunity event. I believe that all individuals should be approached with dignity and respect, regardless of their social status, rank, gender, ethnicity, or educational level and that all individuals, who desire to, can learn and can contribute back to society. I also believe their prior experiences are important to their learning process and that the voices of all students should be heard.

My style as an administrator is teamwork focused. The team (staff) must understand the mission and work collaborately to get the job accomplished. This collaboration must be between the leadership, the instructors and the Drill Sergeants, and any other entity that interacts in the program. Each section has a specific job to accomplish and must work together in order to make the system work. In addition to this, the leadership must fully support all initiatives and routinely evaluate them to ensure that improvements are occurring within the program.

My perspective as an educator is that all students should be given an adequate opportunity to learn. I am concerned about the overall quality of medic that is being sent to the unit. Although they come to the course possessing the cognitive skills required to be successful, their ability to effectively manage the noncognitive factors that more often impact their lives, is not always as clear. Because not all soldiers will enter the program with the same level of coping skills, resources should be available with which to address these issues as they occur. My perspective as an educator is to allow all students the opportunity to succeed. This means creating an environment which fosters a positive attitude of success and excellence. A staff (administrators, DS and instructors) who presents themselves as caring leaders and who nurtures those soldiers who want to be there should focus on student strengths and steer them toward remaining focused on the goal. Ensuring that soldiers not only know what to do but also know why they are doing it is another one of my educational perspectives. In order to foster this level of understanding, soldiers must learn information in as much detail as is possible. Adequate time must be provided for study and interaction with the material to increase their competence if they are to be expected to provide the level of care being demanded on the battlefield. Finally, understanding that unmotivated students present a strain on the staff as well as the morale of others students, reclassifying or relieving them should continue to be expeditious.

Statement of the Problem

There is a vast amount of literature that discusses the impact of noncognitive factors on student attrition at the collegiate level. While a few studies have explored

noncognitive factors in the military setting, none have focused specifically on how it impacts attrition rates within recruits completing their initial job training during their first six months of military service. Tinto (1993) indicates that noncognitive factors contribute more to attrition than previously thought and that a combination of cognitive and noncognitive factors presents a more thorough picture of an individual at risk for attrition. Because of the shift of focus at the collegiate level to recognize the impact of noncognitive factors in attrition, there is a need for the military to also study the impact of how they help explain attrition rates among military recruits.

The Health Care Specialist, Military Occupational Specialty (MOS) consists of approximately 40,000 active-duty and Reserve personnel and forms the third-largest military occupational specialty in the Army (Army Medical Department, 2007). The course is a continuation of the soldier transformation process and is conducted in a high stress environment designed to challenge soldiers to work under extreme conditions. The Army Medical Department Center and School (AMEDDC&S) commanders have reported attrition rates as high as 20% for Health Care Specialists during AIT. Attrition is measured as the percentage of personnel who are unable to successful complete their MOS training with their original units. Although all soldiers, upon arrival for training, are deemed qualified to perform successfully as a Health Care Specialist, fully one-fifth of them either recycle and graduate, or fail the course and are reclassified or discharged from the Army. High attrition rates are costly to the Army, in terms of wasted resources and a lower number of fully qualified Health Care Specialists that they are able to provide to forward units. It is important to understand why recruits who have the cognitive ability to succeed nevertheless fail when in this academic environment.

Purpose of the Study

The purpose of this study is to understand how noncognitive factors contribute to attrition in the Health Care Specialist Program.

Research Questions

The study addresses these specific research questions:

- 1. What noncognitive factors explain attrition among Health Care Specialists students during AIT?
- 2. How do these noncognitive factors work together to result in drop out?
- 3. How do noncognitive factors, when combined with cognitive test scores, serve as a predictor of academic success in the military setting?

Definitions

The following is a list of terms as the researcher defined them for this study:

Advanced Individual Training (AIT). Job training that occurs immediately

following basic training that prepares trainees to work in their assigned military occupational specialty.

Armed Forces Qualification Test (AFQT). Multiple choice tests (subtest of ASVAB) used to determine qualification for enlistment in the United States military.

Armed Services Vocational Aptitude Battery (ASVAB). A timed, multi-aptitude test that helps students identify their abilities via eight modules: word knowledge, paragraph comprehension, mathematics knowledge, arithmetic reasoning, general

science, auto and shop information, mechanical comprehension, and electronics information.

Availability of strong support person. Seeks and takes advantage of a strong support network or has someone to turn to in a crisis or for encouragement

Community involvement. Participates and is involved in his or her community *Double-tap*. Failure of the same test twice.

Failing Student. A student who has failed to meet the minimum standards to move from the training environment to the work environment

First-term. Students who are still serving time under their first military enlisted contract.

GT (general technical) knowledge. A combination of word knowledge, paragraph comprehension and arithmetic reasoning scores on the ASVAB.

Knowledge acquired in a field. Acquires knowledge in a sustained or culturally related way in any field.

Leadership experience. Demonstrates strong leadership in any area of his or her background (church, sports, non educational groups, etc.)

MENAS. Military Environment Noncognitive Adjustment Scale. The modified Volunteer Survey that was renamed to more accurately describe its function.

Noncognitive factors. Factors not measured by cognitive tests, such as self concept, social support, motivation to achieve, leadership experience, community involvement, level of commitment, study skills, level of social integration, and social involvement

Nontraditional Student. Defined by Sedlacek (1991) as students who are other than White, middle to upper class, males.

Passing Student. A student who is maintaining a "C" average in AIT coursework.

ST (skilled technical) knowledge. A combination of general science, word knowledge, paragraph comprehension, mathematic knowledge and mechanical comprehension scores on the ASVAB.

Preference for long-term goals. Able to respond to deferred gratification; plans ahead and sets goals

Positive self-concept. Demonstrates confidence, strength of character, determination, and independence.

Reclassify. Reassignment of the trainee's MOS because of failure to complete the course.

Recycles. Trainees who have failed the current course with their assigned company but are given an additional opportunity to join another company to attempt to pass the remaining coursework.

Relief. Trainees who are released from the Health Care Specialist program who may either be reclassified into another MOS or discharged from the military.

Realistic self-appraisal. Recognizes and accepts any strengths and deficiencies, especially academic, and works hard at self-development; recognizes need to broaden his or her individuality

Successfully handling the system (racism). Exhibits a realistic view of the system on the basis of personal experience of racism; committed to improving the existing
system; takes an assertive approach to dealing with existing wrongs, but is not hostile to society and is not a "cop out"; able to handle racist system.

Assumptions

My approach to this study will be influenced by my own experiences of both participating in Health Care Specialist course as a student and interacting with the course administrators as an officer. I assume that my military experience and exposure to the procedures of the training environment will provide me with the background necessary to understand the experiences of my participants. I also assume that each student will be open and honest when providing feedback.

Limitations

Participants in this study are enlisted military members training in a highly stressful environment and generalizations will be made specifically for that population. The scope of this research will be reduced to include only Army Health Care Specialists within their first six months of military training. Generalizations can only be made to other military AITs with similar characteristics.

Having been not only a student in this environment but also an instructor, my experience may limit what I'm able to perceive and in some way it will shape my interpretation of the data for this study.

Significance of the Study

Few military research studies have explored the impact of noncognitive factors on attrition and none specifically on how noncognitive factors impact upon attrition rates within recruits completing AIT during their first six months of military service. Research specific to this population, within this environment, is extremely important in order to improve the Health Care Specialist program overall. Understanding how these students negotiate their environment to accomplish the task of learning is essential to administrators attempting to improve the course. Ultimately, the beneficiaries of this course will be injured soldiers on the battlefield who will rely on treatment provided by the medics that were produced by this course. I believe that the results from this study will increase the awareness of administrators and improve their ability to recognize important factors which influence attrition. Along with being aware of these factors, it is also believed that administrators will be able to use the results and recommendations to develop and implement future programs by which to decrease attrition, ultimately impacting the success of the medical mission.

CHAPTER II

METHODOLOGY

This chapter explains the research methodology which was employed for this mixed methods research study. The chapter summarizes the overall research design, sampling method and sample, data collection and data analysis for the study. The purpose of the research was to understand how noncognitive factors contribute to attrition in the Health Care Specialist Program. This population was selected for study because of their high attrition rate. Commanders indicate that in the last several years the attrition rate has climbed to approximate 20%, whereas an acceptable rate is 12%. All students selected for the course have met the cognitive requirements as demonstrated by their aptitude scores; however the reason for their continued attrition is not fully known. The data analysis methods that were used for this study included descriptive statistics, logistic regression, and constant comparative method. The goal for this mixed methods study was to make recommendations to commanders for increasing retention of students enrolled in the Health Care Specialist Course.

The research questions are:

- 1. What noncognitive factors explain attrition among Health Care Specialists students during AIT?
- 2. How do these noncognitive factors work together to result in drop out?
- 3. How do noncognitive factors, when combined with cognitive test scores, serve as a predictor of academic success in the military setting?

Research Design

This study was conducted as a mixed methods design consisting of both quantitative and qualitative inquiries. Mixed methods, as defined by Tashakkori and Teddlie (2003) and Creswell and Plano Clark (2007), is accomplished by integrating both quantitative and qualitative data within a study in order to get a better understanding of the research problem. They believe that neither method can paint a thorough portrait when used independently, but when used together they complement each other and allow for better understanding of the subject of study. This mixture of data therefore was chosen in order to provide a thorough understanding of the attrition phenomenon.

I chose Creswell's (2003) mixed-methods sequential explanatory design which involves collecting and analyzing first quantitative and then qualitative data in two consecutive phases within one study. Not only would using mixed methods in this study assist in determining the most important noncognitive factors and their significance in academic attrition but it will also enable us to better understand student perception of their academic failure.

Data collection consisted of quantitative data obtained by using a modified version of the Noncognitive Questionnaire (NCQ) developed by Tracey and Sedlacek (1984) and a modified version of the Volunteer Survey (Rice, Woods, & Bundy, 2004). The students were asked to provide information about their past academic performance, personal characteristics, general background and their perceptions and personal experiences with learning prior to the beginning of their AIT training, using the Modified NCQ, and again, at the midpoint of their training using the MENAS. This information assisted the researcher in determining which factors (cognitive and noncognitive) influenced attrition within the Health Care Specialist Course. For those students who were unable to successfully complete the course with their peers, termed recycled, qualitative data were obtained by using a semi-structured, face-to-face interview. Recycled students were transferred to another unit to be given an additional opportunity to be successful in the course. Qualitative interviews were designed to allow the student to elaborate on specific factors that they attributed to their attrition.

Sample

The sampling method used for this study was purposive sampling. This type of nonprobability sampling, defined by Merriam (1998), consists of selecting subjects with the most knowledge about the phenomenon of study and was appropriate as this explanatory study represented a collection of data from a unique population. Purposive sampling is also useful in qualitative studies to explore the personal experience of a particular population (Merriam, 1998). This method therefore allows for a greater understanding of the experiences of students transitioning though the Health Care Specialist Course.

Soliciting for a research sample through military channels requires permission from and coordination with multiple commands. Initially, a description of the study was presented to the Dean of the Army Medical Department Center and School (AMEDDC&S), Ft. Sam Houston, TX. This description was necessary in order to receive approval of the research topic and permission to conduct the research in the

31

subordinate commands. This presentation was again presented to the Director of the Combat Medic Training and the Health Care Specialist Battalion Commander to gain access to the military unit as well as to coordinate administrative and logistical support required to complete the study.

The AMEDDC&S conducts the Health Care Specialist Course at Ft. Sam Houston, TX, which trains Active Army, Army Reserve, Army National Guard and international military students. The course is sixteen weeks in length and "provides a foundation in fundamental health care knowledge and skills involving the administration of emergency medical treatment, evacuation, force health protection, and routine patient care, on the battlefield and in military treatment facilities" (Hastings and Maness, 2007). The first six weeks focused on performing as a basic emergency medical technician (EMT-B), whereas the remaining ten weeks focused on performing as a combat medic. The training consisted of both classroom and practical exercises.

The target sample size for this study, based on the estimations by Krejcie and Morgan (1970), was 335, in order to provide an appropriate ratio of students per independent variable. A total of 434 students from two companies [Company C (Charlie Company): N=80; and Company E (Echo Company): N=357) who were within their first six months of training, volunteered to participate in the study; these were combined into one group for analysis purposes. There were approximately 2,500 students in the training environment at any given period. The attrition rate in the last several years, as reported by various commanders, had been approximately 20% and because of these statistics approximately 87 students were expected to be in the attrition group. The actual attrition rate for this study was N=82 or 19%, therefore the attrition rate within this study was an adequate sample of students typically training within the Health Care Specialist course.

Demographic information for the participants is provided in Tables 2.1, 2.2, and 2.3. All participants were within their first six months of military training.

Demographic Characteristic	Frequency	Percent	(N=434)
Gender			
Males	281	65	
Females	153	35	
Ethnicity			
Áfrican-American/Blac	k 28	7	
White	325	75	
Asian	19	4	
Hispanic	45	10	
American Indian	3	.7	
*Other	13	3	
No response	1	.2	

Table 2.1. Gender and Ethnicity

* The Other category included thirteen students who identified as bi-racial.

Demographic Characteristic	Frequency	Percent	(N=288)
High School Diploma	121	42	
Some High School with With GED	11	4	
GED	12	4	
Some College	113	39	
College Diploma	31	11	

Table 2.2. Highest Level of Education

 Table 2.3. Age and Aptitude Scores

Demographic Characteristic	Mean	Standard Deviation	(N=434)
Age	22	4.73	
Aptitude Scores			
Skilled Technical	116.92	7.838	
General Technical	116.74	7.080	

Data Collection

The purpose of this study was to understand how noncognitive factors contribute to attrition in the Health Care Specialist course during AIT. Two quantitative surveys were used: the Modified NCQ and the modified Volunteer Survey. The NCQ developed by Terrence Tracey and William Sedlacek is a 29-item instrument with both Likert scale and open-ended questions which has been validated in predicting grades and student retention at the college/university level. This questionnaire was designed to measure eight noncognitive factors: positive self-concept, realistic self-appraisal, understand and deals with racism, prefers long-range goals to short-term needs, availability of strong support person, successful leadership experience, demonstrated community service and knowledge acquired in a field, which may impact student academic performance and ability (Sedlacek, 2004). Slight modifications were made to the survey to gear the questions to students in a military educational environment. For example, Sedlacek stated, "I am as skilled academically as the average applicant to this school" was changed to "I am as skilled as the average student at AIT."

The Volunteer Survey was developed by behavioral science researchers at the Army Research Laboratory (ARL) in the early 1990's and was modified by Rice, Woods, and Bundy (2004) to address a variety of attrition factors. It was one of two questionnaires developed to be administered to soldiers and their battle buddies after their attrition from the Health Care Specialist course. It consisted of a mix of 83 closed and open ended questions. The content of the questions was based on the results of focus groups with Health Care Specialist instructors, Drill Sergeants, and Command Staff, and areas identified as being predictive of academic failure in the general literature. The primary variables in the original questionnaire included: motivation, sleeping patterns, studying habits, stress perception, coping strategies, learning disabilities, health status, leadership abilities, unit cohesion, morale, indices of family status, demographics and other situational and organizational factors that may contribute to attrition (Rice, Woods, & Bundy, 2004). The Volunteer Survey was modified by the researcher to specifically focus on the areas of noncongitive factors and renamed the Military Environment Noncognitive Adjustment Scale (MENAS). For the remainder of the paper, MENAS will be used to identify the modified Volunteer Survey. Variables chosen for this study included motivation, sleep patterns, stress perception, coping strategies, unit cohesion, and morale. After modification, it consisted of a 46-item Likert scale instrument which also contained several open ended questions. It was designed to further explore noncognitive factors which may have influenced the student's academic performance since the student started the course.

A face-to-face semi-structured interview was administered to only the recycled students. It allowed them the opportunity to elaborate on various factors that may have contributed to their academic status. The interview followed the administration of the MENAS and consisted of a series of audio-taped, semi-structured, open-ended questions designed to focus on any factors that the student perceived as contributing to their academic performance. These questions were especially important to: gain a greater depth of the phenomenon (Creswell, 2003); and clarify, restate and receive further elaboration from specific responses on the questionnaire and survey (Merriam and Simpson, 2000). The interviews also served as a tool for the researcher to observe the student's emotional demeanor. The questions asked of the students were two fold; first, they allowed the student the opportunity to elaborate and provided valuable input as to what factors they perceived to have had an impacted upon their academic performance and secondly, they informed the researcher of factors that may not have been expressed during either the NCQ or the MENAS.

All 434 students participated in the Modified NCQ, however, only 288 participated in the MENAS due to student attrition and schedule conflicts. The sample of N=288 therefore is used with any data taken specifically from the MENAS. Of these, a total of 20 students participated in the face-to-face interview.

Because the military environment is based on a rank structure, the researcher did not want student interactions to be overly influenced by the perception of power and negatively impact their participation in the study. Although the researcher introduced herself using rank during all interactions, she chose not to wear her military uniform. This allowed all interactions to be focused on the student's concerns and not on rank and/or perceived positions of power. The actual process of data collection is described below.

Upon receiving approval to conduct the study, an incoming company of students were provided a description of the study and volunteers were solicited during their first inprocessing week. Only the researcher and the ombudsman were allowed to remain in the room during the presentation to the students in order to prevent any external influences either for or against the study by course faculty. The ombudsman was an impartial third party that served as an advocate for the proposed population. They were familiar with the nature of the study and their role was to address any student concerns regarding the research and ensure that student's rights were respected by the researcher. The researcher provided them with specific details pertaining to the nature of the study, their rights as participants, and any risks and benefits that would be associated with participation. Students who agreed to participate in the study were asked to complete the first survey instrument (NCQ). The NCQ was administered at this time in order to obtain the student's perception of their past academic performance, their personal characteristics and information about their general background at the beginning of the course. The completion of the Modified NCQ took approximately 20 minutes.

The MENAS was administered to both passing and recycled students. All passing students completed the MENAS, which was distributed between Weeks 11 and 12, after the National Registry of Emergency Medical Technician (NREMT) Exam, the most academically challenging portion of the course. Again, the timing for administrating the second survey was to receive another picture of the student's perception of their academic performance at that point and any other information that they attributed to their current academic status during the course.

Any student who was unable to successfully progress with their original unit was recycled to the next training unit, typically two weeks behind in the curriculum from the original unit. A student recycled when they failed an exam, was retrained, and subsequently failed the same exam again; in the Army this is called "double tap." Once a student was recycled from the course, personnel from the academic support division of the Department of Combat Medic Training (DCMT) notified the researcher of their status by email. The researcher then coordinated arrangements with the receiving unit to administer the MENAS. The survey was administered individually and was immediately followed by a face-to- face tape-recorded interview which took approximately thirty minutes. This interview was conducted in a private room without external influences from the student's supervisory personnel. The interview was audio-taped to provide qualitative data which would aid in gaining a better understanding of which factors the student attributed to their performance in the course. Prior to this session the researcher reviewed and analyzed the failed student's NCQ responses and used them to probe for noncognitive factors that may have affected the student's academic performance. Although data received from the face-to-face interviews achieved saturation at N=20 and the researcher discontinued using this method for data collection, the remaining recycled students continued to participate in the MENAS and were provided additional space to elaborate on any personal experiences during the course which may have influenced their academic status.

As a final step in the process, the student's GT, ST and GPA test scores were retrieved from DCMT staff to be analyzed as cognitive factors. The ST and GT scores are cognitive/aptitude scores taken from the Armed Services Vocational Aptitude Battery (ASVAB). The results are deemed valid predictors of performance in high school, post-secondary education courses, job skill training, and various military enlisted and civilian occupations (Personnel Procurement, 2005). These scores were retrieved to provide a historical view of prior cognitive scores on standardized qualifying exams. The course GPA provided a current view of the student's cognitive performance in the program. All surveys and interviews were conducted during the student's off time to avoid interference with other course requirements.

A snapshot of the order and typical timing of the procedures are listed below:

1. Researcher received approval and coordinated administrative support and briefing time to explain research study to incoming students with the Battalion Commander (BN CDR)

2. Researcher completed consent process and administered Modified NCQ

3. Researcher retrieved ST and GT scores from unit personnel

4. DCMT personnel informed researcher of course recycles

5. Researcher coordinated with receiving unit to schedule the MENAS and interview with the recycled student.

6. Researcher contacted unit personnel to coordinate a time to conduct the MENAS with passing students.

7. DCMT personnel provided GPA scores for all students

This process is represented schematically in Figure 2.1.



FIG. 2.1. Schematic showing the levels of approval required for the data collection process.

Data Analysis

Data analysis was accomplished using the SPSS 12.0 statistical package. Descriptive statistics were used for demographic data; inferential statistical methods were used to answer quantitative questions, whereas the constant comparative method was used to answer the qualitative questions. The goal of the data analysis was to: explain what noncognitive factors explain attrition; understand how these noncognitive factors work together to result in attrition; and understand how noncognitive factors when combined with cognitive test scores serve as a predictor of academic success.

Logistic regression was used to analyze the closed-ended, Likert scale questions on the Modified NCQ and the MENAS. It was used to study the relationship between attrition and the cognitive and noncognitive variables when answering questions one and three. The goal was to identify the variables that were most useful as predictors for attrition. This approach allowed the researcher to understand which noncognitive variables displayed enough variability to influence academic attrition. Descriptive statistics were also used to analyze demographic information of the sample.

Modified Noncognitive Questionnaire

Because the original version of the NCQ was modified to better accommodate the military population, various levels of modifying and recoding were necessary to correctly score the NCQ prior to statistical analysis. The first question which requested the student's social security number was omitted since each student was provided with a traceable number. Because of this omission, the Modified NCQ had 28 questions instead of the original 29; therefore scoring for the Modified NCQ began with number six instead of number seven. Since this questionnaire was primarily geared for university students, various questions had to be modified and geared to the military population. The following were the modifications for the Modified NCQ:

Q6: How much education do you expect to get during your lifetime? Choices: college, but less than a bachelor's degree; B. A. or equivalent; One or two years of graduate or professional study (master's degree); or Doctoral degree such as M.D., Ph.D., and so on.

Modification: How much education do you expect to get during your lifetime? Choices: military training only; college, but less than a bachelor's degree; B. A. or equivalent; One or two years of graduate or professional study (master's degree); or Doctoral degree such as M.D., Ph.D., and so on. The addition of "military training only" increased the options to five.

Q8: About 50 percent of university students typically leave before receiving a degree. If this should happen to you, what will be the most likely cause? Choices: absolutely certain that I will obtain a degree; to accept a good job; to enter military service; it will cost more than my family can afford; marriage; disinterest in study; lack of academic ability; insufficient reading or study skills; other.

Modification: Approximately 20% of trainees typically leave before completing 68W AIT. If this should happen to you, what will be the most likely cause? Choices: absolutely certain that I will complete 68W AIT; to change my MOS; to

accept a civilian job; marriage or family distractions; disinterested in study; lack of academic ability; insufficient reading or study skills; other.

Q10: The University should use its influence to improve social conditions in the state.

Modification: The military should use its influence to improve social conditions.

Q11: It should not be very hard to get a B (3.0) average at this school.

Modification: It should not be very hard to get a B (3.0) average in AIT.

Q20: I am as skilled academically as the average applicant to this school.

Modification: I am as skilled academically as the average student at AIT.

Q21: I expect I will encounter racism at this school.

Modification: I expect I will encounter racism during AIT.

Q23: My friends and relatives don't feel I should go to college.

Modification: My friends and relatives didn't feel I should come into the military.

Q24: My family has always wanted me to go to college.

Modification: My family has always wanted me to go into the military.

Q25: If course tutoring is made available on campus at no cost, I would attend regularly.

Modification: If course tutoring is made available to me while at AIT, I would attend regularly.

None of the modifications affected the scoring of the survey.

Responses to Q7: Please list three goals that you have for yourself right now; Q9: Please list three things that you are proud of having done; and Q29: Please list offices held and /or groups belonged to in high school or in your community were open ended and their means were used for scoring. When the student did not complete at least three responses as requested on question seven and nine, a zero was placed in that spot and the mean was rounded to the nearest whole number, and then recorded. Questions 12, 15, 17, 22 and 23 were written in a positive format and were scored exactly as the student responded. The remaining questions however, were written in a negative format which required the scoring to be completed in a reverse manner in order to give the value of 5 to the most positive responses.

Internal reliability analyses were calculated for the eight variables from the NCQ. Cronbach's alpha values were used to determine if all items that were designed to measure the same variable actually did measure that variable. George and Mallery (2006) indicated that the closer the Cronbach's alpha is to 1.0 the greater the internal consistency. The Cronbach's alpha for the NCQ eight variables were as follows: positive self-concept, .185; realistic self-appraisal, -.198; understands and deals with racism, .373; prefers long-range goals to short-term or immediate needs, .000; availability of strong support person, -.986; successful leadership experience, .107; demonstrated community service, .060 and knowledge acquired in a field, .058. The results of the internal reliability test were extremely low and according to George and Mallery, unacceptable. Previous studies by Woods and Sedlacek (1988), Ting and Sedlacek (2000) and Carter (2006) either emphasized only the test-retest and inter-rater reliabilities or factor loading in combination with other instruments. This low internal reliability prompted the researcher to conduct a factor analysis to determine if higher Cronbach's alphas could be obtained.

Factor analysis, according to Pedhazur and Schmelkin (1991), is an analysis designed to identify factors which share variance with observed variables. The idea is to place similar variables into a category together. George and Mallery (2006) indicated that it is often used to "identify a small number of factors that may be used to represent relationships among sets of interrelated variables" (p. 246). Once this relationship has been identified, factor loading occurs and assigns a weight to each variable within a similar category or construct. This weight is typically between plus one and minus one. The researcher used Principal Component Analysis with the varimax orthogonal rotation to analyze the data. The eigenvalues (the proportion of variance explained by each factor) from the total variance explained chart and the factor loading from the rotated factor matrix were used to determine which variables were to be retained and reorganized. The results indicated that eleven factors should be retained. After rotation, the eleven factors accounted for the following percentages of variance: 10.20%, 8.47%, 6.70%, 6.53%, 5.90%, 5.86%, 5.25%, 5.18%, 5.11%, 4.50%, and 4.31% resulting in 68% of the total variance (see Table 2.4).

Component	Total	% of Variance	Cumulative %
1	2.753	10.196	10.196
2	2.286	8.466	18.662
3	1.808	6.696	25.358
4	1.765	6.535	31.893
5	1.592	5.898	37.791
6	1.583	5.863	43.654
7	1.417	5.247	48.901
8	1.399	5.183	54.084
9	1.379	5.108	59.192
10	1.214	4.498	63.690
11	1.162	4.306	67.995

 Table 2.4. Modified NCQ Total Variance Explained

Extraction Method: Principal Component Analysis

NCQ Questions					ł	Factor	rs				
	1	2	3	4	5	6	7	8	9	10	11
6. How much education do you expect to get during your lifetime?	- 9										.663
8. Likely reason for attrition?		.973									
Mean of 9.1, 9.2, and 9.3								.622			
19. When I believe strongly in something, I act on it.							.777				
22. People can easily change me even though my mind was made up.										.431	.405
27. My high school grades don't really reflect what I can do.											
11. It should not be very hard to get a B average in AIT.									.81	3	
20. I am as skilled academically as the average student at AIT.									.51	0	
10. The military should use its influence to improve social conditions.								.555			
17. I expect to have a harder time than most students in AIT.		.395									

Table 2.5. Modified NCQ Rotated Factor Matrix

NCQ Questions					Factor	rs				
1	2	3	4	5	6	7	8	9	10	11
21. I expect I will encounter racism during AIT.										
25. If course tutoring is made available to me, I would attend regularly.				,	799					
26. I want a chance to prove myself academically.					740					
Mean of 7.1A, 7.2A, 7.3 A		.886								
12. I get easily discouraged when I try to do something and it doesn't work.									.827	
18. Once I start something, I finish it.						.684				
14. If I run into problems, I have someone who would listen to me and help.			.364	1				.33	4	
23. My friends and relatives didn't feel I should come into the military.										
24. My family has always wanted me to go into the military.			-	845						
13. I am sometimes looked up to by others.			.806	5						
16. In groups where I am comfortable, I am often looke to as leader.	d		.75()						

Table 2.5. Continued

NCQ Questions						Fact	ors				
	1	2	3	4	5	6	7	8	9	10	11
Mean of 28.1A, 28.2A, and 28.3A	.886	1									
15. There is no use in doin things for people, you only find that it doesn't pay off in the long run.	g							.547			
Mean of 28.1B, 28.2B, and 28.3B	.967										
Mean of 7.1B, and 7.2B			.906								
Mean of 28.1C, 28.2C, and 28.3C	.949										

Table 2.5. Continued

All factors within the Rotated Factor Matrix consisted of both positive and negative loadings. The following were the questions with the highest positive loadings on the factors: Factor 1: the mean of 28.1A, 28.2A and 28.3A, the mean of 28.1B, 28.2B, and 28.3B, and the mean of 28.1C, 28.2C and 28.3C; Factor 2: 8 and 17; Factor 3: the mean of 7.1A, 7.2A and 7.3A, and the mean of 7.1B, 7.2B, and 7.3B; Factor 4: 13, 14 and 16; Factor 5: 24; Factor 6: 25 and 26; Factor 7: 18 and 19; Factor 8: the mean of 9.1, 9.2, and 9.3, 10 and 15; Factor 9: 11, 20, and 14; Factor 10: 12 and 22, and Factor 11: 22 and 6 (see Table 2.5). A follow up reliability analysis with all of the rotated factors revealed that 6 of the original factors: Factors 5 and 7-11 continued to have

unacceptable Cronbach's alpha values. The remaining five factors produced acceptable rating of .900, .0823, .575, .663 and .537. Another factor analysis was conducted with the remaining five factors which demonstrated eigenvalues of one or better. The percentages of variance for these factors were the following: 24.40%, 15.48%, 13.67%, 13.67%, and 12.68% and accounting for 79.89% of the total variance (see Table 2.6).

	Rotation Sums of Squared Loadings								
Component	Total	% of Variance	Cumulative %						
1	2.684	24.404	24.404						
2	1.703	15.481	39.885						
3	1.504	13.673	53.558						
4	1.504	13.672	67.231						
5	1.392	12.658	79.889						

 Table 2.6.
 Total Variance Explained

Extraction Method: Principal Component Analysis

All factors within the Rotated Factor Matrix consisted of both positive and negative loadings. The following were the questions with the highest positive loadings on the factors: Factor 1: the mean of 28.1A, 28.2A and 28.3A, the mean of 28.1B, 28.2B, and 28.3B, and the mean of 28.1C, 28.2C and 28.3C; Factor 2: the mean of 7.1B, 7.2B, and 7.3B, and the mean of 7.1A, 7.2A and 7.3A; Factor 3: 8 and 17; Factor 4: 13 and 16; and Factor 5: 25 and 26 (see Table 2.7). After rotation, Factor 1, similar to a category identified by Carter (2006) was labeled demonstrated *school and community*

involvement, which was slightly different from Sedlacek original variable of demonstrated community service. The remaining factors maintained the names originally described by Sedlacek: Factor 2, *prefers long-range goals to short-term or immediate needs*; Factor 3, *realistic self-appraisal*, Factor 4, *successful leadership experience* and Factor 5, *understands and deals with racism*.

MENAS

Internal reliability analyses were also calculated for the variables from the MENAS. The Cronbach's alpha for the MENAS were as follows: battle buddy support, .832; stress, .771; motivation, .708; unit support, .715; expectations, .743; and family support, .634. The results, according to George and Mallery (2006), were acceptable for factor analysis to be conducted. The factor analysis was performed and again, the eigenvalues from the total variance explained chart and the factor loading from the rotated factor matrix were used to determine which variables were to be retained and reorganized. The results indicated that six factors should be retained. After rotation, the six factors accounted for the following percentages of variance: 13.46%, 11.88%, 9.87%, 8.79%, 6.76%, 6.41%, resulting in 57.18% of the total variance (see Table 2.8).

All factors within the Rotated Factor Matrix consisted of both positive and negative loadings. The following were the questions with the highest positive loadings on the factors: Factor 1: 34, 38, 36, 37, and 15; Factor 2: 14, 25, 17, 16, 43, 1, 22 and 10; Factor 3: 18, 19, 22, 10, 21, 20, and 11; Factor 4: 36, 44, 45, and 39; Factor 5: 27 and 26; and Factor 6: 32 and 31 (see Table 2.9).

NCQ Questions			Factors		
	Involvement	Goals	Appraisal	Leadership	Racism
Mean of 28.1B, 28.2B, and 28.3B	.974				
Mean of 28.1C, 28.2C, and 28.3C	.957				
Mean of 28.1A, 28.2A, and 28.3A	.890				
Mean of 7.1B, and 7.2B		.919			
Mean of 7.1A, 7.2A, and 7.3 A		.917			
8. Likely reason for attrition	?		.809		
17. I expect to have a harder time than most students in AIT.			.808		
13. I am sometimes looked up to by others.				.859	
16. In groups where I am co I am often looked to as leade	mfortable, er.			.849	
26. I want a chance to prove academically.	myself				.865
25. If course tutoring is mad to me, I would attend regula	e available rly.				.782

Table 2.7. Rotated Factor Matrix

.

Extraction Method: Principal Component Analysis. Rotation Method: Varimax with Kaiser Normalization.

Component	Total	% of Variance	Cumulative %
1	3.500	13.462	13.462
2	3.089	11.881	25.343
3	2.567	9.873	35.216
4	2.286	8.794	44.009
5	1.757	6.756	50.765
6	1.667	6.410	57.176

 Table 2.8. MENAS Total Variance Explained

Extraction Method: Principal Component Analysis

MENAS Questions			Factors	5		
Batt S	tle Buddy upport	Stress	Motivation	Unit Support	Expectations	Family Support
34. I can approach my battle buddy to talk to him/her about personal matters and/or problems in my life.	.831					
38. I perceive my relationship with my battle buddy to be "close."	.816					
35. My battle buddy helps me and is supportive of me in this course.	.810					
37. I often talk to my battle buddy about my personal or academic problems.	.741					
15. I have people to talk to about my problems and/or stress in my life.	.626					
36. There are other people, besides my battle buddy that I can turn to for help and support here.				.387		
14. My stress level affects my academic performance.		.812				

Table 2.9. MENAS Rotated Factor Matrix

MENAS Questio	ns		Factors	5		
	Battle Buddy Support	Stress	Motivation	Unit Support	Expectations	Family Support
25. My personal problems affect						
my academic performance.		.722				
17. I have trouble concentrating	•					
which affects my academic perform	nance.	.674				
16. I am coping v and managing my stress well.	vith V	.596				
43. I am having a difficult time dea with failure.	ling	.569				
1. I attribute my problems with the course to:	e	.487				
18. I am very motivated to pass this course.			.728			
19. I am doing m best to pass the co	y ourse.		.671			
22. I find this couse of the second s	urse have g to	.323	.622			
10. I have what it to be successful i course.	takes n the	.369	.573			

Table 2.9. Continued

MENAS Questions	Factors				
Battle Buddy Support	Stress	Motivation	Unit Support	Expectations	Family Support
21. <i>I did</i> fail this course on purpose.		.545			
20. I have <i>considered</i> failing the course on purpose.		.480			
11. My grades and my academic performance are my responsibility.		.338			
44. I have bonded well with my unit.			.768		
45. I have a strong sense of belonging here.			.707		
39. My unit is supportive of me.			.626		
27. This MOS is very similar to what the recruiter described to me.				.868	
26. The 68W MOS is what I expected.				.821	
32. My family wants me to find a way out of this course and/or get out of the Army.					.791
31. My family or people who are close to me are supportive of me and my role as a combat medic.					.716

Tabl	e 2.9 .	Continu	ed

The factors were assigned the following labels: Factor 1, *battle buddy support*; Factor 2, *stress*; Factor 3, *motivation*; Factor 4, *unit support*; Factor 5, *expectations*; and Factor 6, *family support*.

Interview

Qualitative analysis, using the inductive approach of the constant comparative method was performed to identify key ideas and recurring themes. The open-ended questions from the Modified NCQ, and the MENAS, as well as the results of the final interview, were used to establish categories which reflected the purpose of the study. This approach was used to continuously compare and categorize data as it was received. Merriam (1998) suggests that categories, which reflect the purpose of the study, often answer the research questions.

Merriam and Simpson (2000) recommend the use of four steps to establish a grounded theory when conducting constant comparative analysis. First, the researcher compares incidents, generate tentative categories and code each incident; second, integrate the categories and their properties; third, reduce categories into smaller categories, generate hypotheses, and check data for an overall fit until data saturation occurs; and fourth, establish a theory that reasonably and accurately represents the research subject. In other words, the constant comparative method challenges the researcher to continuously compare and analyze data in order to develop a grounded theory.

Following the guidance of Merriam and Simpson (2000), the researcher proceeded by recording data as it was received and placed it into categories which best represented the topics revealed. After continuous recategorization and recoding, similar themes and hypotheses began to emerge. Data were organized in a meaningful manner to allow the researcher to get a global view and attempt to understand how various factors influenced attrition. As a final piece of the process, the researcher used direct quotes to further support the categories that were revealed. Merriam (1998) states, "When categories and their properties are reduced and refined and then linked together by tentative hypotheses, the analysis is moving toward the development of a theory to explain the data's meaning" (p. 192). The constant comparative method contributed to allowing the researcher to generate a theory that was grounded in the data received from students who were unsuccessful at completing their assigned course within the first six months of training.

When initially categorizing the data, various topics and patterns emerged which reflected specific themes related to cognitive and noncognitive factors as well as environmental distracters. Thinking deeper about the emerging themes revealed that these factors/distracters are often intertwined in each student and it is difficult, if not impossible, to separate when discussing the overall performance. Further analysis resulted in categorization of the data into the following themes: *course structure* and *noncognitive factors.* These themes reflected how all factors/distracters worked together to accurately represent attrition in the military setting as well as help to establish a grounded theory (see Figure 2.2).

59



FIG. 2.2. Qualitative themes showing the interaction between the student, the course structure, and noncognitive factors.

CHAPTER III

COURSE STRUCTURE

The Health Care Specialist course is extremely demanding. Course administrators expect students to enter the environment motivated and ready to tackle the demands involved in learning their future job. Students enter the environment with the anticipation that they will leave the course possessing the skills necessary to become an entry level Health Care Specialist and be able to function as a competent EMT-B. A successful interaction between the student and the environment will lead to graduating on time; however, a mismatch between the two will create conflict, thereby preventing adequate academic and social integration and possibly leading to attrition. Academic integration as described by Tinto's (1975) Theory of Student Persistence is the combination of grades and intellectual development which ultimately leads to goal commitment. Social integration is the combination of peer group and faculty interaction which ultimately leads to institutional commitment. Both integrations are important factors for persistence; however a deficit in either can influence retention.

As previously noted, attrition in the Health Care Specialist Course is a major problem for the Army. While in this course, students may face multiple unexpected challenges that will force them to evaluate their commitment to being successful in the program. In order to understand why attrition occurs, it is necessary to understand the experience of those students who are not successful. This chapter focuses on that group, students who failed an exam two times, and thus recycled to another company. In this study, the majority had done so during Phase 4 when privileges were the most restrictive. These were the students who participated in face-to-face, audio recorded interviews with the researcher shortly after they were recycled and immediately following the administration of the MENAS. Twenty students in total were interviewed. There were 16 (80%) males and 4 (20%) females with a mean age of 21.85 (SD = 4.870). The ethnicity was as follows: Black, 1 (5%); White, 10 (50%); Asian, 2 (10%); Hispanic, 4 (20%); American Indian, 1 (5%); and Other, 2 (10%). Their highest level of education prior to entering training was: high school diploma, 9 (45%); some college with GED, 1 (5%), some college, 8 (40%); and college diploma, 2 (10%). A review of the MENAS surveys was completed prior to the interview and unclear responses were used to probe for noncognitive factors that may have influenced the student's academic performance.

The major challenges students faced stemmed from the pedagogical structure of the course and the environment in which it was situated. Pedagogical factors included the pace of the instruction, the teaching method, test review sessions, reteach, and study hall. Factors deriving from the military environment included the schedule, limitations of time (study and personal), and environmental distractions. The two sets of factors will be discussed in turn.

Pedagogy

Pedagogy is defined as the study of teaching or a combination of daily teaching practices and assessment techniques used by teachers (Newman, 1995). Many scholars (Kember, Leung and Ma, 2007; Ramsden, 1987) argue that a student's learning is influenced by their perceptions of the teaching and learning environment. If actual
learning is taking place, students tend to value it and are more likely to be retained in the course (Tinto, 2002). This process involves effectively engaging students in the process of learning. Teachers who demonstrate the ability to do this have been called effective or exemplary teachers. Effective teachers as viewed by students, according to Feldman (1976), possess the following characteristics: concern and respect for students, knowledge of their particular subject matter, available and helpful, open to others opinion and encourages discussions, able to explain clearly, enthusiastic, able to stimulate interest, and organized and prepared. Maden (2003) argues however, that the most prominent characteristics for effective teachers are the following: demonstrates knowledge of the subject matter, effectively paces and manages their classrooms, and teaches with enthusiasm.

Pace

The speed in which the course was taught was widely cited by the majority of the recycled students as a problem. Hall (2002) argues that the pace of instruction is influenced by the "task complexity or difficulty, relative newness of the task, and individual student differences" (p. 4). The author identifies three benefits for engaging students at a fast pace: greater information is shared, students are highly involved in the activity, and they tend to stay focused during instruction. Although a fast pace may create positive outcomes for some students, for other students it may contribute to them becoming overwhelmed.

The Health Care Specialist course is rigorous and demanding. It requires students to absorb new knowledge very quickly, as the course is taught using a strict time

schedule. Many recycled students stated that the pace of the course was faster than they expected and suggested that it negatively impacted their performance in the course. When asked, "What are some problems that you have experienced in the course that may have affected your academic performance," several students answered, "Just the material coming so fast...with the test right away and slides all day. Then they expect you to read"; "It's like there is class work pounding, pounding, pounding and you never get a break." A third student said:

Everything gets so backed up. We did ten chapters in two days and then we had to do the test. After the test, we took the final and I think that's what really screwed me up; having everything crammed up into a week. I just got so much pushed at me at once.

Another student commented:

It is a lot of back-to-back...we had maybe three or four days to go over and review ten chapters before we had a test and there were days when we had a test on Wednesday and another on Friday and a test on Tuesday, so there was a lot to soak in.

Students commented about not only the pace of the course but also the impact that nonacademic tasks had on their performance. One student noted:

I was really overwhelmed with the speed that we were getting the information. That was week one and a lot of things were going on other than just the classes. We had all the other things like the Drill sergeants...and that was pretty stressful. Finally, some students felt that the fast pace of the course was purposefully used to identify and remove the students who were not likely to pass the course. One student complained:

My head is still spinning. I guess the company that I am with just took Module 5, two days ago, had a weekend...and then took Module 6 and Module 7. Two days after that was the final. I sometimes feel like they are overflowed or something and they are trying to get rid of people or reclassify them or something. Sometimes it just feels that way.

Test Review

Several students attributed their poor performance to the inability to review the test afterwards. Students indicated that this limited their potential to learn because it did not allow them to determine where they made an error in thinking. The review for them only made them aware of their grades but not what was learned. There are typically two versions of each test according to TRADOC Regulation 350-6 (2005) and the reason for the lack of review so that students do not pass test answers on to the next class. One student described the test review process as the following:

When we go to take our test, any questions we miss, we don't get that question back or find out why we missed it. We don't get a chance to review it at all. They just tell you your percentage and the most missed question and the least missed question. Other than that you don't get to look at the test.

Another student said:

They [instructors] don't have the test, so we just have to remember what we learned and what was on the test. We never find out what questions we missed or anything so what happens is when we were in the study hall and we went over everything and talked abut the test and tried to remember the questions and tried to get the correct answers and we would take the test the next day and it was completely different...

This mode of reviewing the results of the tests is ineffective for students struggling to master the course content. The rationale for doing it in this fashion was never clear.

Teaching Method

The teaching method is typically referred to the way that a teacher chooses to deliver information to their students. Students reported that instructors ranged from very interactive to monotone. The majority of the instructors used a lecture approach with a slide show presentation, such as Power Point, at some point, along with handouts or books. Along with these teaching methods, the course has periodic practical exercises for the student to engage in to further aid in knowledge construction. Although some instructors in this course vary somewhat in their teaching methods, the majority of them share a single teaching perspective. Pratt (2005) refers to it as the transmission perspective. Using the transmission perspective, which is typically lecture format, Pratt (2002) notes, "learners are expected to learn the content in its authorized or legitimate forms, and teachers are expected to take learners systematically through a set of tasks that lead to mastery of the content" (p. 7). Due to the technical nature and structured time limitations of this course, instructors often do not have the complete freedom to use

their preferred choice of teaching method. Using this perspective ensures that all students receive the same information in primarily the same format. Although transmission is the most common teaching perspective he found, Pratt suggests that it does have its limitations. Teachers primarily using the transmission perspective tend to be more focused on the specific content being taught rather than on the learner.

The limitations discussed by Pratt (2002) regarding transmission perspective appeared to have influenced some recycled students in this study. One student responded:

The monotony of class was getting to people. A lot of people were having problems staying awake with "Death by Power Point". Each instructor has his own style which brings a lot to it, but some instructors--their style is more boring than anything else.... There are some instructors that I really appreciate. The instructors that I had back at [X] company were enthusiastic about what they talked about and found different ways of relaying the material to us rather than exactly how it is said in the book.

A second student stated, "When we are in the classroom environment we don't really get into hands-on study until 'all skills' time and the rest of the time we're just reading out of a book."

One student commented on the difficulty of having to adapt to a different teaching style and how it affected his performance:

The problem I had in my last class was one of our teachers got promoted and/or reassigned and so I was getting used to the teaching style and his partner had a different teaching style and so it just messed me up a little bit.

Several students were concerned about the level of detail being taught in the course. Several students stated, "As far as some of the classes that they teach, I know that they need a bit more explanation; I know that they have a time limit to teach but a bit more explanation would be good." and "Some of the instructors...seemed like they expected us to know some of this stuff already."

A third student added:

I think they should go more into detail about the chapters so we can try to be medics. They need to structure it a lot slower so we can learn a whole lot more about what ever is in the chapter.

Students also discussed the inconsistencies that occurred with teaching and how that seemed to affect performance. One student stated, "When a teacher comes in and tells you one thing and another one comes in and tells you something else, you don't know which one to go with, you really don't get it." The teaching methods these students encountered were clearly problematic for them.

Learning Style

Effective teachers understand that students will have different learning styles and can easily adapt their class content in order to allow all students to learn. The learning style is the way that a learner perceives, interacts with and responds to their learning environment (Felder and Brent, 2005). Instructors in this environment, as reported by recycled students, also varied in how they approached different learning styles. Although the opportunity existed for instructors to vary their teaching styles to accommodate different learning styles, many recycled students commented that this did not occur enough for them to grasp the material in the manner that they learned best. Felder (1988) argues that when the instructor's teaching style and the student's learning style do not match, students tend to disengaged from the material and this may contribute to future drop out.

Most recycled students were able to identify their preferred learning styles. Several indicated that they learned better with practical exercises. One student replied, "I learn better with hands-on better than just the book. Visual aids are awesome for me. That's how I learned to be a mechanic by just playing with the pieces." Another indicated, "The practical, hands-on stuff helps a lot. When we did the practicum with a lot of hands-on, we understood it a lot more." A third student said:

I was expecting more field training and more hands-on, doing things as you learn it... that's my biggest problem...when it comes to trying to learn everything through the book, and of course test taking, is not one of my things...

Others expressed their preference for group learning rather than studying alone. A student emphasized:

I do better as far as learning in class instead of trying to learn outside of class on my own...If I have a study group it would be easier in group learning instead of individual learning. On my own trying to read and understand, I don't quite comprehend as well as I could as part of a group trying to go over things. Recycled students appeared to prefer more group interaction and a more hands-on approach overall to facilitate their learning rather than passively listening to lectures or reading individually. They identified the mismatch of learning styles and teaching styles as one factor which influenced their performance. Felder (1988) argued:

A class in which students are always passive is a class in which neither the active experimenter nor the reflective observer can learn effectively (p. 678).

Reteach

Reteach is required to be conducted with all students who fail a test. Reteach is designed to retrain students and provide them the opportunity to become reacquainted with the material from the last failed module. This is done in an effort to clarify any misunderstandings and increase the student's learning of unclear concepts. Reteach training is typically conducted the evening prior to the retest, usually on the same day as the original test. After reteach has been completed, students are retested, typically the following morning during PT time, so as not to interrupt regularly scheduled classes.

Recycled students discussed the reteach process and the impact that it had on their performance. When asking student about the effectiveness of reteach, one student responded:

When we were in the class that night, none of that was on the test the next day. It was just completely different. So I guess what we should have done in that class is not go over what was on the last test, but go over a little of everything in general. Then we would have learned something, maybe, that would have been on the test the next day.

Other students discussed the process of reteach and indicated that, "You can [ask questions], but they tell you what you have to study and all that." For these students the reteaching of the material did not increase their learning.

Study Hall

Study hall in this environment was a time set aside for students, two to three times per week, to study with the assistance of an assigned instructor. The length of the study hall sessions varied between one and one in a half hours. Instructors were assigned to conduct study hall sessions on a rotating basis. Students with course point averages of less than 80% were required to attend these sessions; other students were welcome to attend the session on a space available basis only. Students indicated that instructors varied in how they conducted the session and some were more interactive than others. One student described the study hall process as:

If you have recycled or you haven't quite passed your MOD test it [study hall] is mandatory or if your grade point average is below 80% it is mandatory, and after it is filled up there are no more seats for volunteers.

Another student added:

The study hall really depends upon your instructor/teacher. Some of them will go over things that you have already heard that day. Or they will focus on what their class is struggling in and sometimes it doesn't help.

Several students appeared to be unsatisfied with the amount of time allowed for study hall. One student stated:

In study hall...it takes a while for everything to get organized, so that hour and a half that is allotted is not really the hour in a half that you get. It is more like 45 minutes or an hour. After everyone gets settled down, take their seats, stop talking, roll is called, the teacher gets the right Power Point presentation slides to show, and everyone gets their books out, I have yet to see an hour and a half, maybe an hour.

Students indicated that study hall should be offered more often. One student stated, "If they had study hall offered every night that would be better." Another student agreed, indicating:

I wish it was more like when I failed my one class and I had to go to reteach afterwards. If there was that everyday after class or a couple of days of the week after class, that would be super. I think I would understand more and comprehend more if we had that as an option to go to, I would be there.

On the other hand, other students expressed the opposite feeling toward study hall. They noted that study hall was not effective for them because the sessions covered the most basic material.

It actually is a time waster. I would rather be studying on my own because in study hall, what they would do the day before we had the test, they would have a test review where you choose A, B, C, D or E and they would go over the answers. Sometimes you would get people rattling off in the class and that would be 10 minutes gone in the class. I could be studying on my own with that...I prefer to study on my own. When we first did it [study hall], there were people

who were loud and we would complain that people were being too loud, so there were a lot of distractions.

Another student said:

I have been going to study hall. I don't find that it is beneficial at all. They set up the study hall in the form of the way they do classes, and it's structured a lot. I don't think that it is very helpful for people because they usually go over the material that people should already know....I would rather have, like the other day we had a quiet study time for people who have higher GPAs, where we just sat their quietly and studied on our own and I think that helped me out more, rather than having a structured instruction.

One student said, "I didn't think it was as helpful because we would just review slides that we had already gone over in class."

Students commented on study hall adequately covering the material enough to prepare them for tests. One student stated:

If you went to it [study hall] they would go through the basics of everything you already know, and its stuff that you don't know that you need to study. I knew a lot of the stuff already, and stuff that was harder to grasp; study hall didn't seem to go over it because they are just trying to reteach the whole module in an hour and a half.

Another student responded:

Sometimes you don't know what questions to ask, until you get the test. We would go over...very simple basic things, versus where I am most weak at. Maybe it would have been a little more helpful to cover what I'm not grasping.

A final student indicated:

Some of them [instructors] would ask us questions and then we would answer them. He had a list of questions. Some of them were just talking about the same stuff you heard in class all day. Then you are tired because they are not telling you anything you haven't heard.

The study hall in this course was not responding to the needs of these students who were struggling.

Military Environment

The military environment is demanding and requires students to function as both student and soldier. Students reported that three aspects of the military environment were especially difficult for them: the strict schedule, the shortage of time, and contextual distractions.

Schedule

The schedule is very structured, as there are multiple tasks to complete during the day, with little room for deviation. Time slots are designated for both academic and nonacademic events. Drill Sergeants are responsible for ensuring that the schedule is closely followed as most tasks are scheduled, even time for sleep. Other tasks such as Drill Sergeant Time and miscellaneous formations, formal meeting times to disseminate information, are also conducted before the students move from one location to another;

these take additional time that is not represented on the training schedule (see Appendix F). This occurs consistently for the first four weeks. After students transition into Phases V and higher, time in the evenings and on weekends is less structured which allows for greater personal freedom until the final week when they participate in field training exercises.

Students discussed their experiences while in the military environment and emphasized how the schedule impacted upon their time and how they perceived it to subsequently influence their performance. When asking students to discuss the training schedule, one student described it in the following manner:

Well here there are a lot more responsibilities like Drill Sergeant Time; you have formation time. You don't get to eat when you want because you are on their [Drill Sergeant] schedule and not yours, so studying is hard.

Some students indicated that the training schedule did not allow them enough time to study, even on weekends. Although the schedule is typically open after dinner except for reteach or study hall, students revealed that other tasks were often conducted during that time. One student responded:

On the weekends, the Drill Sergeants want you to do a lot of stuff. Maybe there is a time set aside where they say that you are going to study on Saturdays. Maybe half a day you study and the other time you have for personal time. But if you have a Saturday or Sunday where you have pretty much all day, when you can study, they want you to do all this stuff and finally you get an hour or two at night.

Study Time

Managing available study time is vital for the success of students in this course. Due to the nature of the environment and the inflexible schedule, many of the students stated that they became overwhelmed and found that they had to choose how they would best manage their time. Some students reported that they would perform better if they had more time to study and to process the information, before being expected to take a test. One student said:

Everyone is wide open about the fact that this is straight from the fire hose. It's out there in the open. Yes, it's not easy, but maybe a little bit more time to soak it in at the very beginning; a little more time to adjust.

Another student commented about how the schedule impacted upon study time, especially during Phase 4. One student said, "I didn't feel like I had adequate time to read the material." A second student noted:

We didn't have a whole lot of time to study. A couple times of week we would get maybe an hour or two, but it was pretty intense at the beginning...I think the preferable thing would be to have a little bit more time...to soak that stuff in because it's a pretty big thing when you first get here.

One student provided an example of how other events contributed to why there was not enough time to study.

In [X] Company they gave a lot of counseling statements, which is usually a 500 – 1000 word essay. You have to stay up instead of studying for your test--you have to stay up to write the essay.

Other students felt that they needed extra time with the instructor in order to grasp the material. One student stated:

More time with the instructor to study. The time would definitely have to be monitored....I am sure that if they opened up something like, where if the teacher said come here on Saturday, that would make a difference right there.

One student explained how the lack of sleep influenced performance by saying, "We only get a certain amount of hours of sleep and I am like reading it [the book] and falling asleep, and I don't know how long I was studying for".

The lack of real study time was a huge issue for these students.

Personal Time

Personal time is defined as free time for the student to conduct personal hygiene, make phone calls, prepare for the next day, etc. TRADOC Regulation 350-6 (2005) requires that students are given at least one hour of preparation time each day to take care of personal needs. One student described the difficulty of having limited personal time. He stated, "We'd only have an hour and a half of personal time. It was hard to manage an hour and a half to shower, shave, and get the study time in." Another student emphasized how he managed his limited amount of time.

I figured once I got here that I wasn't going to have any free time as it was and any little free time that I got, I wanted to call back home, so I was like I didn't want to give up any extra time that I got...In [X] Company, when we first got there, we weren't hardly getting any time...Like I said whenever I got my personal time, I wanted to call home. Finally one student described the environment by stating, "You are always surrounded by everyone and you never get that break to go on your own." Time is always a limited resource, but for these students the limits were especially severe and it had a detrimental effect on their learning.

Distractions

Some soldiers pointed out that they felt that the distractions in the barracks contributed to their lower academic performance. Although these students participated in reteach, study hall or both, they still felt that the barracks distracters contributed to their poor performance. When asking student to explain some of the distracters, one student responded, "My surroundings [barracks] do not allow me to study". A second student noted:

There have been some problems in the bay like people getting into fights, screaming at each other and different things...it's just like silly things that have been going on that have been distracting me.

A third student stated:

The first week after living in a new bay is kind of hard, but after that everybody adjusts...like in [X] Company the first day, two girls were yelling at each other and the next week they were the best of friends.

Another student summed it up by saying,

It's hard trying to study up in the bays...it's so hard to try to find somewhere where you are allowed to go study...once we phase, when we have all-day passes, then I can go to Starbucks to study or Barnes and Nobles, wherever I can just sit and study and enjoy it, where it is more peaceful to study.

All these factors within the military environment—the strict schedule, the limits on personal and study time, and situational distractions—worked together to make it difficult for these students to learn the course material and pass the tests. In conjunction with the pedagogical issues, these factors contributed to these students' academic failure.

Discussion

Recycled students in this study identified multiple challenges within the course structure, and they believed these contributed to their poor performance in the course. Factors surrounding both pedagogy and the military environment had a negative influence on the students' performance. The pedagogical challenges included the pace of instruction, test review procedures, learning and teaching styles, and the administration of reteach and study hall procedures. The challenges created by the military environment related to the demanding schedule, limited time to complete required tasks, and situational distractions.

Although research indicates that most students perform best when they are challenged with a brisk pace of instruction, others may find the pace to be overwhelming, causing them to become inattentive or to disengage from learning because they find it difficult to keep up. Students in this study indicated that the pace of the course, especially during Phase 4 (first four weeks) was extremely stressful and they found it difficult to absorb enough information to be successful on the tests. Others

79

believed that the pace of instruction together with having other nonacademic responsibilities contributed to their poor performance.

The inability for students to receive their test back to review the questions created added stress for some students. Because they did not know the specific questions that they missed, they were unable to understand where their educational disconnect occurred or to learn from the process.

Instructors used various teaching methods, but students indicated that they were not geared toward different learning styles. As a result students often found the classes to be tedious and they lost interest in the material. Their disinterest made it difficult for them to comprehend the material sufficiently to be successful on the tests. Students said that they valued enthusiastic teachers who were able to explain concepts clearly. Students expressed their preference for more hands-on and group learning rather than reading and individual studying. They also indicated that some instructors had difficulty explaining the course material in enough detail for them to learn. Clearly there are some pedagogical issues that need to be addressed here.

The instructors' approach to conducting reteach and study hall sessions presented challenges for these students. They commented that the structure of these sessions was not conducive to learning the material required for passing the test. They indicated that although questions were allowed to be asked, the instructor conducted a brief review of previous classes and either did not review the concepts that students felt they needed or did not cover the presented material in enough detail for them to grasp it. This reduced, in their opinion, their ability to pass the tests. Some students valued study hall and

80

emphasized that the sessions were not offered frequently enough and the duration was inadequate to allow for a deeper review of the material. On the other hand, other students commented that study hall was ineffective in helping them learn and preferred to have a quiet place to study instead of the structured, supervised environment.

The military environment created additional challenges for students with the training schedule, barracks distracters, and limits on study and personal time. Because of the strict time schedule, students felt especially stressed at the beginning of the course. These students expressed their frustration when scheduled study and personal time was taken away by Drill Sergeants to perform miscellaneous nonacademic tasks, even on weekends. Students reported that they were sometimes given essays to write as a disciplinary measure and that these were due the next day. When confronted with this situation, they sacrificed their study and/or personal time to complete the assignment rather than face additional punishment. Students also expressed the need for a dedicated daily study time and study area to help them absorb the information required to pass the course. Having a study area would decrease the need to study in the barracks thereby reduce the influence of barracks distractions.

Conclusion

Creating an environment that is conducive to learning is vital for student success. Student feedback on the overall course structure should be used by course administrators to improve the quality of the learning experience. First, a different pedagogical approach may be warranted to enhance student learning and reduce attrition. Providing regular faculty development on adult learning theory and on effective teaching strategies would be useful. It does not appear that faculty is currently offered opportunities for improving their teaching. Cross (1991) discussed five assumptions about teaching: good teaching does make a difference in student learning; teachers differ in what they are trying to accomplish and it is largely based on their academic discipline; effective teachers understand the process of learning and are able to determine what needs to be taught; good teaching is observable, especially by students; and improvement is always needed. Good pedagogy also needs to be supported by changes in the learning environment. A second recommendation would be to create a supervised, but quiet, student study area. This area should be readily accessible daily during available study times; this would alleviate the distractions that occur when students try to study in the bay areas. Finally, while changing the intensity of the training schedule within the military environment may not be an option, limiting unnecessary distractions would improve the overall use of designated student study and personal time and ultimately reduce attrition.

Although all of these students were recommended for recycle, they were able to provide valuable insight on the course structure. Their perception of how the course structure negatively impacted their performance should help administrators view these influences through a different lens. When students are engaged in learning and receive the academic support they need, they tend to be more committed to graduating. As Tinto (1993) notes, when a positive interaction between student learning and the environment occur, students will more likely to be successful in the course. Given the high cost of attrition to the military, in terms of lost resources and reduced numbers of trained personnel, it makes sense that administrators of the Health Care Specialist Course give careful consideration to these recommended changes in course structure and setting. There are many factors that contribute to student attrition but these are the only ones over which they have direct control. It is imperative that administrators take action where they can.

CHAPTER IV

MOTIVATION, EXPECTATIONS, AND SELF-APPRAISAL

When new students have unrealistic expectations of their academic program, they experience significant stress which can greatly influence their motivation level to continue the course. Students entering the Army's Health Care Specialist Course not only must adapt to a different military environment as compared to basic training, but they must also adapt to the learning environment which they are entering. The Army's Health Care Specialist course produces soldiers who are highly trained. The redesign of the course in 2001 increased the scope of practice for medics within the Army. Health Care Specialists are now trained in advanced procedures, making them more capable than previous basic medical personnel. Not only are they "combat medics" but they also are EMT-B trained personnel who can use these skills to treat soldiers in the battlefield or to work in military hospitals. The course is divided into two phases: six weeks of EMT-B and ten weeks of combat medical training.

Health Care Specialist students typically enter the program knowing their strengths and weakness academically, and they have high expectations, both realistic and unrealistic. Recognizing and accepting these strengths and weakness is defined by Sedlacek (2004) as self-appraisal. Bean's (1980) Model of Student Departure serves as a theoretical framework for this study. He stressed the importance of the student's satisfaction with their academic environment. Bean believed that a match between the student's expectations and their actual experiences will likely predict persistent students. In agreement, Levitz, Noel and Richter (1999) indicate, "Expectations are critical; they serve as the point from which students make qualitative judgments about an institution" (p.48).

Students typically research the job and the course by computer or by speaking with peers, recruiters, and/or Drill Sergeants to attempt to get an accurate description of the course. Others don't take this extra step; they possess only the information provided by the recruiter and may not know what to expect when they enter the environment. Often recruiters interacting with the students provide incomplete or inaccurate information. Many recruiters are not medical specialists, have few experiences observing Health Care Specialists in their jobs, and do not understand the complexities of learning in the medical environment. Some recruiters continue to pass on the myth that being a medical specialist is a "laid back" job and thereby giving the student the perception that training will not be challenging. Lowe and Cook (2003) note, "The roots of many unrealistic expectations lie in the inappropriate preparation students receive before coming to the university. There needs to be better communication between students and faculty on expectations" (p. 75). In this case, the recruiter may appear to be the best source of information. A student's expectations may be based on many factors. Some may be realistically based on facts while others may be unrealistically based on myths or stereotypes. Often recruits believe the environment will be similar to that of a college (housing, unstructured study time, less interruptions, etc.) and the course will be easier than college classes; once they get into the course they discover these assumptions were wrong.

Students typically select this military occupational specialty (MOS) because they want a medical career; however, others have no real interest in the job, but selected it either to please family members and significant others or because they thought it would be a faster way than attending college to acquire the skills necessary to get a well-paying job. Further complications ensue if their recruiter told them that they might be able to change their MOS once they got to basic training. The terms of the military contract can rarely be changed prior to completion of the training and in rare cases, when changes were made it was due to the student being unable to complete their training. When this does happen the student is usually reclassified into another MOS. The student could be offered jobs that are less prestigious and require lower aptitude scores than that of a Health Care Specialist, or the student could simply be reassigned by the Army.

It is not uncommon for students to make mistakes about the nature of the job itself. Sometimes the assumption is made that a Health Care Specialist is the same as a Medical Assistant in the civilian sector, that is, someone who works in a hospital setting. When students develop their expectations based on incomplete or inaccurate information, they begin the course only to discover that their expectations and the institutions expectations are incompatible. This typically causes a great deal of stress for the student. Many students are able to adapt to the course and the environment and successfully complete the training, whereas other students emotionally withdraw and are unable to fully engage in the course. When this happens they are recycled, which means that the student is sent to a company that is at an earlier point in the curriculum; this gives them another opportunity to be successful in the course. Rowser (1997) indicated that students are more likely to leave the academic environment if they become frustrated and perceive themselves as failures because they were unable to meet the expectations of the institution. The situation is similar in a military setting.

Faculties at institutions also have expectations of incoming students, and this environment is no exception. The Health Care Specialist course faculty expect: students who are fully capable of completing the course; are motivated to be successful; are responsible for their actions; and are self directed in their learning (Rice, Woods & Bundy, 2004). Tinto (2002) suggested that students perform better when they are in an environment that has high expectations for their learning. He also suggested that students identify and respond to institutional expectations based on how well these expectations coincide with their own. As a result, institutional expectations will not influence all students in the same way.

All students begin programs with particular expectations, motivations, and selfappraisal. The same is true for students in the military, but their context is significantly different from that of a college. In order to understand how these factors impacted students in a military academic environment, this study focuses on a group of students enrolled in a Health Care Specialist course.

Method

Participants

The participants in this study were men and women incoming students to a military basic level job training course in Texas. All participants were within their first six months of military training. The mean age for the participants was 22 years old

87

(SD=4.73) and the ethnicity was as follows: Blacks (7%); Whites (74%); Asian (4%); Hispanic (10%); American Indian (.7%); Other (3%); and No response (.5%). A total of 20 of the participants were interviewed by the researcher for the qualitative portion of the study because of the saturation effect of the received data. The participants were recruited from the Health Care Specialist Course who reported to class on October 16, 2006 (Class 01-07) and December 4, 2006 (Class 03-07). The course is conducted over a 16 week period with the capacity to accommodate approximately 500 students per company. The mission of the course is to provide the Army with Health Care Specialists who would provide basic emergency medical care to the Army. A sample of 288 students participated in this mixed method research study.

Quantitative Measures

The Noncognitive Questionnaire developed by Tracey and Sedlacek (1984) was designed to measure eight noncognitive factors: positive self-concept, realistic selfappraisal, understand and deals with racism, prefers long-range goals to short-term needs, availability of strong support person, successful leadership experience, demonstrated community service and knowledge acquired in a field, which may impact student academic performance and ability (Sedlacek, 2004). After a factor analysis was conducted by the researcher, new factors were used: school and community involvement, prefers long-rang goals, realistic self-appraisal, successful leadership experience and understands and deals with racism. This paper will discuss the selfappraisal factor. The MENAS was designed to explore noncognitive factors which may have influenced the student's academic performance in this course. No previous published research studies have been conducted on the validity and reliability of the MENAS, therefore this study provides some insight on its ability to be used in future studies. The factors identified through exploratory factor analysis using Principal Component Analysis and varimax rotation were labeled as: perceived battle buddy support, perceived stress, motivation to complete the course, perceived unit support, expectations of course, and perceived family support. The internal consistency of the MENAS ranged from .63 to .83. This paper will only discuss the various levels of perceived support: battle buddy, family, and unit support.

Students responded to 46 items using a Likert scale which ranged from either 1 to 5 (strongly agree, agree, neutral, disagree or strongly disagree) or 1 to 2 (yes or no). Multiple questions were written negatively to deal with the student's tendency to answer positively regardless of the content and all items were listed in random order. Students were asked to answer the questions honestly and base their responses on the way they have felt since coming to the medical course.

Qualitative Measures

Individual interviews were conducted with only the recycled participants, using semi-structured, open-ended questions, as they were the focus of the study. All interviews were audio-taped and verbatim transcripts were prepared. Interviews can be used by researchers to observe the emotional impact of specific questions. Participant's nonverbal responses may indicate more of what they mean than the actual words they say (Merriam and Simpson, 2000). All interviews were conducted with the recycled students immediately following their completion of the MENAS and lasted approximately 20 minutes. Prior to the interview session the researcher reviewed the MENAS responses and used those responses to probe for noncognitive factors that may have influenced their academic performance.

Results

Quantitative Analysis

The ratings for each of the passing and recycled student items were subjected to an independent-sample t-test with the alpha level set at .05 to identify those items that distinguish between passing and recycled students. The independent-samples t-test is appropriate whenever two means drawn from independent samples are to be compared (SPSS 12.0). The results from the t-test revealed that there was a significant difference between the two groups in motivation and self-appraisal, however no significant differences were observed in expectations. Although there was no significant difference in expectations between the two groups, 45% of the students indicated that they disagreed or strongly disagreed with the item, "The 68W MOS is what I expected." Among the responses to this question, 23% of the students answered neutral, therefore only 31% of the students indicated that agree or strongly agreed with this item. Fortyseven percent of the students disagreed or strongly disagreed with the item, "This MOS is similar to what the recruiter described to me." Among the responses to this question, 26% of the students answered neutral, therefore, only 26% of the students indicated that they agree or strongly agreed with this item. The reasons for the large number of neutral responses is unknown, however these responses indicated that the expectations of both groups appeared to be unmet. The motivation, self-appraisal and expectation means, standard deviations, t values and significance levels for differences of the two groups are presented in Tables 4.1- 4.3.

MENAS Ouestions	Passing		Recycled			
Motivation	М	SD	М	SD	t	Sig.
18. I am very motivated to pass this course.	4.35	.899	3.69	1.158	3.513	.001*
22. I find this course so difficult that I have given up on trying to pass it.	4.73	.595	3.93	1.314	3.874	.000*
10. I have what it takes to be successful in the course.	4.51	.710	3.98	.950	3.495	.001*
21. <i>I did</i> fail this course on purpose.	4.85	.628	4.17	1.378	3.175	.003*
20. I have <i>considered</i> failing the course on purpose.	3.94	1.407	3.26	1.563	2.852	.011*

Table 4.1. Motivation Items, Means, Standard Deviations, t Values and Significance Levels

 $*p \le 0.05$

MENAS Questions Self-Appraisal	Pas M	ssing SD	Rec M	ycled SD	t	Sig.
8. Likely reason for attrition?	3.19	.984	2.78	.982	3.376	.001*
17. I expect to have a harder time than most students in AIT.	3.78	1.077	3.06	1.251	4.778	.000*

 Table 4.2. Self-Appraisal Items, Means, Standard Deviations, t Values and

 Significance Levels

 $*p \le 0.05$

Table 4.3. Expectation Items, Means, Standard Deviations, t Values and Significance Levels

MENAS Questions	Passing		Recycled			
Expectation	М	SD	Μ	SD	t	Sig.
27. This MOS is very						
similar to what the recruiter	2.81	1 244	2 48	1 194	1 630	104
described to me.	2.01	1.244	2.40	1.174	1.050	.104
26. The 68W MOS is						
what I expected.	2.60	1.289	2.48	1.234	.586	.558

 $*p \le 0.05$

Qualitative Analyses

Qualitative data complemented the quantitative data well when observing student motivation and self-appraisal. Although the groups did not differ significantly in level of expectations, the qualitative data did add clarity as to how it initially influenced their ability to negotiate the course. All of the following data were received from the interviews of recycled students in the course.

Motivation

Sedlacek (1991) and Grayson and Grayson (2003) show that motivation to perform well is the best predictor for persistence. The motivation variable in this study included: giving up on passing the course; indicating not having the skills to pass the course; and failed the course on purpose. The quantitative results under this variable revealed that students who indicated that they had given up on trying the pass the course, and who failed the course on purpose were more likely to be recycled to another unit. Thomas (2002) posits that students are more motivated if they perceived that they were valued in the environment. Motivation was divided into three sections during qualitative analysis: motivation to enter the Army, motivation to continue the course, and the impact of motivation on interest in the course.

When asking recycled students about their motivation to enter the Army, they provided various answers such as: improve finances, get a steady job, money for college, begin a new career, get into the medical field, obtain medical benefits for family, and boredom with current lifestyle. Students had various motivations to want to remain in the course. When asked to discuss those motivations, one student responded: I just start thinking a lot about all of the people who thought I couldn't make it really frustrates me because I didn't want to go out like that. I really wanted to prove those people wrong, that I could do it, and that stresses me out. That also keeps me motivated.

A second student replied:

I don't know why I didn't try harder. I guess it just finally hit me right before I failed. This is your job, you are getting paid to do this, you need to start buckling down, this isn't school anymore. This is how you are going to be making your living.

Another student responded:

My mom and grandmother pushed me and encouraged me. They didn't want to see me [drop out] since I withdrew out of college twice and they don't want to see me try to give up when it gets hard, so they pushed me and pretty much motivated me to stay here. They affected me in a positive way.

On the other hand, some students lost all motivation to continue with the course or were disinterested. One student responded, "I sit down in class, I mean I stay awake, but for me to focus, I am already not interested it so I wouldn't really focus on it." A second student responded:

I was told that I was going to be a nurse and when I got here I was told I was going to be an EMT. When I got here and found this out, things went down hill from there for me. I lost the motivation to be here after that.

Another student responded:

I was like," Do I really want to do it?", but it is better than being a cook or mechanic. So I did it. When I got up to basic I went up to talk to the liaison there and I was trying to get him to help me [change jobs] and he was like we can't do it, you might have a better chance talking with your Drill Sergeants there. But once I got here that didn't work at all. So I kind of knew in my head right away that I really didn't want to do it coming into it.

A final student commented that his inability to adjust to the environment resulted in a lack of motivation to complete the course. The student responded:

After basic training...I've been thinking a lot and I think I know what the problem is. I am realizing that I think I honestly made a mistake coming here. I had a loving family at home, my grandmother offered to pay for full college tuition; I could have gone to college for free. I had everything at home that I could want. I had my friends at home, my family at home and this is the first time I have ever left home. I have just been having way too much trouble with the change and this is the biggest change I have had and it's driving me crazy....The course itself is a good course and it is good for the people who are motivated to be here, but I have just lost the motivation. I'd rather be home going to college, doing something that I really want to do. I just don't know what to do, I just feel trapped here.

Expectations

Helland, Stallings, and Braxton (2002) suggest that students are more likely to remain in school if their expectations are met. In agreement with this, a student replied, "I think that I will have a harder time because this it is not what I expected". A student whose expectations were not met and who originally wanted a different MOS replied:

I guess I didn't expect it to be like basic. I thought I was going to get here and it was going to be a little bit better. I thought, at least I was told, that we were going to have three or four man rooms. PT [physical training] wise, I thought we were going to get to do a little bit more...go out by ourselves and run. Once I got here it was like boom, basic all over again from the start...so right there I was like no, this isn't going to work. Then I would sit down in class...I am already not interested in it, so I wouldn't really focus on it... this isn't for me.

Some students did not initially choose to be medics and would have preferred a different MOS than the one for which they were qualified. One student responded by saying, "I really wanted to join the infantry....The recruiters talked me out of it... I didn't have a high school diploma...I had to take another test at MEPS and that twisted the jobs up and I had to switch jobs." Students indicated that along with wanting a different MOS, and due to their own personal time limitations they felt pressured to select the Health Care Specialist course. Several students also indicated that family members encouraged them to select the Health Care Specialist course, although they personally were not interested. One student said:

I want to change my MOS...My fiancé was like you should try medical. I have no health care experience, so I didn't really want to do that...so he [the recruiter] said you are going to be a health care specialist...a combat medic. So right away I was like, do I really want to do it? But it's better than being a cook or mechanic. So I did it. When I got up to basic I went up to talk to the liaison there and I was trying to get him to help me [change my MOS] and he was like, we can't do it, you might have a better chance talking with your Drill Sergeant here [at AIT]. But once I got here, that didn't work at all. So I kind of a knew in my head, right away that I really didn't want to do it, coming into it.

Students who had previously experienced academic problems in high school or college indicated that they also expected to have difficulty in this environment. One student emphasized:

From the start, even when I was going to college, I didn't even want to go to college...Once I came here and found out that we were going to be in the classroom and it was going to be all of this medical stuff I was like, ah man...I knew this was going to be a struggle for me because I was not looking forward to getting back into the classroom. Coming back into the classroom, already not wanting to do it, then finding out that it was all classroom, I was like (shoulder shrug).

Another said:

I just didn't do well in [high school]. I have been to two different colleges and as far as my classes are concerned, when I try to learn everything at a fast pace, it is really hard for me to grasp and comprehend...but just having to go so fast and taking in so much information and then whatever we have going on outside of the classroom...it's hard to try to comprehend everything. Finally, the expectations that students developed based on information received from their recruiters were mixed. Some of the students, in response to "What were your expectations of this course?" included the following: "He [the recruiter] didn't tell me about any tests"; "My recruiter didn't say much about it, he just said it is fast paced, at times it can be a little challenging and you are just going to have to tough it out when you are going through it."

Student indicated that they did not expect it to be as restrictive as basic training. When comparing it to basic training one student stated:

I expected AIT to be easier compared to basic training. So far I would consider basic way easier than this. It is just not what I expected. I didn't expect a slow pace, because of course...but I just thought it would be more relaxed.

Some students did not know exactly what to expect. One student responded:

I have never been familiar with the medical field so I really didn't know what to expect, but I was watching a couple of TV shows and it looked pretty simple. It was like taking care of people, taking blood pressure, so it looked pretty simple to me and when I got over here and they got really into it, it wasn't what I really expected.

Several of the students indicated that they did not fully understand that the course included being trained as a combat medic or that they might be deployed to assist during the wartime mission. One student indicated, "I only heard the EMT part and not the Whisky [combat medic] part." Another replied, "I have Mike 6 on my contract, which is
a hospital nurse, LPN, so I thought that there was a lot less chance of being deployed as compared to a regular 91W." A final student stated:

It was described to me as a health care specialist. I was told that I would probably get stuck in a clinic or something like that, doing clinicals and watching after the guys coming back from Iraq. I didn't really know that it was a field combat medic. They [recruiters] were just saying medical specialist and not combat field medic...it was the way they portrayed it...

On the other hand, some students expected the combat medic portion of the class and not the EMT portion. They expected the course to have fewer classroom sessions. One student stated:

I thought most of it was going to be out in the field training ...we were going to do hands-on class. To me, I could probably get through that, but I got here and you had to sit in the classroom and it was going so fast I was like (shoulders shrug)...

The largest category of expectations developed by students was focused around the academic environment and living conditions. Students described some of their expectations as the following: "My recruiter explained this MOS a little bit. The barrack situation was a shock to me. We had better barracks in basic training...I was expecting a little more privacy... than in basic training";

I had heard that the barracks had been redone and the duty day was from nine to five, of course you did your PT [physical training] before that, and then go to class...then you have a bunch of free time to do studying and all of that.

A second student replied:

They [recruiter] told me differently. I was told that you had a lot of time to study and it was a little environment with two people per room and you had a little place where you could study. I didn't expect this when I got over here with like eighty people per bay.

A final student indicated:

My recruiter told me that it wasn't going to be like basic training. I thought I was going to get here and it was going to be a little bit better...but once I got here it was like boom, basic all over again from the start and I was like whoa, this isn't for me. So right there I was like no, this isn't going to work. Then I sat down in class, I mean I would stay awake, but for me to focus, I am already not interested in it, so I wouldn't really focus on it.

Although most of the students indicated that they were not given an accurate description of the course, some students revealed that they did know what to expect because they talked to peers who were already in the military. One student said:

At first [the recruiter] told me it would be more like a hospital setting and it wasn't a combat medic to begin with, it was a health care specialist. But with me having a roommate and a couple of friends already in the military they told me, 'Boy, you had better look into what they could cross train you into as far as a field medic,' and I did. I asked my recruiter about it and he let me know. Low and behold when I got here they changed the MOS to a combat medic, so it wasn't anything new to me. One final response which focused on environmental expectations included the following: It's worse than basic...I guess I figured we went through basic and should have learned some discipline and we would come here and be a little more laid back or at least not have the tension here. I got here and the shock...this is like so much. Then going to class, I thought it would be more like going to college class and it's not. I was like, oh no, but I can adjust to this. I was expecting more field training and more hands-on, doing things as you learn it...As far as knowing it was going to be hard and knowing that there was going to be a lot...I knew that. Students developed expectations when in each of these circumstances and each chose to negotiate the environment utilizing the skills that they had available.

Realistic Self-Appraisal

Sedlacek (2004) indicated that students who have a realistic self-appraisal tend to do better in school when compared to those who do not possess this skill and that it is a predictor of success for traditional students. Realistic self-appraisal is defined by Sedlacek as recognizing and accepting strengths and deficiencies, especially academic and working at self development. Zajacova, Lynch, and Espenshade (2005) found that academic self-efficacy consistently predicts persistence and college success. The student's ability to deal with failure was also observed when discussing self-appraisal.

When recycled students participated in the MENAS they elaborated on their response to the original statement, "I expect to have a harder time than most students in AIT" by answering: "Probably because of how I used to be in school, my study habits"; "It is not what I expected. This course is too fast for me"; "I guess my problem with it in the beginning was my confidence level in myself" and "I knew this was going to be a struggle for me because I was not looking forward to getting back into the classroom". Other students responded:

I had this class about three times, when we were in reception. They taught the first six chapters and they taught it in a regular class and gave the test. Then they gave a reteach and I still didn't get it. In class I know the answers and all that, but when it comes to test taking...I am not very good at it. My reading comprehension and how they word everything...I know when the tests get worse and when you get to National Registry, it gets even worse than the other tests. That's where I don't have the confidence.

These responses revealed that these students entered the training with low confidence regarding their abilities to be successful and may or may not have intentionally sabotaged their efforts.

When students were asked to discuss their experiences with failure in the course one student responded:

When I recycled, it was the first time I failed anything that I tried. It's kind of hard. A little bit of depression kicked in and my battle buddies in [X] Company motivated me to continue on to try to pass. I don't like to fail.

A second student stated:

It was really frustrating. I went as far as trying to explain my situation to the main person in charge...I really tried to stay in my company and continue on and get another chance at the test...I was really really frustrated with the fact that I

didn't get that extra shot. Yes, sometimes I do get a little frustrated a bit with failing the course because I know I can do this.

One student discussed their response to treatment they received after failing: Sometimes if you fail something, you're really put down until you're discouraged...sometimes in this environment it's harder because you are under a lot of peer pressure when you fail. Sometimes it just gets you down. It gets you in a downward cycle.

Because of the stress associated with being aware of their personal abilities and then failing to succeed initially in the course, some recycled students indicated that they found it difficult to find the motivation to continue.

Discussion

An examination of the quantitative results of this study showed that there was a difference between the two groups in student motivation and self-concept, yet no difference in expectations. The results, however, also revealed that a larger percentage of the group felt that their expectations were unmet in both areas. The two groups of students in this study also did not differ significantly in terms of receiving their first choice and attrition. This means that regardless of whether the student personally desired the MOS or not, it did not influence whether they would be recycled from the course.

Results of the qualitative data indicated that students who entered the course whose expectations were not consistent with that of the institution and who were not motivated had to determine immediately how they were going to cope. As a result, some students were able to cope with the academic environment whereas others were not. A number of students expected that they would not perform well in this academic environment because of previous classroom experiences in high school or college. These prior experiences may have negatively influenced some students, causing them to enter the environment underestimating their skills. Once students were recycled, some viewed themselves as failures and found it difficult to regain the motivation required to complete the course successfully, whereas others were able to overcome the obstacles and proceed to do well in the course. When all things are equal, the specific variable which predicts why some students who have low levels of motivation, expectation and self-concept continue to be successful in school, while others are not, is not clearly understood.

Although possessing unrealistic expectations did not statistically predict attrition in this study, the results of the qualitative data revealed that it did influence how students initially negotiated the academic environment. Effectively coping with the mismatch of their expectations typically resulted in successful completion of the course, whereas the opposite meant being recycled from the course.

CHAPTER V

PERSONAL RELATIONSHIPS

Research conducted by Beil, Reisen, Zea & Caplan (1999) tested whether academic and social integration and commitment predicted retention for university students. Their results were consistent with that of Tinto's (1993) model which indicated that high levels of student integration into the academic and social community lead to greater institutional commitment which subsequently leads to student retention. Their research was consistent with other studies conducted by Berger & Milem, (1999), Braxton, Milem and Sullivan (2000), Elkins, Braxton and James (2000) and Pascarella and Terenzini (1980) which address social and academic integration. They however, primarily addressed the impact of integration within the college or university environment. The current study diverges from this path and proceeds to investigate how social integration contributes to attrition in a military academic environment. Specifically, it will address the influence of perceived battle buddy support, family support, and unit support during the first six months of the student's training. Recognizing the level of social integration experienced by these students is an important component to understanding their retention patterns.

Previous research studies have not addressed the peculiarities of military training and the value of the support system. The Health Care Specialist Course is vital to the Army and the Army Medical Department (AMEDD). It trains all Health Care Specialist in the Active Army, Army Reserves and Army National Guard, as well as some international students. The mission of the course is to train competent Health Care Specialists in basic emergency medical and routine patient care to treat injured soldiers on the battle field or in military treatment facilities. Retention in the course is vital to ensure adequate levels of medical support to forward units. High rates of attrition not only limit the number of competently trained Health Care Specialists that can be pushed forward, but also waste valuable resources.

Influence of Relationships

Although the military training environment can most often be compared to that of the community college in terms of technical academic training, it is more closely compared to the college/university setting in terms of social interaction. Hoffman, Richmond, Morrow and Salomone (2002) describe the college/university setting as a learning community. Students in a learning community are frequently engaged in campus activities, take the same courses, and study the same materials. Students in this learning community also spend large amounts of time together and are more likely to create lasting relationships which often extend beyond the classroom into social arenas. Astin (1999) indicated that living in a campus environment significantly enhanced retention because the students were better able to become involved in campus activities. A student interacting with other individuals on campus as a way to become a member of that environment is termed social integration (Tinto, 1975).

Battle Buddy System

The military learning community is very similar to the learning community discussed by Hoffman et al. (2002). Many of the students in the Health Care Specialist course attended basic training together and some had already a lasting relationship with

each other. Students who attend the same course rotation are typically placed in the same company/unit. Members of the same company sleep, eat, train, and study together. This closeness allows students to develop deep support systems with each other.

Fass and Tubman (2002) conducted a study which focused on determining if peer and family attachment predicted academic success. They found that students who had a strong attachment to parents and peers demonstrated a higher level of self-esteem and sense of self. Peers provide each other support and provide feedback in various ways. When students are accepted into peer groups they feel a sense of belonging and tend to be more persistent in school. Astin (1984) and Bean (1980, 1983) identified peer support as an important aspect of retention. They argue that rather than faculty influence, it is peer relationships and the participation in peer groups that have a greater influence on student's overall attitudes towards school. Ultimately, these relationships shape the student's social environments.

The Army recognizes the importance of peer support. It has created the battle buddy system which is designed to assist soldiers in developing responsibility, initiative, and dependability (TRADOC Regulation 350-6, 2005). Leaders introduce the battle buddy concept at the beginning of the training cycle. During this time soldiers are placed in two-person teams with emphasis on: providing the soldier an immediate peer support system; encouraging teamwork through peer relationships; and ensuring peers are accounted for and safe during training to decrease the potential for serious incidents, such as sexual harassment, misconduct, or attempted suicide. This system is continued throughout training until soldiers are sent to their permanent duty stations. One of the policies of the system is that team changes should be minimized. The significance of the battle buddy system has not only become a successful program for the Army but also has become valuable to the students. The supervisory chain of command has the benefit of having an early alert system which notifies them of potential problems, while the soldiers have another source of support that is available to them. Nora (1987) found that encouragement from others greatly influenced the student's social integration which positively impacted retention.

If students fail a test and are recycled to another company, they are immediately transferred to a follow-on unit to be given an additional opportunity to successfully complete the course. When this occurs, however, they lose their original battle buddy because they must physically relocate to another barracks where they are assigned to another battle buddy or added to another already established battle buddy team. Changing battle buddies during one of the most stressful period in the soldiers training can negatively affect the student's performance and overall motivation to continue and complete the course.

Family Support Issues

Family support is typically the first source of support that most students receive outside of peer relationships. Holahan, Valentiner and Moos (1994) indicate that a high level of parental support is associated with better psychological adjustment for students. Finn and Rock (1997) emphasized that positive parental support promotes higher grade point averages, general academic attainment, cognitive engagement, and academic persistence among students. In contrast, low levels of attachment to parents have been identified as a potential risk factor for poor academic performance. Poor parent-child communication or relationships (Finn, 1989) and low educational expectations or encouragement of children (Dornbusch, Ritter, Leiderman, Roberts, & Fraleigh, 1987) have been identified as placing students at risk for poor academic outcomes.

The availability of a strong support person, whether from family, friends, peers or academic faculty, has been determined to be a predictor of college student retention (Astin, 1999). Sedlacek (2004) notes that having a strong support person has also been shown to be a "significant correlate of grades, retention and graduation for African Americans, women, athletes of all races, and students in special support programs" (p. 46). Tinto (1993) indicated that students often need to break away from family and friends back home when coming to college to assist with their institutional integration. Guiffrida (2005), however, conducted a study which focused on African American undergraduate students to uncover patterns in their perception of family relationships. This study challenged Tinto's stated assumption. He notes that African American students with a strong family support tend to persist in school; however, he recognized that the support received from their family and friends from home could be both positive and negative in terms of helping them integrate into the academic environment. If the student viewed their relationship with family and friends from home as encouraging and not as a dependent relationship, they tended to do better in school.

Carney-Crompton and Tan (2002) conducted a study which showed that nontraditional students (learners over the age of 25) had more stress and fewer sources of support, yet performed at a higher academic level than did traditional students. They also observed that traditional students tended to be more reliant upon family support (parents, grandparents) which nontraditional students relied more on their spouse and/or non-family sources.

Unit Support/Institutional Integration

The perceived support provided by the institution is necessary for institutional integration. Hoffman, Richmond, Morrow and Salomone (2002) conducted a study in which they were developing, testing, and refining a "sense of belonging" instrument. They were interested in studying persistence and retention characteristics of incoming university freshmen and the impact of "sense of belonging." They found that the greater "sense of belonging" that students had to the institution, the greater their commitment to finishing their education. The authors argued that a "sense of belonging" is directly linked to social support, which they defined as having developed sufficient social networks, and is the opposite of loneliness, which they defined as failure to connect with others. They also found that a learning community contributes to creating a feeling of being cared about, which had a positive impact on the student's performance in the environment. Students in this study also reported having higher levels of peer and faculty support, as well as experiencing lower levels of isolation.

Two theoretical models help us understand why students persist in college or dropout. Tinto's (1975) Model of Student Persistence showed the importance of social and academic integration and commitment to remain in college. His theoretical approach is rooted in Durkheim's (1961) Theory of Suicide which posits that suicide occurs when there is a lack of societal integration. Tinto used this analogy to emphasize that attrition occurs when there is a lack of academic and social integration. He emphasized the necessity of these two factors to establish adequate levels of goal and institutional commitment and argues that a deficiency in either factor may potentially lead to attrition. Academic integration as defined by Tinto consists of grade performance and intellectual development High levels of these factors influence students to develop a commitment to finish school. Social integration consists of peer group and faculty relationships. High levels of these factors influence the students' institutional commitment which influences their decision to remain in school. Tinto emphasizes that these commitments are influenced by the student's family background, individual characteristics, and past educational experiences. Tinto (2002) underlined four conditions that if met contribute to student retention: first, the consistency of the student's expectations in comparison to the institution's expectations; second, the academic support (study groups, tutoring, etc.) and social support (mentoring, counseling, peers, family, etc.) received; third, the overall involvement in the institutional environment (academic and social integration); and fourth, that actual learning is taking place. Tinto (1975) states, "Given individual characteristics, prior experiences and commitments, the model argues that it is the individual's integration into the academic and social systems of the college that most directly relates to his [or her] continuance in that college" (p. 96). This study primarily addressed the second and third conditions that were emphasized by Tinto.

The second model is offered by Bean (1980). His model of Student Departure stresses the importance of the student's satisfaction with their institutional commitment.

He suggested that expectations + actual experience= either retention or attrition. His model is based on theories of student socialization. Bean argued that the student's level of institutional commitment greatly influences their decision to remain or leave the academic environment. He constructed a conceptual model of dropout syndrome which included factors related to academics (academic performance, academic integration), social-psychological concerns (goals, social life, alienation), and environment (finances, peers, opportunity to transfer). He contends that these three factors are expected to influence socialization factors which include: college grades, institutional fit, and institutional commitment. Of the three socialization factors, Bean believes that institutional commitment is the most important since it directly affects drop-out syndrome.

Method

Participants

The participants in this study were men and women incoming students to a military basic level job training course in Texas. All participants were within their first six months of military training. The mean age for the participants was 22 years old (SD=4.73) and the ethnicity was as follows: Blacks (7%); Whites (74%); Asian (4%); Hispanic (10%); American Indian (.7%); and Other (3%). A total of 20 of the participants were interviewed by the researcher for the qualitative portion of the study. The participants were recruited from class cycles 01-07 and 03-07 of the Health Care Specialist Course. The course is conducted over a 16 week period with the capacity to accommodate approximately 500 students per company. The mission of the course is to

provide the Army with Health Care Specialists who would provide basic emergency medical care to the Army. A sample of 288 students participated in the mixed method research study.

Quantitative Measures

The MENAS was designed to explore noncognitive factors which may have influenced the student's academic performance in this course. No previous published research studies have been conducted on the validity and reliability of the MENAS, therefore this study will provide some insight on its ability to be used in future studies. The factors identified through exploratory factor analysis using Principal Component Analysis and varimax rotation were labeled as: perceived battle buddy support, perceived stress, motivation to complete the course, perceived unit support, expectations of course, and perceived family support. The internal consistency of the MENAS ranged from .63 to .83. This paper will also only discuss the various levels of perceived support: battle buddy, family, and unit support.

Students responded to 46 items using a Likert scale which ranged from either 1to 5 (strongly agree, agree, neutral, disagree or strongly disagree) or 1 to 2 (yes or no). Multiple questions were written negatively to deal with the student's tendency to answer positively regardless of the content and all items were listed in random order. Students were asked to answer the questions honestly and base their responses on the way they have felt since coming to the medical course.

Qualitative Measures

Individual interviews were conducted with only recycled participants using semistructured, open-ended questions. All interviewed were audio-taped and transcribed. Interviews enable the researcher to observe the emotional impact of specific questions. Participant's nonverbal responses may indicate more of what they mean than the actual words they say (Merriam and Simpson, 2000). All interviews were conducted with the recycled students immediately following their completion of the MENAS and lasted approximately 20 minutes. Prior to the interview session the researcher reviewed the MENAS responses and used those responses to probe for noncognitive factors that may have influenced their academic performance.

Results

Quantitative Analyses

The ratings for each of the passing and recycled student items were subjected to an independent-sample t-test with the alpha level set at .05 to identify those items that distinguish between passing and recycled students. The results from the t-test reflected a statistically significant difference between the passing students and the recycled students in perceived battle buddy support and unit support. The battle buddy support item stated, "I can approach my battle buddy to talk to him/her about personal matters and/or problems in my life" and received a significance value of .011 with a mean of 3.80 for passing students and 3.29 for recycled students. The unit support item stated, "There are other people, besides my battle buddy, that I can turn to for help and support here" and received a significance value of .050 with a mean of 1.69 for passing students and 1.52 for recycled students. The results from the t-test, however, did not find a statistical difference between the two groups of students in their level of perceived family support. The perceived battle buddy support, family support, and unit support mean and standard deviations for the two groups, t values and significance levels for differences between the mean are presented in Tables 5.1-5.3.

MENAS Questions Battle Buddy Support	Passir M	ng SD	Recycl M	led SD	t	Sig.
						0
34. I can approach my battle buddy to talk to him/her about personal matters and/or problems in my life.	3.80	1.240	3.29	1.111	2.544	.011*
38. I perceive my relation- ship with my battle buddy to be "close."	3.83	1.107	3.55	1.173	1.533	.126
35. My battle buddy helps me and is supportive of me in this course.	4.04	.974	3.81	.943	1.956	.147
37. I often talk to my battle buddy about my personal or academic problems.	1.59	.501	1.64	.485	641	.522
15. I have people to talk to about my problems and/or stress in my life.	3.60	1.193	3.24	1.226	1.817	.070

 Table 5.1. Battle Buddy Support Items, Means, Standard Deviations, t Values, and
 Significance Levels

 $*p \le 0.05$

MENAS Questions	Pass	sing	Recy	cled		
Family Support	М	SD	М	SD	t	Sig.
32. My family wants me to find a way out of this course and/or get out of the Army.	4.19	1.124	3.98	1.137	1.122	.263
31. My family or people who are close to me are supportive of me and my role as a combat medic.	4.22	1.038	3.95	1.058	1.537	.125

Table 5.2. Family Support Items, Means, Standard Deviations, t Values, and Significance Levels

 $*p \le 0.05$

Table 5.3. Unit Support Items, Means, Standard Deviations, t Values, and
Significance Levels

MENAS Questions	Pass	sing	Recycled			
Unit Support	М	SD	Μ	SD	t	Sig.
36. There are other people, besides my battle buddy that I can turn to for help and						
support here.	1.69	.497	1.52	.594	1.956	.050*
44. I have bonded well with my unit.	3.50	1.120	3.19	1.174	1.665	.097
45. I have a strong sense of belonging here.	3.33	1.132	3.05	1.268	1.443	.150
39. My unit is supportive of me.	3.30	1.180	3.50	.994	-1.054	.293

 $*p \le 0.05$

Qualitative Analysis

The purpose of the qualitative research was to explore how noncognitive factors work together to contribute to attrition in a military academic environment. The theme of personal relationships emerged as important in preventing attrition. The characteristics of the student's personal relationships with their battle buddy, their family, and the institution shed some light onto how valuable these relationships were to the student's success in an academic program.

Following the guidance of Merriam and Simpson (2000), the researcher proceeded by recording data as it was received and placing it into categories which best represented the topics revealed. After continuous recategorization and recoding, similar themes and hypotheses began to emerge. Data were organized in a meaningful manner to allow the researcher to get a global view and attempt to understand how various factors influenced attrition. Merriam (1998) states, "When categories and their properties are reduced and refined and then linked together by tentative hypotheses, the analysis is moving toward the development of a theory to explain the data's meaning" (p. 192). The constant comparative method contributed to allowing the researcher to generate a theory that was grounded in the data received from students who were unsuccessful at completing their assigned course within the first six months of training.

Unit Support

The only statistically significant difference between passing and recycled students in this study was the level of perceived unit support. The theme of personal relationships that emerged from the qualitative analysis supported this finding. Sedlacek (2004) emphasized the importance of faculty support and its impact upon student retention. A study conducted by Sheldon (2003) indicated that students who perceived an adequate level of faculty support were more likely to persist in a nursing program. Tinto (2000) also found that students tended to persist if they were provided with academic, personal, and social support. Recycled students identified that once they recycled to another company they perceived the new unit as unsupportive. When asked if they felt supported by their new unit, one student replied, "I felt very stressed. I felt alienated; I didn't know anyone in the class. I just sat in the corner." Another responded:

I have not bonded nearly as well with this unit as I had with [X] Company. I mean I don't really feel the connection that I had with the people in [X] Company. I mean I felt connected to the Drill Sergeant, I felt connected to every member of my platoon. I knew practically every one in my platoon and people all over the company who were good friends, not as acquaintances but actually as good friends that I would go hang out with at the River Walk or something like that. They were a lot closer to me than anyone here in [X] Company. I don't know anyone here.

Several students indicated that they were immediately treated differently and negatively labeled when assigned to the new unit. They responded: "We were known as the Echo guys and we were going to be trouble". Another responded,

It's hard with my new Drill Sergeant. The first day we were here he told us, "I don't like kids from [X] Company, they have always been trouble and I don't

like them...when they assign them to my platoon it screws everything up," so I am like "Great!"

The latter student indicated that they immediately felt discouraged and slighted just because they came from a different company. Another student discussed how they were treated after being recycled. "I guess it is more of just people hazing me, [saying] 'I can't believe you double tapped' [failed a test twice]. Sometimes it hits me." A different student expressed the following:

I personally don't see any reason to burden other people with your problems. If someone is getting paid for it that fine but ...I have seen the chaplain on stage, but I don't know if I would talk to him about my problems.

Battle Buddy Support

The majority of the responses from both groups indicated that they were close to their battle buddy and that they could talk to that person when they had problems. These relationships were more likely to help them deal with the demands of the course. Students indicated that they valued the battle buddy relationship that they have developed, many since basic training. The personal bonds that they established aided in their ability to progress through the training period. These personal relationships allowed them to perform activities, such as study together as a team, participate in activities, or provide emotional support. For recycled students, having a close relationship with their battle buddy provided that immediate support they needed to get refocused in order to continue with the course. Recycled students, however, indicated that their support system was severely damaged when they recycled to another company. Recycling forced them to leave their battle buddy in their original unit and to be assigned another battle buddy at their new unit. Often these new relationships didn't work well. One recycled student stated, "There were also problems especially when you have been with people for a couple of months in reception and basic and then leaving is harder than most people would probably think." Several other students responded:

The issue that I am having with that [recycling] is that I know a lot of people over there and have had a lot of heart-to-heart discussions with people over there and they are more like family to me and I just did not want to leave there...That is one of the major stressors that I am having right now;

Being over here we don't really have any battle buddies. I mean there are some people that I would go talk to that still live over there [X Company]. I just got moved over here last night and my previous battle buddy still lives over there.

A final student indicated:

It's a new company for me and a lot of people that I went to basic with, all of them are in...[X] Company. A lot of them were really close friends, and in the new company I barely even know anybody.

Some students stated that they have lasting battle buddy relationship and value their support. One student stated:

My friends help me out a lot. I have four friends here. We have been together since reception at basic, so we have all been helping each other out a lot. We are all so close. We have spent the last 4 months together.

Another stated:

I don't want to leave my battle buddies, the guys I went to basic with. These are the guys that I actually trust, more than the guys back home. These are the guys I'm willing to take a bullet for. I just don't want to leave and that's pretty much it.

On the other hand, some students did not feel comfortable talking to their battle buddy about their personal problems. One student said:

I feel awkward going up to [my battle buddy] to talk to them about my problems, especially if they have their own problems. If you are observant you can actually see the problems that others are having. It's like, why burden someone else down with your problems? I mean I have listened to others and let them vent and everything because it makes them feel a little bit better. As for me, I am just like a duck and I let it roll off my feathers. I mean it just helps them out with being able to let it go, so why bother them with your issues because you don't know how they are going to handle it.

Family Support

When focusing on the level of perceived family support, students indicated that the support they received was adequate from the people they felt were the closest to them. When asked if their family was supportive of them and their role in the military one respondent answered, "Actually, my mom supports me 100% in what I do regardless of what it is, but it doesn't necessarily mean that she is happy with it". Another student responded, "My wife was totally for me going into the military. She wanted to just get the family stabled out. My mom and dad were not too thrilled about it." On the other hand, a few students indicated that although they now have family support, this was not true with the initial decision to join the military. One student replied:

When I called my dad from MEPS [Military Entrance Processing Station], he wouldn't talk to me for two months afterwards. My cousins, my uncles, and my aunts said we don't believe you should join the Army because of the whole war in Iraq. They would watch the news and see how many people died per month and they would say you can't go to the Army. After a while they learned that I would be okay and they said they really supported me.

A second student answered:

My sister is [supportive]...my dad, sometimes he is. There are some people who are like, you should have stayed home. Some of my friends are like, what are you doing and I tell them the course that I am taking is to be an EMT, and they are like well, you could have done that at home.

Another student responded this family support continues to be inadequate:

I haven't really been too successful at school, so I have burned some bridges in my family regarding that. It was my fault, so my family is not too supportive. They would rather see me do something else and stick with it and retire and I am not ready to do that, so I guess that's where we disagree. Finally, some students acknowledged that they have family support, however their family does not understand the system or can not help them when they have problems. One student stated:

I feel that I have someone that I can talk to, but I don't feel that they can help me. Maybe I can get some emotional help. I have my sister and my mom that I can talk to and I have grandparents who are old, but they are sickly.

Another student answered:

I don't really have anyone I can talk to about my personal problems. I don't really like to open up....My mom didn't even graduate high school and my gramps, she didn't graduate high school. My uncle is a tanker so he doesn't know about this so. I don't think they would understand.

Most of the recycled students indicated that they had adequate family support.

Discussion

Quantitatively, the differences found in this study between passing students and recycled students suggest that recycled students perceived that their battle buddy support and their unit support was insufficient after they recycled. The t-test failed to reveal any statistically significant difference observed between the two groups with respect to perceived family support. The students in this environment, although they may have indicated that they had adequate family, needed to know that there was someone local, in their new unit, that they could turn to for support to address their feeling of being "slighted" or being viewed as "trouble makers" when they entered the new unit. Possessing a "sense of belonging" to the unit, as Hoffman, Richmond, Morrow and Salomone (2002) note, contributes to creating a feeling of being cared about and feeling less isolated, which greatly impacts upon their performance in the environment.

The qualitative data not only complemented the quantitative data, but also contributed greatly to further understanding how relationships interact to influence attrition. Previous studies (Astin, 1999; Finn and Rock, 1997; Guiffrida, 2005; Sedlacek 2004 &; Tinto, 1993) indicated that a high level of family support contributed to retention. Students in this study reported that they received various levels of family support for their decision to enter the military. They indicated that although they received good family support, they also understood that many of their family members could not personally relate to their experiences in the military environment because of their lack of military experience. This indication reemphasized the need to have a good level of unit support from those who are familiar with the environment i.e., Chaplain, instructors and/or Dill Sergeants. Both groups indicated they had some problems with family members who were not supportive, however, they all indicated that the support they received from the people they felt were closest to them was adequate. This perceived family support provided the extra encouragement the student needed to remain motivated to continue with the course.

Similar results were also true for both groups regarding battle buddy support. The value of having a battle buddy in this environment greatly improved the student's perception of institutional integration. The majority of the responses from both groups indicated that they were close to their battle buddy and that they could talk to them when they had problems, and that this relationship enabled them to handle the demands of the

course more effectively. Recycling students, however, indicated that they lost this valuable support system. No longer did they have someone close to confide in; instead they were required to attempt to recreate this bond with a new student. Many recycling students indicated that starting this process over again and being thought of as an outsider was too overwhelming, and they would rather not create a deep bond with their new peers. Although changing units for recycling students may not be avoided, providing the additional unit support may be necessary in order to give students the necessary motivation to complete the course with their new company successfully.

Conclusion

In conclusion, the findings in this study support the published literature in regards to the value of various relationships and academic and institutional integration. The relationships discussed in this study, although termed differently, are not particularly unique to this academic environment. The possession of adequate levels of battle buddy, family, and unit support appears to be essential for most students in this environment in order to be academically successful. Although the course administrators do not have an influence on the level of family support that students receive, they do have the ability to influence battle buddy and unit support. Being aware of the value of these relationships gives the course administrators the opportunity to create an environment that cultivates support, identify when these relationships are deficient, and provide ways to address these deficiencies.

CHAPTER VI

SUMMARY AND DISCUSSION

The purpose of this explanatory mixed methods research study was to understand how noncognitive factors contributed to attrition in the Health Care Specialist course. Three instruments were used to analyze data: Modified Noncognitive Questionnaire (NCQ), the MENAS and a face-to-face audio recorded interview. The Modified NCQ was designed to measure noncognitive factors which impacted student academic performance and ability and was conducted prior to the beginning of the course. The MENAS was designed to further explore noncognitive factors which may have impacted upon their academic performance after they started the course. Interviews were designed to allow the recycled students the opportunity to further elaborate on their perceptions of how noncognitive variables impacted their performance in the course and they were conducted after the student was recycled to another unit to be given another opportunity to complete the course. After factor analysis and reliability testing was conducted on both the Modified Noncognitive Questionnaire (NCQ) and the MENAS, the researcher identified a total of 11 noncognitive variables to be further analyzed. Two cognitive variables from the Armed Services Vocational Aptitude Battery (ASVAB) and data used from face-to-face audio recorded interviews were also further analyzed. The noncognitive variables included in this study were: school and community involvement, preference for long-term goals, realistic self-appraisal, leadership experience, ability to handle racism, perceived battle buddy support, ability to handle stress, motivation, perceived unit support, expectations, and perceived family support. The cognitive

variables were: skilled technical (ST) and general technical (GT) scores. The themes from the qualitative data collected from the interviews were: environmental structure, personal relationships and student expectations. This chapter presents a description of the sample, a review of the findings, discussion of the findings and limitations, as well as implications and recommendations.

Findings

In this study, passing students were defined as those who were able to successfully complete the training course with their original unit, whereas, recycled students were defined as those who were unable to successfully complete the course with their original unit and were recycled/transferred into another unit in order to be provided an additional opportunity for success. For this study 352 (81%) students were defined as passing and 82 (19%) were identified as recycled students. Table 6.1 provides a summary of the means and standard deviations for the five independent variables (school/community involvement, preference for long-term goals, realistic self-appraisal, leadership experience and successfully handles the system-racism), with a dependent variable of attrition, for both groups, utilizing the Modified NCQ. Table 6.2 provides a summary of the means and standard deviations for the six independent variables (perceived battle buddy support, stress, motivation, perceived unit support, expectations and perceived family support), with the dependent variable of attrition, for both groups, utilizing the MeNAS.

Modified NCQ	Passing (N M	N = 352) SD	Recycled (N M	N = 82) SD
School/community involvement	3.52	2.39	3.31	2.74
Goals	4.66	1.44	4.25	1.23
Realistic self-appraisal	6.97	2.06	5.84	2.23
Leadership	4.84	1.80	5.16	1.87
Racism	3.94	1.86	3.74	1.64

 Table 6.1. Modified NCQ Descriptive Statistics

MENAS	Passing (N M	l = 246) SD	Recycled (N M	N = 42) SD
Battle Buddy Support	16.62	5.05	13.24	4.59
Stress	28.43	8.46	26.77	6.51
Motivation	30.37	5.59	26.77	8.00
Unit Support	10.13	3.43	9.74	3.44
Expectations	5.41	2.53	4.96	2.43
Family Support	8.41	2.16	7.93	2.20

Table 6.2. MENAS Descriptive Statistics

An independent-samples t-test with an alpha level set at .05 was conducted to compare the two group's (passing and recycled) means when analyzing both surveys. George and Mallery (2006) indicated that the t-test is the most commonly used method to evaluate the differences in means between two groups. This analysis was also important to compare group differences with the understanding that there was no chance of overlap within the group memberships.

The results from the t-test reflected a statistically significant difference between the passing students and the recycled students in several areas within the Modified NCQ and the MENAS. Items from the Modified NCQ which were labeled preference for longterm goals, realistic self-appraisal and successfully handles racism, showed a statistically significant difference between the two groups. Table 6.3 provides a summary of the means, standard deviations, t Values, and significance levels for the Modified NCQ. Items from the MENAS which were labeled perceived unit support, able to cope with stress and motivation to continue the course, showed a statistically significant difference between the two groups. Table 6.4 provides a summary of the means, standard deviations, t Values, and significance levels for the MAS.

Modified NCQ Questions	Pas M	sing SD	Recy M	ycled SD	t	Sig.
<i>Long-term goals</i> Mean of 7.1B, and 7.2B.	2.27	.558	2.10	.580	2.441	.038*
Mean of 7.1A, 7.2A, and 7. 3	2.39	.578	2.15	.650	3.303	.305
<i>Realistic self-appraisal</i> 8. Likely reason for attrition.	3.19	.984	2.78	.982	3.376	.578
17. I expect to have a harder time than most students in AIT.	3.78	1.077	3.06	1.251	4.778	.023*
Successfully handles racism 26. I want a chance to prove myself academically.	2.30	1.086	2.00	.889	2.618	.004*
25. If course tutoring is made available to me, I would attend regularly.	1.64	.772	1.74	.750	-1.141	.321

Table 6.3. Modified NCQ Items, Means, Standard Deviations, t Values, and Significance Levels

 $*p \le 0.05$

MENAS Questions	Pass	sing	Recy	Recycled		
-	Μ	SD	М	SD	t	Sig.
Perceived unit support 36. There are other people, besides my battle buddy that I can turn to for help and support here.	1.69	.497	1.52	.594	1.956	.050*
44. I have bonded well with my unit.	3.50	1.120	3.19	1.174	1.665	.097
45. I have a strong sense of belonging here.	3.33	1.132	3.05	1.268	1.443	.150
39. My unit is supportive of me.	3.30	1.180	3.50	.994	-1.054	.293
<i>Stress</i> 14. My stress level affects my academic performance.	2.60	1.097	2.48	1.131	.682	.848
25. My personal problems affect my academic performance.	2.91	1.202	2.52	1.065	1.958	.224
17. I have trouble concentrating which affects my academic performance.	3.18	1.04	2.83	.986	2.029	.363
16. I am coping with and managing my stress well.	3.67	1.007	3.31	1.047	2.136	.689
43. I am having a difficult time dealing with failure.	3.19	1.387	2.33	1.141	4.334	.009*
1. I attribute my problems with the course to:	3.64	1.421	2.67	1.141	4.908	.000*

 Table 6.4. MENAS Items, Means, Standard Deviations, t Values, and Significance

 Levels

Table 6.4. Continued

MENAS Questions	Dessing		Recycled			
MENTIS Questions	M	SD	M	SD	t	Sig.
<i>Motivation</i> 18. I am very motivated to pass this course.	4.35	.899	3.69	1.158	3.513	.001*
19. I am doing my best to pass the course.	4.26	.781	4.17	.935	.696	.164
22. I find this course so difficult that I have given up on trying to pass it.	4.73	.595	3.93	1.314	3.874	.000*
10. I have what it takes to be successful in the course.	4.51	.710	3.98	.950	3.495	.001*
21. <i>I did</i> fail this course on purpose.	4.85	.628	4.17	1.378	3.175	.003*
20. I have <i>considered</i> failing the course on purpose.	3.94	1.407	3.26	1.563	2.852	.005*
11. My grades and my academic performance are my responsibility.	3.73	.574	3.57	.703	1.364	.179

 $*p \leq 0.05$

Research question 1

What noncognitive factors explain attrition among Health Care Specialists students during AIT?

A forward logistic regression analysis was conducted to determine how well the five noncognitive variables from the Modified NCQ predicted attrition of students within their first six months of training. Logistic regression is a method used for determining the relationship between predictor variables and a dichotomously coded dependent variable (George and Mallery 2006). The predictor variables were school/community involvement, preference for long-term goals, realistic self-appraisal, leadership experience and successfully handling the system (racism). The dependent variables were predictive: realistic self-appraisal and preference of long-term goals (See Table 6.5). These variables revealed a negative correlation to attrition, indicating that the lower the realistic self-appraisal score and the lower the preference to long-term goals score, the more likely the student would be recycled to another unit. This model, using the two variables, correctly predicted attrition in the data 81.6% of the time (See Table 6.6).

Modified NCQ	В	S.E.	Wald	Sig.	Exp(B)	
Long-term goals	540	.215	6.331	.012*	.583	
Self-Appraisal	478	.107	20.062	.000*	.620	
Constant	1.412	.573	6.069	.014*	4.105	
* <i>p</i> ≤ 0.05						

 Table 6.5. Modified NCQ Logistic Regression Variables in the Equation

Table 6.6. N	Modified N	NCQ L	ogistic 1	Regression
--------------	-------------------	-------	-----------	------------

	Observed		Attrition	Predicted	Percentage Correct
			Pass	Fail	
Step 2	Attrition	Pass	351	1	99.7
		Fail	79	3	3.7
Overall Percentage					81.6

Again, a forward logistic regression analysis was conducted to determine how well the six noncognitive variables from the MENAS predicted attrition of students within their first six months of training. The predictor variables were: perceived battle buddy support, able to cope with stress, motivation to continue the course, perceived unit support, compatible expectations of the course and perceived family support. The dependent variable was attrition. Results of the regression analysis indicated that four variables were predictive: perceived battle buddy support, perceived unit support, able to cope with stress, and motivation to continue the course (See Table 6.7). These variables
revealed that students: who did not perceive that their battle buddy support was adequate; who did not perceive their unit support was adequate; who demonstrated a high level of perceived stress (high level of stress, indicated difficulty dealing with failure, attributed both academic and nonacademic problems to stress); and who lacked the motivation to complete the course (given up on the course, intentionally failed course, low self confidence) were found to be more likely to recycle to another unit. This model, using the four variables, correctly predicted attrition in the data 91.3% of the time (See Table 6.8).

MENAS	В	S.E.	Wald	Sig.	Exp(B)
Battle Buddy Support					
bbsupport3	1.369	.536	6.513	.011*	3.932
support	506	.214	5.575	.018*	.603
Unit Support					
support 2	-1.107	.470	5.545	.019*	.331
unit support	.690	.224	9.464	.002*	1.993
Stress					
stress	.594	.210	7.967	.005*	1.811
failure	585	.186	9.926	.002*	.557
self-appraisal	411	.152	7.352	.007*	.663
given up	785	.268	8.566	.003*	.456
self concept	552	.270	4.173	.041*	.576
Motivation					
did fail	528	.201	6.941	.008*	.589
Constant	6.720	1.614	17.337	.000*	828.949

 Table 6.7. MENAS Logistic Regression Variables in the Equation

 $*p \le 0.05$

	Observed		Attrition	Predicted	Percentage Correct	
Step 10	Attrition	Pass Fail	Pass 243 22	Fail 3 20	98.8 47.6	
Overall Percentage					91.3	

Table 6.8. MENAS Logistic Regression

Research question 2

How do noncognitive factors, when combined with cognitive test scores, serve as a predictor of academic success in the military setting?

A forward logistic regression analysis was conducted to determine how well the two cognitive variables predicted attrition of students within their first six months of training. The predictor variables were skilled technical knowledge (ST) and general technical knowledge (GT). The dependent variable was attrition. Results of the regression analysis indicated that one variable was predictive: ST score (See Table 6.9). This variable revealed that the lower the student's ST score, the more likely they were to be recycled into another unit to complete training. This model, using the ST variable, correctly predicted attrition in the data 81.1% of the time (See Table 6.10).

	-	2 -		~.	
Cognitive Score	В	S.E.	Wald	Sig.	Exp(B)
ST Score	053	.017	9.523	.002*	.948
Constant	4.743	1.997	5.641	.018*	114.781
$*p \le 0.05$					

Table 6.9. Cognitive Variables in the Equation

Table 6.10. Cognitive Logistic Regression

	Observed		Attrition	Predicted	Percentage Correct	
Step 1	Attrition	Pass	Pass 352	Fail 0	100.0	
	Overall Pe	Fail prcentage	82	0	.0 81.1	

The statistically significant cognitive findings in addition to the noncognitive findings, as discussed above, combined to produce a more effective model for predicting attrition in this setting. The following seven variables were found to statistically predict attrition in this study: low ST scores, unrealistic self-appraisal, preference for short-term goals, low perception of battle buddy support, low perception of unit support, a high level of perceived stress, and a low level of motivation to complete the course.

Research question 3

How do these noncognitive factors work together to result in drop out?

The results of the qualitative data were broken down into two primary themes: course structure and noncognitive factors. These themes incorporated both cognitive and noncognitive variables because both were often intertwined in each student and were difficult, if not impossible to separate when analyzing. First, the course structure theme included aspects related to the pace of instruction, the process of test review, learning and teaching styles and the administration of reteach and study hall. Second, the noncognitive factors theme related to motivation, expectations, self-appraisal, stress management, and family, unit and battle buddy support. The following paragraphs will review how the previously discussed quantitative results and the qualitative results complimented each other to provide for a deeper understanding of the variables that influence attrition.

As covered in the quantitative analyses section of this chapter, the following variables were found to statistically predict attrition in this study: low ST scores, unrealistic self-appraisal, preference for short-term goals, low perception of battle buddy support, low perception of unit support, a high level of perceived stress, and a lack of motivation to complete the course.

Cognitive Ability

Having a lower ST score was statistically predictive for attrition in this study. The quantitative results indicated that the lower the score on the ST subsection of the ASVAB the more likely they were to be recycled to another unit. The mean for ST scores of recycled students was 114.49 (SD=5.587) and passing students was 117.48 (SD=8.178). The mean for GT scores of recycled students was 114.63 (SD=5.081) and passing students was 117.23 (SD= 7.389). The mean ST score for all students in this study was 116.92 (SD=7.838) and their mean GT score was 116.74 (SD=7.080). All of the scores were obtained from the student's personnel files and no qualitative questions were asked of the students regarding the results of these scores.

Self-Appraisal

Having unrealistic appraisal of self was statistically predictive of attrition. Selfappraisal was previously discussed in Chapter IV, however this section provides a brief summary of the student findings. Sedlacek (2004) describes self-appraisal as the student's ability to recognize and accept strengths and weakness, while simultaneously working to improve those deficits, especially with academics. Students in this study who indicated that they expected to have a harder time than most students in AIT were more likely to be recycled. It was also observed that students who entered the environment with a low level of self-appraisal had a more difficult time adjusting to the academic and military environment.

Preference for Long-term Goals

Possessing a preference for long-term goals was statistically predictive of attrition. Students who indicated that they preferred short-term goals over long-term goals were more likely to be recycled in this study. Passing students tended to specify goals that were geared toward the future, typically directly related to education, and could be completed after they finished their job training. Recycled students, on the other hand, tended to identify goals that were immediate, vague and unclear, typically shortterm, and could be completed while they were in their current job training. Hull-Blanks, Kurpius, Befort, Sollenberger, and Nicpon (2005) reported that students who identified academic goals were more likely to remain in school when compared to students with unknown goals. Academic goals, they indicated, tended to be long-term and geared toward a specific outcome. These goals also appeared to motivate students to follow through with their academic decisions as well as provide them with the persistence necessary to face daily challenges. Researchers have historically shown that the ability to set long-term academic goals is a predictor for success in college for traditional students (Fore, 1998; Hull-Blanks, Kurpius, Befort, Sollenberger, Nicpon & Huser, 2005; Sedlacek, 2004; Ting, 1997; Tinto, 1993). Sedlacek argues however, that nontraditional students may have a more difficult time completing this task due their lack of exposure to adequate role models or the lack of consistent reinforcement within their cultural backgrounds.

Battle Buddy Support

Having a low perception of battle buddy support was statistically predictive for attrition in this study. The quantitative results indicated that students who did not have a good battle buddy support system were more likely to recycle to another unit. In this study the qualitative results supported the quantitative results. The quantitative results also revealed that students who indicated talking to their battle buddy about their problems were more likely to recycle. The reason for this relationship is unclear, but qualitative data suggest that it may be that students who responded in that manner talk to their battle buddies but do not discuss specific personal problems because they do not perceive that their buddy can help or are able to provide them with the necessary support to change their situation.

Unit Support

The quantitative results revealed that the perception of an adequate level of unit support was predictive for attrition in this study. Students who responded that their unit was supportive were more likely to be recycled to another unit. The reason for this relationship is unknown. Students who responded may have perceived an adequate level of unit support from their current unit however may have already been transitioning into another unit due to being recycled. Again, the qualitative results provided a deeper understanding of the student's perception of unit support. Although more students who were ultimately recycled indicated that their current unit was supportive of them, the students who were passing indicated less often that they received the same support. This also may be due to the fact that recycled students were provided with one-on-one counseling sessions after they have been selected to recycle whereas passing students seldom received any one-on-one counseling.

The majority of the students, from both groups, commented that too much information was presented in a short amount of time, and along with other military duties, they indicated that there was little time to study, at least until their privileges were increased. Also, while students with grade point averages below 80 were required to participate in study hall, passing students indicated that there was no quiet place for them to study. They indicated that they wanted to increase their knowledge base to compete for honor graduate or just to feel better about themselves.

Again, some passing students emphasized their low perception of unit support because they felt that the attention was geared more toward students who were performing below average than students who were passing and wanting to excel. *Perceived Stress*

Having a perception of a high level of stress is a predictor of attrition in this study. Because of the nature of this course, a certain level of stress has intentionally been built into its structure. The quantitative results under this variable also revealed that the more difficulty a student had when dealing with failure, the more likely they were to be recycled to another unit. Bean and Metzner (1985) emphasized that possessing a high level of stress contributed to attrition. They also indicated that this is typically true when related to all types of stress (family, academic, financial, etc.). Pritchard and Wilson (2003) observed in their study that students with a high level of stress were more likely to have a lower GPA, which in the military environment would result in being recycled to another unit. Also, they noted that the students in their study who intended to stay in the academic environment typically demonstrated positive coping strategies. The variables in that study included: having a high level of stress, personal problems, and trouble concentrating which affected their academic performance; possessing poor coping and stress management skills; difficulty dealing with failure; and experiencing both academic and nonacademic problems. Studies have shown that students who have

had opportunities to develop skills addressing stress prior to a stressful situation do better when placed in another situation which requires those same skills. *Motivation*

The quantitative results revealed that a lack of motivation was predictive for attrition in this study. Motivation and expectation was observed to influence each other when talking to recycled students. For this study, motivation was divided into three categories: motivation to enter the Army, motivation to continue the course, and the impact of motivation on disinterest in the course. Most students expressed that their motivation to continue with the course was based on positive feedback received from family and friends; however, their motivation for not wanting to continue the course was primarily based on unmet expectations. Other students expressed a lack of interest in the course, but noted that it would provide them with the skills to take care of their families. Many of these students began the course knowing that they were not interested and found it difficult to remain motivated enough to complete the course.

Discussion of Findings

The purpose of this explanatory mixed methods study was to understand how noncognitive factors contribute to attrition in the Health Care Specialist Program. This section will discuss the major findings and how they related to attrition. The participants in this study were purposefully selected because they represented the typical student completing job training and only students in the course can identify specific factors and provide the information necessary to increase the awareness of their challenges. These students were from diverse backgrounds with varying levels of support, motivation,

143

confidence, stressors and academic abilities, yet they shared a common bond of becoming a future "soldier medic." Their job while in this environment was to develop the skills necessary to be competent in their field. How they learned to negotiate the environment while in this course set the stage for how they would continue to negotiate situations. Their experiences ultimately will impact how they view the military, whether positively or negatively. Some will have found this environment to be an enjoyable experience where many friends were made whereas, unfortunately, others will only remember the struggles and hardships they endured.

There is a vast amount of literature that discusses the impact of noncognitive factors on student retention in the civilian sector; however, few published studies that have explored noncognitive factors in the military setting. Of these military studies, none have focused specifically on how noncognitive factors impact upon attrition rates for recruits completing AIT during their first six months of military service. Generalizations from other studies which address students participating at the community college or university level may not always apply to this population. The high stress environment in which the students in this study are expected to perform is quite different from the typical classroom where there is an abundance of time to study, a great deal of personal time, and a wealth of resources available to address student's needs on demand.

This study adds to the body of knowledge by specifically focusing on the experiences and challenges presented by students participating in the Health Care Specialist course. The significance of this study is threefold. First, it provides a method

with which to identify noncognitive factors that influence attrition. Second, it increases the understanding of how noncognitive factors work together to result in attrition, an understanding which will provide the necessary information for administrators to establish retention programs for students who are currently in the system. Finally, combining noncognitive and cognitive predictive factors enables administrators to first, identify students who are likely to be at risk for recycle and second, to utilize various approaches to assist at-risk students and help them perform at their peak level, resulting in a better overall quality of graduates.

The theoretical framework taken by the researcher for this study was a combined lens of Tinto's (1975) Model of Student Persistence, Bean's (1980) Model of Student Departure, and Astin's (1984) Theory of Involvement. Tinto (1975) and Pascarella and Terenzini (1980) emphasizes that the greater the student's social and academic integration into an institution, the greater the institutional commitment. Bean (1980) argues that the more satisfied the student is within the institution, the greater the institutional commitment; and Astin (1984) asserts that students with high levels of institutional involvement are more likely to demonstrate increased institutional commitment. These models are consistent in regards to the importance of the student's institutional commitment (fit) and the positive influence of peer groups on retention.

The following quantitative variables were found to statistically predict attrition in this study: low ST scores, unrealistic self-appraisal, preference for short-term goals, low perception of battle buddy support, low perception of unit support, a high level of perceived stress, and a lack of motivation to complete the course. The results to the

145

qualitative data proved to provide a deeper understanding of how noncognitive factors worked together to influence attrition.

Having a low ST score was a predictor for attrition in this study. The lower the ST score the more likely the student was to be recycled to another unit. The ST score, taken from the ASVAB, is a composite aptitude score which consists the following combined subtests: general science, word knowledge, paragraph comprehension, mathematic knowledge and mechanical comprehension. These subtests have been deemed valid predictors and are used to screen and determine the job(s) a soldier will be most successful in. A study conducted by Jimerson, Ferguson, Whipple and Anderson (2002) showed that students with high ASVAB scores were more likely to remain in school, as was true in this study.

Battle buddy support has been recognized by the Army as a support system that is a valuable resource for not only the Army but also for the student. According to TRADOC Regulation 350-6 (2005), the battle buddy system was designed to assist soldiers in developing responsibility, initiative, and dependability. Beginning during basic training, it provides students the opportunity to develop deep relationships with peers who are experiencing the same or similar stressors so that a sense of immediate support is readily available when needed. During this study it was observed that many students may not receive the necessary family support while in this environment. This may be because many of their family members or close friends may not understand how the military system works, and this factor may contribute to attrition. Because of this potential lack of family support, the Army battle buddy system provides a substitute, in

146

terms of an immediate support system for the student, if they are able to make a trusting bond with a peer.

The perception of unit support for students in this study proved to have a significant influence. Students, especially those who were recycled to another unit tended to agree that they did not feel support from their new unit. Rather many indicated that they were immediately treated differently and negatively labeled, which impacted their bonding with the new unit. Of the three observed support systems, at least battle buddy support and unit support can be addressed at the local level. Studies indicate that once students perceive that they are cared about and are valued in the environment their confidence increases and they are more likely to remain in the academic arena.

The results indicated that a high level of stress was a predictor of attrition. This is true especially when the student does not possess the skills necessary to cope effectively with the situation. This academic course is the third largest military occupational specialty training program in the Army. It is a rigorous and demanding course designed to challenge students while in a high stress environment. This course not only incorporates the academic aspects of learning the skills necessary to become a Health Care Specialist, but also is a continuation of the transformation process required to become a soldier. This high stress environment overall challenges the student to be confident in their skills as a student and as a soldier. Students may enter the environment with preexisting personal stressors and must also cope effectively with additional military stressors in order to perform their best in the course. Research shows, as previously discussed, that students who have had prior experiences coping with stressors tend to be more successful addressing issues as they arise, when compared to those students who have not had those opportunities. Effective coping skills are essential for students to remain focused and not become overwhelmed in this high stress environment. The student's inability to appropriately manage these stressors will contribute to attrition from the course. Students come to the environment with varying levels of confidence, and those who demonstrate the ability to effectively handle stress and are able to cope with new personal and academic demands tend to perform better in this course.

The lack of motivation was identified as a predictor of attrition in this study and was consistent with research. Gorman and Thomas (1991) identified motivation as a powerful factor involved in learning. Students enter the environment with preconceived expectations of both the course and the environment and if these expectations were not consistent with their previous notions, motivation tended to diminish. Tinto (1975) explored how expectations impact upon attrition. He argued that students whose expectations were consistent with that of the institution tended to be more committed to the institution and do better in school. He defined this relationship as an "institutional fit." To perform at their potential, students must come into the environment with a high level of motivation and it was manifested in several ways. Although many students may have come to the course motivated, it may have been altered when their expectations were not met. Many of the recycled students indicated that their motivation decreased immediately upon identifying that their prior expectations were not consistent with reality upon arrival. Because of these unmet expectations, some students tended to withdraw and fail the course on purpose in order to be moved to another job specialty or out of the military altogether. Motivation in this study also involved how well the student was able to identify their strengths and weaknesses as well as their overall confidence in their abilities. Possessing motivation and maintaining it throughout the course was essential for success in students participating in this study. Students who were confident in their skills, able to adapt to the environment despite their prior expectations, and who took academic responsibility for their performance tended to remain in the course until graduation.

A combination of the above factors worked together to explain attrition among Health Care Specialists. An attrition model, similar to Tinto's (1975) model, is used to describe a student's attrition from the course. Tinto's model describes the student's departure decision as based on the strength of the relationship between the student's academic (academic abilities and grades) and institutional commitment. He indicates that it is a combination of this commitment which influences their decision to depart the academic environment. The Health Care Specialist model (See Figure 6.1) also describes the importance of the interaction of the student's academic and institutional commitment. In this model, the student enters the environment with cognitive and noncognitive attributes based on their prior experiences. They develop academic goals (academic abilities and grades) and institutional (peer-group and faculty interactions) commitment prior (before entering the military and in basic training) to engaging in the course. Once they interact with the course structure, they reevaluate their academic goals and institutional commitment. If this reevaluation results in adequate levels of commitment, then the student is likely to remain in the environment, whereas if the opposite is true, the student is likely to recycle.



FIG. 6.1. Health Care Specialist Attrition Model demonstrating student's negotiation of the course utilizing both cognitive and noncognitive factors which influences their attrition decision.

Overall, the quantitative and qualitative results were complementary. The results provided a deeper understanding of the noncognitive factors that influence attrition, helped to explain how these factors worked together to influence attrition, and identified predictive factors influencing attrition in students attending this course.

Implications and Recommendations

This study was conducted to provide recommendations for the course administrators to effectively identify factors that influence attrition. Students matriculating through the course provided insightful information that was meant to address some inconsistencies currently existing in the program. Based on the results of this study various implications and recommendations for change were apparent.

First, changing the method in which students are educated about the job requirements of becoming a Health Care Specialist is needed. This education begins during the recruiting process. There exists a need for Army recruiters to provide a more thorough and realistic discussion of job requirements prior to the student committing to the MOS. Because not all recruiters are proficient in the duties and responsibilities of becoming a Health Care Specialist, providing consistent updated information on course expectations should be initiated by the Army Medical Department and communicated to the recruiting personnel through the use of various media, such as with videos or brochures. Seidman (1989) indicated that the process of decreasing attrition and increasing retention begins with the admission process. He emphasized that in order to increase retention, institutional information presented to the student must reflect realistic expectations of the academic setting in order to ensure that the student makes an informed choice and that the academic setting represents a good "fit" for the student. The expectations for the course and environment should again be addressed once the student arrives in the academic environment in order to ensure that expectations are fully understood and maximize the opportunity to establish a fit for the student with the occupation.

The level of motivation that students possessed upon entering the course was also a major predictor for attrition. Because of the high stress military environment, students were expected to participate in both soldier and student activities. The student's prior expectations regarding these activities greatly influenced their level of motivation to complete the course successfully. The majority of students expressed some disappointment with the realities of the course and/or environment. Some had been assigned the MOS with little explanation of the requirements of the job or what to expect of the physical environment. Many of the recycled students commented that due to the environmental conditions or their perception of low unit support they became unmotivated to continue the course. Some students were able to work through their disappointments whereas others allowed the disappointment to overwhelm them and result in an emotional withdrawal from the course. The staff has little control over how the students initially handle reality, however, they must ensure that the environment is conducive for student learning and facilitates students working at their highest potential.

Second, a thorough examination of the structure and effectiveness of the reteach and study hall process is needed. The results of this study suggest that the current reteach and study hall sessions are not sufficiently contributing to the recycled student's

152

learning. These students especially emphasized a mismatch between the teaching style of the instructor and their learning style. Also, providing a supervised quiet area in which students can study will limit the distractions of attempting to study while in the barracks.

Third, consistently ensuring that instructors and Drill Sergeant's receive routine faculty development is vital for the success of the program. Being aware of pedagogical changes and recognizing the importance of different learning styles will help instructors to become more engaging while teaching and facilitate effective student learning.

Fourth, recognizing that students possess different learning styles is important. Getting an assessment of the variety of learning styles may be necessary prior to the beginning of class to ensure that teaching is geared toward addressing student learning differences. Understanding and addressing these differences will ensure that all students have an opportunity to learn. The results from this study also suggested that increasing the ratio of hands-on tasks as compared to didactic instruction would be beneficial to all students.

Fifth, the perception of unit support was a major factor for both groups of students in this study. Unit support can be immediately addressed by unit personnel. Providing an increased level of unit support to incoming recycled students appeared to make a difference in their performance. Results showed that recycled students perceived that they received more support by their current unit than passing students. They indicated that this perceived unit support was provided by the original unit and not the new unit. Recycled students generally perceived their new units to be unsupportive because they were immediately treated differently and/or negatively labeled by faculty

and peers. A study conducted by Hoffman, Richmond, Morrow and Salomone (2002) identified that the greater the student's "sense of belonging" to the institution, the greater their commitment, resulting in institutional integration and retention. This "sense of belonging" equated to the student feeling that the faculty cared about their progress in the course and their overall well-being. Further, Hoffman, and her colleagues found that students who reported a higher level of perceived peer and faculty support experienced lower levels of perceived isolation. Although the environment is highly stressful, all students indicated that they felt more cared about when they received routine feedback from the staff about their performance. Students who recycled were provided with additional personal counseling along with required routine counseling sessions. Passing students, however, indicated that they rarely received any counseling other than what was required. Many of these passing students commented that they felt somewhat slighted because most of the attention went to the "trouble makers" instead of on those who were performing at their peak. Several passing students indicated their intention of competing for honor graduate and commented that they could benefit from routine counseling (formal or informal) and/or frequent encouragement from the staff. The implication for practice is that providing routine unit counseling, whether formal or informal, would positively influence the student's perception that their well-being is important to the unit and that the unit cares about them.

In addition to the routine counseling conducted by the staff, the addition of an ombudsperson is recommended to provide support and counsel for students to learn to succeed in the military system. Preferably the ombudsperson would be an educator who understands pedagogical concepts. Their presence would allow the student to promptly address areas of immediate concern, while remaining in the training area. Although these services are currently provided in the local medical treatment facilities, having someone in the battalion's immediate area would decrease the amount of class time missed by the student while attending an appointment. In addition to the above, it is recommended that this ombudsperson conducts assessments with all entering students prior to the beginning of the course. This would be used as an attempt to create an early warning system to identify at-risk students as well as provide intermittent classes that address common student issues, with the goal of enhancing student success. Overall, the implication of these results is that units should strive to be supportive to all students, regardless of whether the student has recycled into the unit or are original members of the unit. The student's perception of the level of support is what really mattered in this study.

In order for any recommendations to be successful, staff and the students must work together. Grayson and Grayson (2003) suggested that the leadership must be totally invested in all initiated programs if they are to be successful. Sedlacek (2004) reported that some staff detach themselves from attrition or student issues and attribute all student-related issues to the students themselves. Students, however, commented that the staff made an impact on their motivation and perceived unit support, whether positive or negative. In this study, students commented that some instructor's teaching style did not accommodate different learning styles; therefore they struggled with learning and may not have done so in a manner that was easiest for them. Many students commented that they learned better while participating in a hands-on practicum and using multiple visual aids rather than merely reading from a book or reading from the PowerPoint presentation. Others commented that they learn better with group discussion instead of the typical lecture presentation. Ensuring that administrators and instructors receive this type of student feedback is relevant to the unit. It increases their awareness of inconsistencies in student learning and provides them with further options to explore when attempting to improved student learning.

Sedlacek (2004) discussed three groups of faculty and how they can be approached to make changes. First, faculty who are committed to doing something to improve the situation will need incentives to get them geared toward making changes. Second, faculty who are fair-minded and committed, although busy, may need more convincing to take serious action. He indicated that this is usually the largest group. Finally, faculty who oppose any changes to the current structure and adamant about their views are typically resistant to change. Fortunately, although the varying groups of faculty may exist, even within the military structure, full support by the leadership will typically get faculty motivated to make changes.

Conclusions

The purpose of this study was to understand how noncognitive factors contribute to attrition in the Health Care Specialist Program. The noncognitive factors which influence attrition, how these noncognitive factors work together to influence attrition and the predictive cognitive and noncognitive factors were reviewed. The use of explanatory mixed methods research provided the analysis of both quantitative and qualitative data in which the results of one method was used to complement the other. The results of the logistic regression analysis identified six noncognitive variables and one cognitive predictive variable which influenced attrition. The practical implications for the findings of this study may increase course administrator's awareness of the student's perception of how various noncognitive factors impact attrition.

This course required the student to possess a high level of personal confidence in order to be successful. It is a demanding course designed to challenge the student to perform their best under a highly stressful conditions, much like the environment that they may soon encounter. Confidence in knowing the medical material is not immediately expected, however entering the environment with a high level of confidence in personal and academic skills is a valuable asset. Students are expected to be optimistic about their personal capabilities and focus on personal strengths required for successful completion of the course. Students who have a realistic self-appraisal, prefer long-term goals, perceive an adequate level of support (battle buddy, family, and/or unit), possess a high level of motivation, are able to effectively manage stress and have confidence in themselves will more likely be successful in this course. On the other hand, if any component is lacking, the student may be at a higher risk for attrition in this environment.

Future research into the impact of noncognitive variables on attrition within the military environment should be conducted in other military training programs in different locations to determine if results are consistent. Other areas of future research include: studies investigating the retention of medics who enter the military already possessing a medical background; studying the experiences of passing students and their

157

persistence; and studying the experience of the instructor in this course would be beneficial by adding information to what is already known about these populations. Finally, studies to investigate the predictability of other noncognitive variables in the military population would be invaluable as it could potentially add to the creation of a screening tool to be used at the Military Entrance Processing Station to assist with determining admission standards for military service.

REFERENCES

- Army Regulation 635-200. (2005). *Active Duty Enlisted Administrative Separations*. Retrieved May 7, 2007, from http://www.sdmcp.org/Regs/armyenlistedseps.pdf
- Astin, A. W. (1984). Student involvement: A developmental theory for higher education [Electronic version]. *Journal of College Student Personnel*, *25*, 297-308.
- Astin, A. W. (1999). Student involvement: A developmental theory [Electronic version]. Journal of College Development, 40(5), 518-529.
- Bean, J. P. (1980). Dropouts and turnover: The synthesis and test of a causal model of student attrition. [Electronic version]. *Research in Higher Education*, 12(2), 155-187.
- Bean, J. P., & Metzner, B. S. (1985). A conceptual model of nontraditional undergraduate student attrition [Electronic version]. *Review of Educational Research*, 55(4), 485-540.
- Beil, C., Reisen, C. A., Zea, M. C. & Caplan, R. C. (1999). A longitudinal study of the effects of academic and social integration and commitment on retention
 [Electronic version]. *National Association of Student Personnel Administration Journal*, *37*(1), 376-385.
- Berger, J. B. & Milem, J. F. (1999). The role of student involvement and perceptions of integration in a causal model of student persistence [Electronic version].
 Research in Higher Education, 40(6), 641-664.

- Braxton, J. M., Milem, J. F., & Sullivan, A. S. (2000). The influence of active learning on the college student departure process: Toward a revision of Tinto's theory [Electronic version]. *The Journal of Higher Education*, *71*(5), 569-590.
- Carney-Crompton, S., & Tan, J. (2002). Support systems, psychological functioning and academic performance of nontraditional female students [Electronic version]. *Adult Education Quarterly*, 52(2), 140-154.
- Carter, L. R. (2006). Noncognitive predictors of academic success and persistence for
 Hispanic American/First year college students at select community colleges in
 West Texas. Unpublished doctoral dissertation, Texas Tech University, Lubbock.
- Center for Personnel Policy Analysis. (2000). *Recent Projects*. Alexandria, VA: Human Resources Research Organization. Retrieved February 6, 2005, from: http:// www.humrro.org/corpsite/html/programs&personnel/ceppprojects.html
- Creswell, J. (2003). *Research Design: Qualitative, Quantitative, and Mixed Methods Approaches* (2nd ed.). Thousand Oaks, CA: Sage.
- Creswell, J. & Plano Clark, V. L. (2007). *Designing and Conducting Mixed Methods Research*. Thousand Oaks, CA: Sage.
- Cross, K. P. (1991, June). On college teaching. Paper presented at the AnnualConvention of the American Society for Engineering Education, New Orleans,LA.
- Dornbusch, S. M., Ritter, P. L., Leiderman, P. H., Roberts, D. F., & Fraleigh, M. J. (1987). The relation of parenting style to adolescent school performance [Electronic version]. *Child Development*, 58(5), 124-1257.

- Elkins, S. A., Braxton, J. M., & James, G. W. (2000). Tinto's separation stage and its influence on first-semester college student persistence [Electronic version]. *Research in Higher Education*, 41(2), 251-268.
- Fass, M. E., & Tubman, J. G. (2002). The influence of parental and peer attachment on college students' academic achievement [Electronic version]. *Psychology in the Schools*, 39(5), 561-573.
- Felder, R. M. & Brent, R. (2005). Understanding student differences [Electronic version]. *Journal of Engineering Education*, 94(1), 57-72.
- Feldman, K. (1976). The superior college teacher from students view [Abstract][Electronic version]. *Research in Higher Education*, 5(3), 243-288.
- Finn, J. D. (1989) Withdrawal from school [Electronic version]. *Review of Educational Research*, 59, 117-142.
- Finn, J. D., & Rock, (1997). Academic success among students at risk for school failure [Electronic version]. *Journal of Applied Psychology*, 82(2), 221-234.
- Fort Jackson, SC. (2007). Overview: Basic combat training. *U.S. Army*. Retrieved May 8, 2007, from http://www.jackson.army.mil/BCT/what_to_expect.htm#day
- George, D., & Mallery, P. (2006). *SPSS for Windows Step by Step: A Simple Guide and Reference* (6th ed.). Boston: Pearson Education, Inc.
- Gorman, L., & Thomas, G. (1991). Enlistment motivations of army reservist: Money, self-improvement, or patriotism [Electronic version]. Armed Forces & Society, 17(4), 589-599.

- Grayson, J. P., & Grayson, K. (2003). *Research on Retention and Attrition* (Millennium Research Series, pp. 1-56) [Electronic version]. Montreal, Canada: Canada Millennium Scholarship Foundation.
- Guiffrida, D. (2005). To break away or strengthen ties to home: A complex issues for African American college students attending a predominantly white institution [Electronic version]. *Equity & Excellence in Education, 38*, 49-60.
- Hall, T. (2002). Explicit Instruction. Wakefield, MA: National Center on Accessing the General Curriculum. Retrieved May 12, 2007 from http://www.cast.org/publications/ncac/ncac_explicit.html
- Habley, W. R., & McClanahan, R. (2004). What works in student retention: Four year public institutions (pp. 1-25). Iowa City, Iowa: ACT, Inc.
- Hastings, P. R. & Maness, D. L. (2007). Student Evaluation Plan 300-68W10 Health Care Specialist Course. Ft. Sam Houston, TX: U.S. Army Medical Department Center and School.
- Helland, P. A., Stallings, H. J., & Braxton, J. M. (2002). The fulfillment of expectations for college and student departure decisions [Electronic version]. *Journal of College Student Retention: Research, Theory and Practice, 3*(4), 381-396.
- Hoffman, M., Richmond, J., Morrow, J., & Salomone, K. (2002). Investigating "sense of belonging" in first-year college students [Electronic version]. *Journal of College Student Retention*, 4(3), 227-256.

- Holahan, C. J., Valentiner, D. P., & Moos, R. H. (1994). Parental support and psychological adjustment during the transition to young adulthood in a college sample[Electronic version]. *Journal of Family Psychology*, 8(2), 215-223.
- Hull-Blanks, E., Kurpius, S. E. R., Befort, C., Sollenbergen, S., Nicpon, M., & Huser, L.
 (2005). Career goals, and retention-related factors among college freshmen
 [Electronic version]. *Journal of Career Development*, *32*(1), 16-30.
- Jimerson, S. R., Fergusn, D., Whipple, A. D., & Anderson, G. E. (2002). Exploring the association between grade retention and dropout: A longitudinal study examining socio-emotional, behavioral and achievement characteristics of retained students [Electronic version]. *The California School Psychologist*, 7, 51-62.
- Kember, E., Leung, D. Y. P., & Ma, R. S. F. (2007). Characterizing learning environments capable of nurturing generic capabilities in higher education [Electronic version]. *Research in Higher Education*, 48(5), 609-632.
- Krejcie, R. V., & Morgan, D. W. (1970). Determining sample size for research activities [Electronic version]. *Educational Psychology Measurements*, 30(2), 607-610.
- Levitz, R. S., Noel, L., & Richter, B. J. (1999). Strategic moves for retention success. In *New Directions for Higher Education*. No. 108. (pp. 31-49). San Francisco, CA: Jossey-Bass.
- Lowe, H. & Cook, A. (2003). Mind the gap: Are students prepared for higher education [Electronic version]. *Journal of Further and Higher Education*, 27(1), 53-76.

- Madsen, K. (2003). The effects of accuracy in instruction, teacher delivery and student attentiveness on musicians' evaluation of teacher effectiveness [Electronic version]. *Journal of Research in Music Education*, *51*(38-50.
- Maxwell, W. E. (2000). Student peer relations at a community college [Electronic version]. *Community College Journal of Research & Practice*, 24(3), 207-217.
- Merriam, S. B. (1998). *Qualitative Research and Case Study Applications in Education*. San Francisco, CA: Jossey-Bass Publishers.
- Merriam, S. B., & Simpson, E. L. (2000). A Guide to Research for Educators and Trainers of Adults (2nd ed., Updated). Malabar, FL: Krieger Publishing Company.
- Miville, M. L., & Sedlacek, W. E. (1991). Profiles of potential persisters and nonpersister university students [Electronic version]. *The College Student Affairs Journal*, 11(1), 45-53.
- Napoli, A. R., & Wortman. P. M. (1998). Psychosocial factors related to retention and early departure of two-year community college students [Electronic version]. *Research in Higher Education*, 39(4), 419-455.
- Nora, A. (1987). Determinants of retention among Chicano college students: A structural model [Electronic version]. *Research in Higher Education*, *34*(2), 243-263.
- Pascarella, E. T., & Terenzini, P. T. (1980). Predicting freshman persistence and voluntary dropout decisions from a theoretical model [Electronic version]. *The Journal of Higher Education*, 51(1), 60-75.
- Pedhazur, E. J., & Schmelkin, L. P. (1991). Measurement, Design, and Analysis: An Integrated Approach. Hillsdale, NJ: Lawrence Erlbaum Associates.

- Personnel Procurement. (2005). Armed services military personnel accession testing program, Washington, D. C., October 2005 (AR 601-222). Department of the Amy, Navy, Air Force, U. S. Marine Corps, Coast Guard: Author.
- Pritchard, M. E., & Wilson, G. S. (2003). Using emotional and social factors to predict student success [Electronic version]. *Journal of College Student Development*, 44(1), 18-28.
- Ramsden, P. (1987). Improving teaching and learning in higher education: The case for a relational perspective [Electronic version]. *Studies in Higher Education*, 12(3), 275-286.
- Rice, V. J., Woods, Y., and Bundy, M. (2004). [Focus group surveys among academic instructors, Drill Sergeants and commanders regarding 91W-Health Care Specialist Advanced Individual Training]. Unpublished raw data.
- Rocco, T. S., Bliss, L. A., Gallagher, S, & Perez-Prado, A. (2003). Taking the next step:
 Mixed methods research in organizational systems [Electronic version].
 Information Technology, Learning, and Performance Journal, 21 (1), 19-29.
- Rowser, J. F. (1997). Do African American students' perceptions of their needs have implications for retention [Electronic version]. *Journal of Black Studies*, 27(5), 53-76.
- Sedlacek, W. E. (1991). Using noncognitive variables in advising nontraditional students
 [Electronic version]. *National Academic Advising Association Journal*, 11(1), 75-82.

- Sedlacek, W. E. (2004). Beyond the Big Test. Noncognitive Assessment in Higher Education. San Francisco, CA: Jossey-Bass.
- Seidman, A. (1989). Recruitment begins with retention: Retention begins with recruitment. *Colleague*. Bedford, NH: Center for the Study of College Student Retention. Retrieved March 29, 2007, from http://www.cscsr.org/article_recruitment_begins.htm
- Sheldon, E. N. (2003). Faculty support and student retention [Electronic version]. *Journal of Nursing Education*, 42(2), 68-76.
- Stetto, J. E., Gackstetter, G. D., Cruess, D. F., and Hooper, T. I. (2004). Variables associated with attrition from Uniformed Services University of the Health Sciences Medical School [Electronic version]. *Military Medicine*, 169(2), 102-107.
- Tashakkori, A., & Teddlie, C. (Eds.). (2003). Handbook of Mixed Methods in Social and Behavioral Research. Thousands Oaks, CA: Sage Publications.
- Thomas, L. (2002) Student retention in higher education: The role of institutional habitus [Electronic version] *Journal of Education Policy*, *17*(4), 423-442.
- Ting, S. R., & Sedlacek, W. E. (2000). Validity of the Noncognitive Questionnaire-Revised 2 in Predicting the Academic Success of University Freshmen.
 Counseling Center Research Report no. 1-00. College Park: University of Maryland.
- Tinto, V. (1975). Dropouts from higher education: A theoretical synthesis of recent research [Electronic version]. *Review of Educational Research*, *45*(1), 89-125.

- Tinto, V. (1993). Leaving College: Rethinking the Causes and Cures of Student Attrition. (2nd ed.) Chicago: University of Chicago Press.
- Tinto, V. (2000, April). Taking student retention seriously. Rethinking the first year of college. Paper presented at the Annual meeting of the American Association of Collegiate Registrars and Admissions Officers, Minneapolis, MN.
- Tinto, V. (2002, June). Promoting student retention: Lessons learned from the United States. Paper presented at the 11th Annual Conference of the European Access Network, Prato, Italy.
- Tracey, T. J., & Sedlacek, W. E. (1984). Noncognitive variables in predicting academic success by race [Electronic version]. *Measurement and Evaluation in Guidance*, 16, 171-178.
- Training and Doctrine Command Regulation 350-6. (2005). Training: Enlisted Initial Entry Training (IET) Policies and Administration. Monroe, VA: Department of the Army. Retrieved May 7, 2007, from http://www.tradoc.army.mil/tpubs/regs/r350-6.htm
- U.S. Army Medical Department (2007). The Army's 91W health care specialist. *Army Medicine*. Retrieved May 7, 2007, from http://www.armymedicine.army.mil/about/tl/facts91w.htm
- U.S. Army Medical Department Center and School. (2007). Mission statement. *Army Medical Department*. Retrieved May 7, 2007, from http://www.cs.amedd.army.mil/68w/news/mission.htm

- U.S. Code, Title 10. (2005). Uniform Code of Military Justice. Retrieved May 11, 2007, from http://uscode.house.gov/download/pls/10C47.txt
- Woods, P. A., & Sedlacek, W. E. (1998). Construct and Congruent Validity of the Noncognitive Questionnaire (NCQ). Counseling Center Research Report no. 6-88. College Park: University of Maryland. Retrieved November 15, 2006, from http://williamsedlacek.info/publications/articles/construct688.html
- Zajacova, A., Lynch, S. M., & Espenshade, T. J. (2005). Self-efficacy, stress and academic success in college [Electronic version]. *Research in Higher Education*, 46(6), 677-706.

APPENDIX A

Cover Letter/Consent Form

Dear Student,

I am conducting research that will attempt to identify the key factors that effect academic drop-out during the 68W AIT course other than information received by standardized testing (ASVAB). I am asking for your participating in this research because you are currently enrolled in a class that has historically had a moderately high drop-out rate. You are being asked to participate in a 28-item questionnaire today, followed by a 46-item questionnaire and a short interview at a later date. The questionnaires will focus on your academic performance, personal characteristics, and general background, while the interview is designed to provide you with an opportunity to elaborate on your responses to the questionnaire. The total time required for your participation in this study is 1 hour and 30 minutes and you will not have to answer any questions that make you uncomfortable. The final interview will be conducted in an isolated room without input from your chain of command, creating an environment conducive for discussing relevant issues which impact learning. The final interview session will be audio-recorded. Research records and the recorded audiotape will be stored securely and only the researcher and her dissertation chairperson will have access to them.

I will compare your ST, GT and AIT scores to the responses that you make on the questionnaire and during the interview.

There are no personal benefits or monetary compensation from participating in this study except for the satisfaction of knowing that you are contributing information that may be beneficial in helping researchers to determine factors which impact academic performance.

There are no anticipated risks to you as a participant in this research study. You are free to withdraw your consent to participate and may discontinue your participation in the interview at any time without consequence.

If you have any questions about this research protocol, please contact me at 210-378-6390 or my local faculty supervisor, Dr. Rebecca Hooper, at 916-4108. Questions or concerns about your rights as a research participant may be directed to the Director of Clinical Investigation, Brooke Army Medical Center, 3851 Roger Brooke Drive, Ft. Sam Houston, TX 78234; 210 916- 3511.

Please print your name and the last 4 digits of your social security number on the next page and return it to the researcher or civilian ombudsman along with the completed attached questionnaire. Please tear off this page as it provides you with a description of

the study. By signing the next page, you give me permission to report your responses confidentially.

Yvette Woods MAJ, SP US Army

I have read the procedure described above for the research study entitled: The Relationship of Noncognitive Factors and their Contribution to Attrition in a Health Care Specialist Program at Ft. Sam Houston, TX. I voluntarily agree to participate in the questionnaires and interview and I have received a copy of this description.

Printed name of volunteer

Last 4 SSN

Number on questionnaire
APPENDIX B

Instruments

MODIFIED NONCOGNITIVE QUESTIONNAIRE (NCQ)*

232nd Medical Battalion is trying to improve its retention procedures by studying additional information about students. Results will be reported to the researcher only and no individuals will be identified. Please fill in the blank or circle the appropriate answers as they relate to you.

- 1. Your sex is:
 - 1. Male
 - 2. Female

2. Your age is: _____years

3. Your father's occupation:

4. Your mother's occupation: _____

- 5. Your race is:
 - 1. Black (African-American)
 - 2. White (not of Hispanic origin)
 - 3. Asian (Pacific Islander)
 - 4. Hispanic (Latin American)
 - 5. American Indian (Alaskan native)
 - 6. Other: _____
- 6. How much education do you expect to get during your lifetime?
 - 1. Military training only
 - 2. College, but less than a bachelor's degree
 - 3. B.A. or equivalent
 - 4. 1 or 2 years of graduate or professional study (Master's degree)
 - 5. Doctoral degree such as M.D., Ph.D., etc.

7. Please list three goals that you have for yourself right now:

8. Approximately 20% of trainees typically leave before completing 91W AIT. If this should happen to you, what would be the most likely cause?

- 1. Absolutely certain that I will complete 91W AIT
- 2. To change my MOS
- 3. To accept a civilian job
- 4. Marriage or family distractions
- 5. Disinterested in study
- 6. Lack of academic ability
- 7. Insufficient reading or study skills

8. Other: Explain:

9. Please list three things that you are proud of having done:

- 1. _____
- 2. _____
- 3. _____

Please indicate the extent to which you agree or disagree with each of the following items. Respond to the statements below with your feelings at present or with your expectations of how things will be. Write in your answer to the left of each item.

- 1 = Strongly Agree
- 2 = Agree
- 3 = Neutral
- 4 = Disagree
- 5 = Strongly Disagree
- 10. The military should use its influence to improve social conditions.
- _____11. It should not be very hard to get a B (3.0) average in AIT.
- _____12. I get easily discouraged when I try to do something and it doesn't work.
- _____13. I am sometimes looked up to by others.
- _____ 14. If I run into problems concerning school, I have someone who would listen to me and help me.
- _____ 15. There is no use in doing things for people, you only find that it doesn't pay off in the long run.
- _____16. In groups where I am comfortable, I am often looked to as leader.
- _____17. I expect to have a harder time than most students in AIT.
- _____18. Once I start something, I finish it.
- _____19. When I believe strongly in something, I act on it.
- _____ 20. I am as skilled academically as the average student at AIT.
- _____ 21. I expect I will encounter racism during AIT.
- _____ 22. People can pretty easily change me even though I thought my mind was already made up on the subject.
- _____23. My friends and relatives didn't feel I should come into the military.
- _____24. My family has always wanted me to go into the military.

- _____ 25. If course tutoring is made available to me while at AIT, I would attend regularly.
- _____ 26. I want a chance to prove myself academically.
- _____ 27. My high school grades don't really reflect what I can do.

28. Please list offices held and/or groups belonged to in high school or in your community.

1. _____

2. _____

3. _____

*Adapted from the Noncognitive Questionnaire (NCQ) in Tracey, T. J., & Sedlacek, W. E. (1984). Noncognitive variables in predicting academic success by race. *Measurement and Evaluation in Guidance*, 16, 171-178.

Scoring Key: Modified NCQ

Questionnaire Item	Variable Name (Number)
6	Use to score for Self-Concept (I) Option 1 and $2 = 1$; $3 = 2$; $4 = 3$; $5 = 4$; No response = 2
7	A. Options for Long Range Goals (IV) Each goal is coded according to this scheme:
	1 = a vague and/or immediate, short-term goal (for example, "to meet people," "to get a good schedule," "to gain self confidence")
	 2 = a specific goal with a stated future orientation which could be accomplished during undergraduate study (for example, "to join a sorority so I can meet more people," "to get a good schedule so I can get good grades in the fall," "to run for a student government office")
	3 = a specific goal with a stated future orientation which would occur after undergraduate study (for example, "to get a good schedule so I can get the classes I need for graduate school;" "to become president of a Fortune 500 company")
	B. Options for Knowledge Acquired in a Field (VIII) Each goal is coded according to this scheme:
	1 = not at all academically or school related; vague or unclear (for example, "to get married," "to do better," "to become a better person")
	2 = school related, but not necessarily or primarily educationally oriented (for example, "to join a fraternity," "to become student body president")
	3 = directly related to education (for example, "to get a 3.5 GPA," "to get to know my teachers")

	Find the mean for each dimension (for example, long-range goals) and round to the nearest whole number.
Questionnaire Item	Variable Name (Number)
8	Use to score for Self-Concept (I) and Self-Appraisal (II) Option $1 = 4$; 2 through $9 = 2$; No response = 2
9	Use to score for Self Concept (I) Each accomplishment is coded according to this scheme:
	1 = at least 75% of applicants to your school could have accomplished it (for example, "graduated from high school," "held a part-time summer job")
	2 = at least 50% of applicants to your school could have accomplished it (for example, played on an intramural sports team," "was a member of a school club")
	3 = only top 25% of applicants to your school could have accomplished it (for example, "won an academic award," "was captain of football team")
	Find the mean code for this dimension and round to the nearest whole number.

For items 10 through 28, positive (+) items are scored as is. Negative (-) items are reversed, so that 1 = 5, 2 = 4, 3 = 3, 4 = 2, and 5 = 1. A shortcut is to subtract all negative item responses from 6.

Questionnaire Items	Direction	Variable Name (Number)
10	-	Use to score for Racism (III)
11	-	Use to score for Realistic Self-Appraisal (II)
12	+	Use to score for Long-Range Goals (IV)
13	-	Use to score for Leadership (VI)
14	-	Use to score for Availability of Strong Support (V)
15	+	Use to score for Community Service (VII)
16	-	Use to score for Leadership (VI)

Questionnaire Items	Direction	Variable Name (Number)
17	+	Use to score for Racism (III)
18	-	Use to score for Long-Range Goals (IV)
19	-	Use to score for Positive Self-Concept (I)
20	-	Use to score for Realistic Self-Appraisal (II)
21	-	Use to score for Racism (III)
22	+	Use to score for Positive Self Concept (I)
23	+	Use to score for Availability of Strong Support (V)
24	-	Use to score for Availability of Strong Support (V)
25	_	Use to score for Racism (III)
26	_	Use to score for Racism (III)
27	-	Use to score for Positive Self Concept (I)
28		Use to score for Leadership (VI), Community Service (VII) and Knowledge Acquired in a Field (VIII). Each organization is given a code for A, B, and C below. Find the mean for each dimension (for example, Leadership) and round to the nearest whole number.

A. Leadership (VI)

1 = ambiguous group or no clear reference to activity performed (for example, "helped in school")

2 = indicates membership but no formal or implied leadership role; it has to be clear that it's a functioning group and, unless the criteria are met for a score of "3" as described below, all groups should be coded as "2" even if you, as the rater, are not familiar with the group (for example, "Fashionettes," "was part of a group that worked on community service projects through my church")

3 = leadership was required to fulfill role in group (for example, officer or implied initiator, organizer, or founder) or entrance into the group was dependent upon prior leadership (for example, "organized a tutoring group for underprivileged children in my community," "student council")

B. Community Service Relatedness (VII)

1 = no community service performed by group, or vague or unclear in relation to community service (for example, "basketball team").

2 = some community service involved but it is not the primary purpose of the group (for example, "Scouts")

3 = group's main purpose is community service (for example, "Big Brothers/Big Sisters")

C. Knowledge Acquired in a Field (VIII) same coding criteria as used for item 7B

(NCQ) WORKSHEET FOR SCORING

1. POSITIVE SELF-CONCEPT OR CONFIDENCE item 6* + item 8* + item 9* + (6 - item 19) + item 22 + (6 - item 27)

2. REALISTIC SELF-APPRAISAL item 8* + (6 – item 11) + (6 – item 20)

3. UNDERSTANDS and DEALS with RACISM (6 - item 10) + item 17 + (6 - item 21) + (6 - item 25) + (6 - item 26)

4. PREFERS LONG-RANGE GOALS to SHORT-TERM or IMMEDIATE NEEDS item $7A^*$ + item 12 + (6 - item 18)

5. AVAILABILITY of a STRONG SUPPORT PERSON (6 - item 14) + item 23 + (6 - item 24)

6. SUCCESSFUL LEADERSHIP EXPERIENCE (6 - item 13) + (6 - item 16) + item 28A*

7. DEMONSTRATED COMMUNITY SERVICE item 15 + item 28B*

8. KNOWLEDGE ACQUIRED in a FIELD item 7B* + item 28C*

*Recoded item.

Sedlacek, W. E. (2004). *Beyond the big test. Noncognitive assessment in higher education.* San Francisco, CA: Jossey-Bass.

Survey 2 with Scoring

Military Environment Noncognitive Adjustment Scale (MENAS)*/Volunteer Survey

The information requested is essential for improving this course. Please answer these questions honestly and base your responses on the way you have felt since you came to AIT.

Definitions -- For the purposes of this survey: (1) "family" refers to the individuals such as your mom, dad, brother, sister, wife, husband and children. "Family" may also include people not mentioned in the definition but who you consider to be your family. Battle buddy" refers to an individual who is currently with you or were assigned to you at the beginning of your participation in this study (i.e. Battle buddy or designated friend).

Code Number _____

- 1. I attribute my problems with the course to:
 - (5) n/a. I am not having problems
 - (4) Academic Reasons
 - (3) Non-Academic Reasons
 - (2) Both
 - (1) Other
- 2. The highest level of education I have completed is:
 - (1) High School Diploma
 - (2) Some high school education with a GED
 - (3) GED
 - (4) Some College education
 - (5) College Diploma, if yes, what type?

- 3. My current academic grade status in the 68W MOS is:
 - (4) 90-100
 - (3) 80-89
 - (2) 70-79
 - (1) Below 70
- 4. My GPA in high school was:
 - (4) 3.5-4.0 (Mostly A's)
 - (3) 3.0-3.5 (A's and B's)
 - (2) 2.5-3.0 (B's and C's)
 - (1) 2.0-2.5 (C's and D's)
- 5. I have been recycled from this course.
 - (1) Yes (2) No

5a. If yes, how many times?

6. In the year prior to my enrollment, I had _____ (choose number) of job(s).

(5) 0 (4) 1 (3) 2 (2) 3 (1) 4 +

7. I have previous training in health care.

(2) Yes (1) No

If yes, what type of previous training?_____

8. This is my first time away from home.(1) Yes (2) No

9. Some problems I experienced in the course that may have affected my academic performance were:

10. I have what it takes to be successful in the course.

Strongly Strongly Agree Neutral Disagree Disagree Agree (5) (4) (3) (2)(1)11. My grades and my academic performance are my responsibility Often Sometimes Seldom Never Always (4) (3) (2)(1)(0)12. I have trouble staying awake in class. Always Often Sometimes Seldom Never (1)(2)(3) (4) (5) 13. I manage my time well. Always Often Sometimes Seldom Never (4) (3) (1)(0)(2)14. My stress level affects my academic performance. Always Often Sometimes Seldom Never (1)(2)(3) (4) (5) 15. I have people to talk to about my problems and/or stress in my life. Strongly Strongly Neutral Disagree Disagree Agree Agree (5) (4) (3) (2) (1)16. I am coping with and managing my stress well. Strongly Strongly Agree Agree Neutral Disagree Disagree (5) (4) (3) (2)(1)17. I have trouble concentrating which affects my academic performance. Always Sometimes Seldom Never Often (1)(2)(3) (4) (5) 18. I am very motivated to pass this course. Strongly Strongly Agree Agree Neutral Disagree Disagree

(5)

(4)

(3)

(2)

(1)

19. I am doing my best to pass the course.

Strong	Strongly			
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

20. I have *considered* failing the course on purpose.

Strongl	Strongly			
Agree	Agree	Neutral	Disagree	Disagree
(1)	(2)	(3)	(4)	(5)

21. I did fail this course on purpose.

Strongly	T			
Agree	Agree	Neutral	Disagree	Disagree
(1)	(2)	(3)	(4)	(5)

22. I find this course so difficult that I have given up on trying to pass it.

Strongly			Strongly	
Agree	Agree	Neutral	Disagree	Disagree
(1)	(2)	(3)	(4)	(5)

23. The following reason(s) were my motivation to join the army (choose all that apply):

(1)	College Funds	(4)	Parents
(2)	Employment	(5)	Poor home life
(3)	Job Training	(6)	To serve your country
		(7)	Other:

24. The 68W MOS was my first choice for training.

(2) Yes (1) No

Why? _____

25. My personal problems affect my academic performance.

Strongly			Strongly		
Agree	Agree	Neutral	Disagree	Disagree	
(1)	(2)	(3)	(4)	(5)	

26. The 68W MOS is what I expected.

Strongl	Strongly			
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

27. This MOS is very similar to what the recruiter described to me.

Strongly				Strongly
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

28. The 68W MOS is different than what I expected because:

29. I was not aware when I signed up that I might be deployed into combat as a combat medic.

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(1)	(2)	(3)	(4)	(5)

30. The people I consider my family or people closest to me include the following: (check all that apply)

My spouse My children My parents My friends Other _____

31. My family or people who are close to me are supportive of me and my role as a combat medic.

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

32. My family wants me to find a way out of this course and/or get out of the Army.

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(1)	(2)	(3)	(4)	(5)

33. The wishes of my family affect my academic performance.

(1) Yes (2) No

34. I can approach my battle buddy to talk to him/her about personal matters and/or problems in my life.

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

35. My battle buddy helps me and is supportive of me in this course.

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

36. There are other people, besides my battle buddy that I can turn to for help and support here.

(2) Yes (1) No

37. I often talk to my battle buddy about my personal or academic problems.

(2) Yes (1) No

38. I perceive my relationship with my battle buddy to be "close."

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

39. My unit is supportive of me.

Always	Often	Sometimes	Seldom	Never
(5)	(4)	(3)	(2)	(1)

40. There is not enough tutoring or mentoring available during this course.

Strong	ly			Strongly
Agree	Agree	Neutral	Disagree	Disagree
(1)	(2)	(3)	(4)	(5)

41. I have a spouse or dependent family member here with me.

(1) Yes (2) No

If yes, please list relationships (wife, children etc.)

42. The presence of my spouse or dependent family member here with me:

- (2) Helps my academic performance
- (1) Hinders my academic performance
- (0) Does not affect me
- (0) N/A

43. I am having a difficult time dealing with failure.

Always	Often	Sometimes	Seldom	Never
(1)	(2)	(3)	(4)	(5)

44. I have bonded well with my unit.

Strongly				Strongly
Agree	Agree	Neutral	Disagree	Disagree
(5)	(4)	(3)	(2)	(1)

45. I have a strong sense of belonging here.

Always	Often	Sometimes	Seldom	Never
(5)	(4)	(3)	(2)	(1)

46. I have failed out of the 68W MOS course?

(1) Yes (2) No

*Adapted from Rice, Woods and Bundy (2004). Personal Factors Related to Student Performance and Retention Among 91W Health Care Specialist at Ft. Sam Houston, TX

Interview

Each volunteer will be interviewed by the PI after they fill out the questionnaire. This is to give the volunteer the opportunity to elaborate, as well as to enable the researcher to make certain they fully understand the issues the volunteer feels helped or hindered their ability to do well in the 91W AIT program.

Four main areas will be covered during the interview: asking for elaboration on questions from the questionnaire; first, the PI will ask for elaboration on any question that falls on either extreme of the spectrum of the Likert Scale and/or each open-ended question on the questionnaire for clarification.; second, ask the volunteer to tell the PI, in their own words, what the main issues were that hindered and that helped their performance; third, ask the volunteer if there were any particular areas of concern (motivation, expectations of training, etc.); and finally, ask if there was any particular mechanism (study hall, counseling, exam review, etc.) that they thought facilitated their success or failure in AIT. In addition to the above areas the volunteer will also be asked if they have any additional information they would like to offer to the PI that may help them fully understand the issues that may influence academic performance during 91W training.

Examples of questions:

1. Could you please tell me more about your answer on number _____?

2. Could you explain why you strongly agree (or strongly disagree) with _____ on number _____?

3. Could you explain if there were any particular personal areas of concern (motivation, expectations of training, etc.) that you felt either helped you or hindered your performance?

4. What, if any academic assistance (study hall, counseling, exam review, etc.), did you find helpful in this experience. Please describe how it was beneficial to your success or hindered your success?

APPENDIX C

HIPAA Form

BROOKE ARMY MEDICAL CENTER/WILFORD HALL MEDICAL CENTER AUTHORIZATION TO USE AND DISCLOSE PROTECTED HEALTH INFORMATION FOR RESEARCH (APHI Template Version 3, February 04)

You are being asked for permission to use or disclose your protected health information for research purposes in the research study entitled The Relationship of Non-Cognitive Variables and Their Contribution to Attrition Among Health Care Specialists at Fort Sam Houston, Texas.

The Health Insurance Portability & Accountability Act of 1996, Public Law 104-109 (also known as HIPAA), establishes privacy standards to protect your health information. This law requires the researchers to obtain your authorization (by signing this form) before they use or disclose your protected health information for research purposes in the study listed above.

Your protected health information that may be used and disclosed in this study includes:

• Demographic Information for example age, sex, race, etc.

Your protected health information will be used for:

• exploration of and understanding the impact of variables such as determination, motivation, self-development, goal setting ability, support system, leadership experience and community involvement as pertains to successful completion of AIT.

The disclosure of your protected health information is necessary in order to be able to conduct the research project described. Records of your participation in this study may only be disclosed in accordance with state and federal law, including the Privacy Act (5 U.S.C. 552a) and the Health Insurance Portability and Accountability Act of 1996 and its implementing regulations (45 CFR 160 & 164). Note: Protected health information of military service members may be used or disclosed for activities deemed necessary by appropriate military command authorities to ensure the proper execution of the military mission.

By signing this authorization, you give your permission for information gained from your participation in this study to be published in medical literature, discussed for educational purposes, and used generally to further medical science. You will not be personally identified; all information will be presented as anonymous data.

The Principal Investigator may use and share your health information with:

- The BAMC/WHMC Institutional Review Board
- State and Federal Government representatives, when required by law
- BAMC, WHMC or Department of Defense representatives
- Texas A&M University, Department of Educational Administration and Human Resource Development

The researchers and those listed above agree to protect your health information by using and disclosing it only as permitted by you in this Authorization and as directed by state and federal law.

You need to be aware that some parties receiving your protected health information may not have the same obligations to protect your protected health information and may redisclose your protected health information to parties not named here. If your protected health information is re-disclosed, it may no longer be protected by state or federal privacy laws.

You do not have to sign this Authorization. If you decide not to sign the Authorization:

- It will not affect your treatment, payment or enrollment in any health plans or affect your eligibility for benefits.
- You may not be allowed to participate in the research study.

After signing the Authorization, you can change your mind and:

- Notify the researcher that you have withdrawn your permission to disclose or use your protected health information (revoke the Authorization).
- If you revoke the Authorization, you will send a written letter to Yvette Woods, MAJ SP, Occupational Therapy Section, Brooke Army Medical Center, 3851 Roger Brooke Drive, Ft. Sam Houston, TX 78234 to inform him/her of your decision.
- If you revoke this Authorization, researchers may only use and disclose the protected health information <u>already</u> collected for this research study.
- If you revoke this Authorization your protected health information may still be used and disclosed should you have an adverse event (a bad effect).
- If you withdraw the Authorization, you may not be allowed to continue to participate in the study.

If you have not already received a copy of the brochure entitled "Military Health System Notice of Privacy Practices," you may request one. DD Form 2005, Privacy Act Statement - Military Health Records (located on your medical records jacket), contains the Privacy Act Statement for the records. If you have any questions or concerns about your privacy rights, you should contact the Brooke Army Medical Center Privacy Officer at phone number (210) 916-1029 or Wilford Hall Medical Center Privacy Officer at (210) 292-4599.

This Authorization does not have an expiration date.

You are the subject or are authorized to act on behalf of the subject. You have read this information, and you will receive a copy of this form after it is signed.

Volunteer's Signature or Volunteer's SSN Date Volunteer's Printed Name or **Sponsor's SSN**

Signature of Witness

190

Date

APPENDIX D

Retrain/Retest Flowcharts

WRITTEN OR CBT EXAMINATION RETRAINING AND RETESTING





PRACTICAL EXERCISE RETRAINING AND RETESTING



NREWT-B WRITTEN OR OBT EXAMINATION RETRAINING AND RETESTING

APPENDIX E

EXAMINATIONS

Lesson Plan	EXAMINATIONS	CBT or Written Exam	GPE
C191W025	CPR	Х	х
C191W207	Module 1 (EMT-B)	Х	
C191W209	Module 2 (EMT-B)	Х	
C191W211	Module 3 (EMT-B)	Х	
C191W214	Module 4 (EMT-B)	Х	
C191W216	Module 5 (EMT-B)	X	
C191W218	Modules 6 and 7 (EMT-B)	Х	
C191W221	DCMT EMT Final	Х	
C191W220	NREMT-B Practical and Certification Exam	Х	Х
C191W055	Initiate and Manage an IV Infusion		Х
C191W003	Place a Tourniquet		Х
C191W014	Place an Occlusive Dressing		Х
C191W014	Needle Chest Decompression		Х
C191W029	Invasive Procedures/Core Skills	X	
C191W031	Combat Trauma Treatment/Evacuation	х	
C191W001	Assessment and Management of the		
	Trauma Patient		Х
C191W074	Force Health Protection	X	
C191W124	Limited Primary Care/CBRNE	х	
C191W130	Clinical Rotation-Documentation		Х
C191W130	Clinical Rotation- Venipuncture		X
C191W130	Clinical Rotation- Vital Signs		х
C191W130	Clinical Botation- Patient Assessment		X
C191W126	STX- Combat Trauma Lane		Х
C191W127	STX- Battalion Aid Station		X
C191W115	STX- MOUT Site		Х
C191W114	STX- Dismounted Patrol		X
C191W225	FTX- Combat Trauma Lane		x
C191W121	FTX- Battalion Aid Station	·	x
C191W122	FTX- MOUT Site		x
C191W120	FTX- Dismounted Patrol		x

C191W02 C191W025 C191W02 C191W02 C191W02 C191W0 C191W0 191WC FRIDAY 20 October 2006 Location: Company Area & Bldg 1375 Rms: 328, 340, 341, 350, 351, 362, 363 PR Exam Review Day 5 Breakfast Value Trng L CPR PE 7/13 **CPR PE 10/13** CPR PE 11/13 **CPR PE 12/13** R PE 13/13 CPR PE 9/13 **PR PE 8/13** C191W025 C191W025 C191W025 C191W025 C191W02 C191W02 C191W02 C191W03 Cycle 4 THURSDAY 19 October 2006 Personal Hygiene Breakfast Value Trng: Loyalty CPR Lecture 3/4 PR Lecture 4/4 Day 4 PR PE 1/13 PR PE 2/13 CPR PE 4/13 PR PE 5/13 PR PE 3/13 PR PE 6/13 C 191W025 LPC191W08 LP C191W11 LP C191W11 C191W02 LP C191W1 **Vcle** WEDNESDAY 18 October 2006 Lunch Med Terminology 2/3 Med Terminology 3/3 ffective Study Skills Med Terminology 1/3 Personal Hygiene Breakfast Value Trng. Loyalty Initial Counseling itial Counseling CPR Lecture 1/4 CPR Lecture 2/4 Day 3 Dinner Cycle 2 ycle TUESDAY 17 October 2006 2 October 2006 SUNDAY Personal Hygiene Breakfast Value Trng: Loyalty CIF Issue (0800-UTC) ommunity Health mmunity Health ire and Safety Spiritual Fitness Spiritual Fitness Lunch Day 2 ritual Fitness Lunch AWOL Brief SARC brief CIF Issue CIF Issue Breakfast Personnel Records (0930-0955) Finance (0955-1020) ER / NG Cycle 1 MONDAY 16 October 2006 October 2006 AR & NG Liaison (1130-1155) ACAP (1155-1220) SATURDAY DCMT CMD & SEP Brief Lunch IG (1330-1355) SAEDA (1355-1420) Day1 CMD/ 1SG Brief ersonnel Records C & CSM Brief dreakfast EFMP ASAP Brief Blood Drive Standardizatio Dinner Chaplain eakfast ance 0730-0830 0830-0920 0930-1020 1030-1120 1200-1300 1300-1350 1400-1450 1500-1550 1600-1650 1730-1930 1730-1930 0630 - 0700 0700-0750 0800-0830 0515 - 0630 1915-2200 0830 - 0920 0930 - 1020 1815 1905 1030 - 1120 1130 - 1220 1230-1320 1530 1620 1630 - 1720 1330-1420 1430-1520 Week I Hours Week I Hours

X COMPANY, CLASS 00-00 232D MEDICAL BATTALION, CENTER AND SCHOOL

Program of Instruction

Week 2	MONDAY								
Hours	23 October 2	006	TUESDAY		WEDNESDAY		THURSDAY	FRIDA	~
			24 October 2006		25 October 2006	26	October 2006	27 October	2006
	Day 6	Cycle 8	Day 7 C	vcle 9	Davis Cusia 10	4	I		
0515 - 0630			BETECT		DADET # 1	Day 3	Cycle 11	Day 10	Cycle 12
0620 0200			ICTION		DAPFI # I			RETEST	
00/0 - 0:00	Personal Hygiene		Personal Hygiene		Personal Hvaiene	11			
0/00-0/20	Breakfast		Breakfast		Breakfast	Prost fast		Personal Hygiene	
00750-0830	Value Trng: Duty		Value Trng: Duty		Value Trace Duty	Dreaklast		Breakfast	
	CPR Exam		Human Body- Ch 4 2/		Vitals & SAMDIE - ChS / DEV	Value Trng. Duty		Value Trng. Duty	
0830 - 0920						Module I Exam		Airway Management (PE)	Ch7 7/8
	CPB Bottern Cutton	C191W02	25	C191W159	C191	V164	C191W20	_	CIGIMIEI
	CLN FOSIEXAM Critique		Human Body- Ch 4 3	3/3	Vitals & SAMPLE-Ch 5 (PE) 3/4	Mod 1 PostExam Cr	itique	Airway Management (PE) C	h7 8/8
0930 - 1020		C191W02		001111010					0/0 /1
	Intro to Emer Medical Care-Ch1	1/1	Lifting and Moving PLCh6 1/	CIVINION C	C191	V164	C191W20	7	C191W161
0201-0201				2	VITAIS & SAMPLE -Ch 5 (PE) 4/4	Airway Management	(PE) Ch7 1/8	Communication/ Documentation	n 1/3
0711 - 0601		C191W15		021201012					
	Well-Being of the EMT-Ch 2	1/2	Liftine and Moving Pt-Ch6 2/2	2 C151 W 10U	C191	V164	C191W16	1	LP C191W168
1130 - 1220					Aurway Management (LE) Ch7 1/4	Airway Management	(PE) Ch7 2/8	Communication/ Documentation	an 2/3
		C191W15	2	CIGIWIGO					
1230-1320	Lunch		Inch	C121 W 100	C191	V161	C191W16		LP C191W168
	Well-Being of the EMT-Ch 2	6/6	T iffine and Manine Di CLC 7/2		unch	Lunch		Lunch	
1330 - 1420		1			Airway Management (LE) Ch7 2/4	Airway Management	(PE) Ch7 3/8	Communication/ Documentation	n 3/3
		C191W15	7	C191W160	101.0	1917			
	Med/Legal Issues-Ch 3	1/2	Vitals & SAMPLE -Ch5 1	1/2	Airway Management (LE) Ch7 3/4	Airway Management	(PE) Ch 7 1/6	Paris a	LP C191W168
1430 - 1520		1 Interest					(FE) CIL 1 4/0	Patient Assessment (LE)	Chp 8
	Med/Legal Issues-Ch 3	00 01617	Viel P. CAMPIT OL	C191W164	C191	V161	C191W161		LP C191W165
1530 - 1620		1	VITAIS & SAMPLE -Ch5 2/	2	Airway Management (LE) Ch7 4/4	Airway Management ((PE) Ch7 5/8	Module 2 PreExam Review	
		C191W158	8	C191W164	C1917	161	121/11101		
	Human Body- Ch 4	1/3	Vitals & SAMPLE -Ch5 (PE)	1/4	Module 1 Pre Fram Review	Aires Management	(BE) CL7 48		C191W209
1630 - 1720						All way tytanagement ((FE) CD/ 0/8	E/O PO	HS
1800-1850	Dinner	SCI M ISIN		C191W164	C1911	/207	C191W161		
	Defeach		Dincr		binner	Dinner		Dinner	
1900-2200		C101012					Reteach		
		C70 M 1 / 10					C191W207		
Week 2	SATURDAY		SUNAV						
Hours	28 October 200	×							
		Cvela 13	24 UCIODEL 2006						
0730-0830	Breakfast	ar 110	Brackfart	CIE 14					
0830-0920	Museum/ Clothing Sales		Shiritual Ethner		Location: Bldg	1375 Rms: 328, 340	1, 341, 350, 351, 362, 36	33	
0930-1020	Museum/ Clothing Sales		Solition Electron						
1030-1120	Museum/ Clothing Sales		Spiritual Firmee						
1200-1300	Lunch		Lunch						
1300-1350	Museum/ Clothing Sales								
1400-1450	Museum/ Clothing Sales								
1500-1550	Museum/ Clothing Sales								

UNIFORM: BDUs (unless otherwise indicated)

1600-1650

Week 3	MONDAV		TIPERAV						
Hours	30 October 20	90(31 October 2006	I Nove	mber 2006	THURSDA 2 November 2	Y 2006	FRIDAY 3 November 20	006
	Dav 11	Custoffe	2						
	11 (80)	CIERS	Day 12 Cycle 16	Day 13	Cycle 17	Day 14	Cycle 18	Day 15	Cycle 19
0515 - 0630			RETEST					RETEST	
0630 - 0700	Personal Hygiene		Personal Hygicne	Personal Hvoiene		Derecual Husiana			
0700-0750	Breakfast		Breakfast	Breakfast		A USUIDI LINGICIC		reisonal riygiene	
00750-0830	Value Trng: Respect		Value Trng: Respect	Value Truo: Respect		Mahin True Bassies		Breaklast	
	Module 2 Exam		Patient Assessment (LE) Chp 8	8 Patient Assessment Lab #	(3 (PE)	Written Fram Module 3		CV Emanancias Child	6.7
0830 - 0920		CIGIWOW		2210				CA DIRERGINCES CITZ	110
	Module 2 PostExam Critique	11110	Patient Assessment (LE) Cho	W 103	4 (PE) LP C191W16	Mod 3 PastFram Cuitions	LP C191W211		LP CI9IW173
0930 - 1020		C191W/206						CY Emergencies Uniz	0/1
	Patient Assessment (LE)	Chp 8	Patient Assessment (LE) Challer	W 100 Patient Assessment Lab #	4 (PE) LP C191W16:	Resn Emergencies Ch 11	LP C191W211	CV Emeranaiae CL13	LP CI91W173
1030 - 1120		I P CI GI WI K					617	CV Emergencies Ch12	111
	Patient Assessment (LE)	Chp 8	Patient Assessment Lab # 1 (PE)	W103	LP CI91W163	Daen Emanatata Ch 11	LP CI 91WI 72		LP C191W173
1130 - 1220		I P C I 01 W I 64			1-		eic	Ulabelic Emergencies Ch 15	2/1
1230-1320	Lunch		Lineh	C0 M	LP C191W165		LP C191W172		LP C191W212
	Patient Assessment (LF)	Chae	Detions Assessment 1 -1 -4 - (DE)	TUNCH		Lunch		Lunch	
1330 - 1420		cub o	rauent Assessment Lab # 1 (PE)	Patient Assessment Lab #	5 (PE)	CV Emergencies Chp12	1/1	Diabetic Emergencies Ch 15	2/2
	D	LP C191W16:	5 LP C1911	W165	LP C191W165		LP C191W173		LP C191W215
1430 - 1520	Patient Assessment (LE)	Chp 8	Patient Assessment Lab # 2 (PE)	General Pharmacology C	1/1 01 42	CV Emergencies Ch12	2/7	Neurologic Em Ch 13	1/2
		LP C191W165	5 LP C1911	W165	LP C191W171		LP C191W173		1.P C191W203
1530 - 1620	Patient Assessment (LE)	Chp 8	Patient Assessment Lab # 2 (PE)	Resp Emergencies Ch 11	1/3	CV Emergencies Ch12	3/7	Neurologic Em Ch 13	2/2
		LP C191W165	LP C1911	W165	LP C191W172		LP C191W173		1.P C101W202
	Patient Assessment (LE)	Chp 8	Patient Assessment Lab # 3 (PE)	Mod 3 PreExam Review		CV Emergencies Ch12	4/7	Acute Abdomen Chp 14	1/1
1630 - 1720		LP C191W165	LP C191V	V165	LP C191W211		I P C I GI W173		
1800-1850	Dinner		Dinner	Dinner		Dinner		Dinner	LF C171W190
1900-2200	Reteach					Reteach			
		C191W209	0				LP C191W211		
Week 3	SATURDAY		SUNAV	Г					
Hours	4 November 200	9	5 November 2006						
	_	Cycle 20	Cycle 21	T					
0730-0830	Breakfast		Breakfast		Location: Bldg 137	5 Rms: 328, 340, 341, 34	20 351 362 36		
0830-0920	ICTT ICTT		Spiritual Fitness	Γ.	A				
1030-1120	СП		Ispuritual Fitness						
1200-1300	Lunch		Lunch	Т					
1300-1350	On-post pass			Г					
1500-1550	On-post pass								
1600-1650				_					
1/30-1830	Duner		Dinner						
1020-4130									

Week 4	MONDAV		TIPET	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~					
Hours	6 November 200	06	7 Novembu	er 2006	8 November	2006 - 2006	9 November 20	200	FRIDAY
	2		-						
	Day 16	Cycle 22	Day 17	Cycle 23	Day 18	Cycle 24	Day 19	Cycle 25	Day Cvcle 26
0515 - 0630			RETEST				Phase V Run		VETERAN'S DAV
0630 - 0700	Personal Hygiene		Personal Hygiene		Personal Hvoiene		Darconal Husiana		
0700-0750	Breakfast		Breakfast		Breakfast		Rreakfast		Productions
00750-0830	Value Trng: Selfless Service		Value Trng. Selfless Service		Value Trng: Selfless Service		Value Trno: Selfless Service		Dreaklast Value Tree Dammed Courses
	Allergic Reactions/Enven Ch 16		Cardiac Scenarios (Clinical)		Respiratory Scenarios (Clinica	(JE	Written Exam Module 4		value 1118, retsonal courage
0830 - 0920	1/2	LP C101W17		22 171101-2 G I	, ,				
	Allergic Reactions/Enven Chp 1	6	Cardiac Scenarios (Clinical)	PE CI2IMICS	Kinematics of Trauma Ch 22	1/2	Past - Evam Criticus Mod 4	LP C191W214	
0930 - 1020	2/2	1 P C 101 W17		ar intro of a t					
	Subtance Abuse/OD Ch 17	1/2	Cardiac Scenarios (Clinical)	- PL CI91W1/3	Kinematics of Trauma Ch 22	LP C191W197 2/2	Bleeding Chp 22	LP C191W214 1/2	
1030 - 1120		LP C191W176		1 P C101W173					
	Subtance Abuse/ OD Chp 17	2/2	Cardiac Scenarios (Clinical)		Mod 4 PreExam Review	TEL C131 M137	Bleeding Chp 22	2/2 2/2	
1130 - 1220		LP C191W176		1 P C101W173					
1230-1320	Lunch		Lunch	FI CIZIMITO	Lunch	LF C191W214		LP C191W179	
	Env Emergencies Ch 18	1/2	Cardiac Scenarios (Clinical)		Warrior Etho	c Training	Shock Chp 23	1/2	L'unch
1330 - 1420		LP C191W177		LP C191W173		9			
	Env Emergencies Ch 18	2/2	Cardiac Scenarios (Clinical)		Warrior Etho	Training	Shock Chp 23	LP C191 W202 2/2	
1430 - 1520		LP C191W177		1 P C 101W173				and the second se	
	Beh Em Ch 19	1/2	Cardiac Scenarios (Clinical)	C11111211217	Warriow Fthos	Turinian	Soft Tissue injuries Cho 24	LP C191W202	
1530 - 1620		LP C191W178		1 P C 191W173	WALLINE DUILD				
	Beh Em Ch 19	2/2	Respiratory Scenarios (Clinic	al)	AWOL Apprehe	ension Class	Soft Tissue Injuries Ch 24	2/2	
1630 - 1720		LP C191W178		LP C191W172	:			T P C 101W180	_
1800-1850	Dinner		Dinner		Dinner		Dinner	PL C121 M 100	Dinner
1900-2200							Reteach		
								LP C191W214	
Week 4	SATURDAY		VINDA	A					
Hours	11 November 2004	9	12 Novembe	r 2006					
		Cycle 27		Cycle 28					
0730-0830	Breakfast		Breakfast						
0830-0920			Spiritual Fitness		Location: C	ompany Area &	Bldg 1375 Rms: 328, 34	0, 341, 350, 351	, 362, 363
1030-1120			Spiritual Fitness						
1200-1300	Lunch		Lunch						
1300-1350									
1400-1450				-					
1600-1650									
1730-1830	Dinner		Dinner						
1830-2130									

C XACEK S		IONDAY	TUE	SDAY	WED	NESDAY	THURSD	DAY	CDIN	
Hours	13 NG	ovember 2006	I4 Nove	mber 2006	15 Nove	ember 2006	16 Novembe	er 2006	17 Novem	Ser 2006
	Day 20	Cycle 29	Dav 21	Cvcle 30	Dav125	Cuela 11	Dav. 43	0.1.14		
0515 - 0630	RETEST				143 44	16435	1st Combativos	Cycle 32	Day[24	Cycle 33
0630 - 0700	Dereonal Huniana						131 CUIIDAUVC			
0700-0750	Breakfact		Personal hygiene		Personal Hygiene		Personal Hygiene		Personal Hygiene	
00750-0830	Volue Trans Descend C		breaklast		Breakfast		Breakfast		Breakfast	
0000-00100	value i rng: Personal C	ourage	Value Trng: Honor		Value Trng: Honor		Value Trng: Honor		Value True Honor	
0000 - 0030	TRAININ	IG HOLIDAY	Eye Injuries Ch 25	1/1	Bleeding Control & Shoc	k Mgmt (PE)	Face & Throat Injuries	(FE)	Written Exam Mod 5	
0760 - 0.000				LP C191W19	3	LP C191W17		LP C191W182		AICWI01741
0030 - 1020			Chest Injuries	1/1	Bleeding Control & Shoc	k Mgmt (PE)	Face & Throat Injuries	(PE)	Mod 5 Post-Exam Critique	
1701 - 0000				LP C191W196	9	LP C191W17		1 P C 101W183		
			Abdomen & Genital Injuri	ics 1/1	Bleeding Control & Shoel	k Mgmt (PE)	Face & Throat Injuries	(PE)	OB / GYN Emergencies	LF U 71W 210
1030 - 1120				LP C191W195		I P C I 91 W 124		Common d I	0	
			Musculoskeletal Care Chp	9 29 1/2	Bleeding Control & Shocl	k Mgmt (PE)	Face & Throat Injuries	(PE)	OB / GYN Emergencies	LP C191W184
1130 - 1220				LPC191W181		2 IMIOLO GI				
1230-1320	Lunch		Lunch			LF CI9IWIA		LP C191 W182		LP C191W184
			Musculoskeletal Care Chn	016 064	Diading Control & Charl	Marri /DEV	Lunch		Lunch	
1330 - 1420				4		k ingmi (FC)	Musculoskeletal Care	(PE)	OB / GYN Emergencies	
				LP C191W181		LP C191W179		LP C191W181		LP C191W184
1430 - 1520			Head & Spine Injuries Chr	p30 1/3	Bleeding Control & Shock	k Mgmt (PE)	Musculoskeletal Care	(PE)	OB / GYN Emergencies	
				LP C191W204		LP C191W175		1.P.C191W181		1 B / 101010 4
			Head & Spine Injuries Chp	330 2/3	Bleeding Control & Shock	c Mgmt (PE)	Musculoskeletal Care	(PE)	Geriatric Assessment	1/2 LF C121W 104
1530 - 1620				LP C191W204		T D C 101 01 70				
			Head & Spine Injuries Ch3	10 3/3	Face & Throat Injuries Ch	p 26 1/1	Pre - Exam Review Mod 5	LECI91W181	Gariatric Accacement	2/2 LP C191W167
1630 - 1720									Ucidatic Assessment	717
1800-1850	Dinner		Dinner	PL C131W 204	Dinner	LP C191W182		LP C191W216		LP C191W167
000 0001							Dinner		Dinner	-
0077-0061										LP C191W216
Week 5	LVS	TURDAY	STINI	VAU	_					
Hours	18 Nov	'ember 2006	10 Novem	ther 2006						
		Cycle 34		Cvela 35	_					
0730-0930	Breakfast		Breakfast							
0930-1020			Spiritual Fitness			Location: Bldø 137	5 Rmc 328 340 341	35 131 151 151		
1030-1120			Spiritual Fitness							
0011-001	Tunch		Spiritual Fitness							
1400 1460	Lunco		Lunch							
1500-1550										
1600-1650										
				-	_					

UNIFORM: BDUs (unless otherwise indicated)

1930-2130

	MONDAY		TUESDAY	WEDNESDAY	THIBSDAY	ſ		
	20 November 2006		21 November 2006	22 November 2006	23 November 2006		24 November 2006	
	Day 25 Cy	vcle 36	Dav 26 Cvele 37	Daulor Contract				
2	TEST		DADET # 3	Cyclel35	Day Cycle 3	6	Day	vcle 40
	10110		DALF1 # 2	2nd Combatives	THANKSGIVING D	AY	TRAINING HOL	JDAY
	onal Hygiene		Personal Hygiene	Personal Hygiene	Personal Hydiene		Dereonal Husiana	
	IKTASC		Breakfast	Breakfast	Breakfast		Breakfast	
Valt	ue I rng: Integrity		Value Trng: Integrity	Value Trng: Integrity	Value True: Integrity		V-line T	
cdi	atric Assessment 1/2		OB / GYN Scenerios (PE)	All Skills Prep			value 1118 Integrity	
		LP C191W185	LP C191W184		010			
edi	atric Assessment Ch 31 2	2/2	OB / GYN Scenerios (PE)	All Skills Pren	×17			
		LP C191W185						
eds	Airway/Med Em Ch 32	1/2	Pediatric Airway / Medical Emergencies (PF)	All Skills Dran	219			
		1.P.C191W186		data suite irei				
ed	s Airwav/Med Em Ch 32 3	00	Bodiario Almanda I. V. J. Tr. Ant.	LP C1911	219			
		4	r culatific Alitway / Medical Emergencies (PE)	All Skills Prep				
		LP C191W186	LP C191W186	LP C191V	219			
Ĕ,	=		Lunch		Lunch		linch	
eds	l fauma	1/1	Pediatric Airway / Medical Emergencies (PE)	All Skills Prep				
		LP C191W166	LP C191W186	LP C191W	219			
ain	ing Access		Pediatric Trauma (PE) 1/1	All Skills Prep				ĺ
		LP C191W188	LP C191W166					
ЪЪ Пр	ulance Operations		Response to Terriorism/WMD	Bra. Even Barlow Madula & 8.7	617			
		LP C191W187		LTC- LANIN NEVIEW MIGGINE O & /				
Sec	ial Operations		Response to Terriorism/WMD	Pre- Even Deview Module & 2	817	T		
		T P C I D I M I SO						
U		PE C12 M 102	LP C191 W 169	LP C191M	218			
틾	er		Dinner	Dinner	Dinner		Dinner	
	A TUBLAV					1		
	SALUKDAY		SUNDAY					
ļ	25 November 2006		26 November 2006					
ļ	Cyc	le 41	Cycle 42					
ġ.	dast		Breakfast					
f	at Medic Run		Spiritual Fitness	Location: Bldg	375 Rms: 328, 340, 341, 350, 351	. 362. 363		
			Spiritual Fitness		the local land land land a local second land			

Spiritual Fitness Lunch 1130-1150 1200-1450 1400-1450 1500-1550 1500-1650 1700-1720 1730-1930 1930-2130 200

Week 8	MON									
Hours	admond b	2006		DAY	WEDN	ESDAY	THURSDAY		FRIDAY	
		0007 13	mood c	Der 2000	6 Decem	iber 2006	1 7 December 2006		8 December 2	1 900
	Day 33	Cycle 50	Day 34	Cycle 51	Day 35	Cycle 52	Day 36 Cycle 53		Dav 37	Cvcle 54
0515 - 0630			NREMT Run							
0630 - 0700	Personal Hygiene		Personal Hygiene		Personal Hverene		Personal Hypeiene	Da	areonal Humana	
0700-0750	Breakfast		Breakfast		Breakfast		Breakfast		cisultar riygicric	
00750-0830	Value Trng: Loyalty		Value Trng: Loyalty		Value Trng. Lovalty		Value True: Lovalty		alue True: Lovalty	
	Module Review (LE)		NREMT Written Exam 1 5	0 questions	Sickcall and Med Docume	intation	Sickcall and Med Documentation (PE)	SF	KIN DISEASES	
0830 - 0920		LP C191W220		National Registry		I DC101W063		1011105		
	Module Review (LE)		NREMT Written Exam 1 50	0 questions	Sickcall and Med Docume	intation of CLUIT WOOD	Pharmacology/Medication Admin	CCUW171	KIN DISFASES	LFC191W043
0930 - 1020		LP C191W226		National Passister				5		
	Module Review (LE)		NREMT Written Exam 1 50	0 questions	Sickcall and Med Docume	ntation LFC171W03	Pharmacology/Medication Admin	INIW194	KIN DISEASES	LPC191W043
1030 - 1120		LP C191W226		National Revistry		I PC191W053				
	Module Review (LE)		MicroSim Prep		Sickcall and Med Docume	ntation	Pharmacology/Medication Admin	EE EE	ENT DISORDERS	LFC191W043
1130 - 1220		LP C191W220				Lacontrol Del				
1230-1320	Lunch		Lunch		Lunch	PECTAT MOS	L unch	191 W 194	how	LPC191W206
	Module Review (LE)		MicroSim Prep		Sickcall and Med Docume.	ntation (PE)	Immunizations/Chemoprophylaxi	is EE	ENT DISORDERS	
1330 - 1420		LP C191W220								
	Module Review (LE)		MicroSim Prep		Sickcall and Med Documer	ntation (PE)	LPC Immunizations/Chemonronbylevis	191W048	ENT DISODDEDS (BEV	LPC191W206
1430 - 1520		1.P.C.191W220						2	ENT DISONDERS (LE)	
	Module Review (LE)		MicroSim Prep		Sickcall and Med Documer	ntation (PE)	Basic Wound Care	191W048	ENT DISORDERS (PE)	LPC191W206
1530 - 1620		LP C191W220				I DC10111053	Č.			
	10% Commanders Survey		MicroSim Pren		Sicked and Med Decime	LFU.171WUSS	D.L.N.	220 M 161		LPC191W206
1630 - 1720	6				Sickcall and Med Docume.	ntation (PE)	Basic Wound Care	3	ENT DISORDERS (PE)	
1800-1850	Dinner		Dinner		Dinner	LFUL71W030	Dinner	191 W 022	Ten	LPC191W206
1900-2200										
									NREMT RESULTS/ S'	TUDY HALL
Week 8	SATURD	AV	SUND	AV.						
Hours	9 December	r 2006	10 Decemt	ver 2006						
		Cycle 55		Cycle 56						
0730-0930	Breakfast		Breakfast							
0930-1020			Spiritual Fitness		-	Location: Bldg 1375 Rn	18: 328. 340. 341. 350. 351. 362. 363			
1030-1120			Spiritual Fitness							
1200-1400	Lunch		opuruar ruces							
1400-1450	NREMT STUDY HALL		NREMT STUDY HALL							
1500-1550		-								
1600-1650		_	_							
1730-1930	Dinner		Dinner							
1930-2130										

Week 9	MONDAY		THESDAY	Salvua/M			
Hours	11 December 2006	_	12 December 2006	13 Dacembr	2006	THURSDAY	FRIDAY
	n. 30				0007 12	14 December 2006	15 December 2006
	UBY 30 Cycle 57	Da	v/39 Cycle 58	Day 40	Cycle 59	Dav141 Cvclo160	Dave 45
0515 - 0630						RFTFST/Ath Combating	Uayaz I Cyclefol
0630 - 0700	Personal Hygiene	Darsonal Li	(a) and			NETEST/400 COMPANYES	
0700-0750	Breakfast	Breakfact	diene	Personal Hygiene		Personal Hygicne	Personal Hygiene
00750-0830	Value Trng. Duty	Value Trno	Duty	Dreaklast		Breakfast	Breakfast
	ORTHOPEDICS	DECPIDAT	OPV DICODDEDE	value Ling. Duty		Value Trng: Duty	Value Trng. Duty
0830 - 0920		NEW JOST	UNI DISUKDEKS	Introduction to CBRNE		Decontamination of Chemical Casualties	WRITTEN EXAM Primary Care/ CBRNE
	OPTHODEDICS	91W223	LPC191W200		LP C191W017	1.P.C191W117	
	ONTHOLEDICS	RESPIRAT	ORY DISORDERS	CBRNE EQUIPMENT OVE	ERVIEW	Decontamination of Chemical Casualties (PF)	POST-FYAM BEVIEW BUILDED CONTRACT
0930 - 1020		20072				The second se	FOST-EAAM REVIEW Primary Care/ CBRNE
	ORTHOPEDICS	RESPIRAT	ODV DISODDEDS (DE)	Contract of the second	LP C191W046	LP C191W117	LP C191W124
1030 - 1120				MERVE AGEN IS		Decontamination of Chemical Casualties (PE)	RETEACH
	ORTHOPEDICS (PE)	DECDIDAT	OBV DISOBDERS (DEV		LP C191W047	LP C191W117	LP C191W124
1130 - 1220			ONE DISONDERS (FE)	Overview of Biological Warl	fare	Decontamination of Chemical Casualties (PE)	Class A / Barracks Inspection for Cl
1230-1320	LPCIS	1W223	LPC191W200		LP C191W070	LP C191W117	r- 1 30
	ODTIODENICS (DEV	Lunch		Lunch		Lunch	In Lea
1330 - 1420	UNITOFEDICS (FE)	ABDOMIN	AL DISORDERS	Vesicants and Cyanide Blood	Agents	Decontamination of Chemical Casualties (PE)	Class A / Barracks Inspection for Cl
	CPC19	01W223	LPC191W195		LP C191W069	I D CIGINITI	
1420 1620	UNIHUPEDICS (PE)	ABDOMIN	AL DISORDERS	Treat Exposed Nuclear Casu	alty	Decontamination of Chemical Casualties (PE)	A : dhor Torre
0701 - 0041	1.PC19	1W223	LPC191W195		oromoto d I		Alubag Issue
	ORTHOPEDICS (PE)	ABDOMIN	AL DISORDERS (PE)	Treat Exnosed Nuclear Casi	ale:	LP C191W117	
1530 - 1620					any	Decontamination of Chemical Casualties (PE)	Warrior Ethos Training
	ORTHOPEDICS (PE)	ABDOMIN	LPC191W195		LP C191W030	LP C191W117	
1630 - 1720			(E) (E)	Decontamination of Chemica	Casualties	PRE-EXAM REVIEW- Primary Care/ CBRNE	Warrior Ethos Training
1800-1850	Dinner LPC19	1W223	LPC191W195		LP C191W117	LP C191W124	0
		Dinner		Dinner		liner	Dinner
1900-2200							
Week 9	V Mailtro						NREMT STUDY HALL
Hours			SUNDAY				
SINCE	lo December 2006		17 December 2006				
0100 0010	Cycle 62		Cycle 63				
0560-0570	Breaklast	Breakfast		1 22	ation: DIda 1276 D.	110 111 111 111 111 111 111 111 111 111	
0930-1020		NREMT RET	TEST#1		my c/ci gbid :none:	:: 328, 340, 341, 350, 351, 362, 363	
1130-1150				Loc	ation: Bldg 1374 Rms:	227, 228, 240, 241, 363 Bldg 1375 Rms: 262, 263	
1200-1400	Lunch	Lunch					
1400-1450	NREMT STUDY HALL						
1500-1550							
1 700-1720							
1730-1930	Dinner	Dinner					
1930-2130							

Week 10	MONDAY	THESDAY	WEDNESDAY		
Hours	18 December 2006	19 December 2006	20 December 2006	7HURSDAY	FRIDAY
					22 December 2006
	Dayl43 Cycle 64	Day Cycle 65	Day Cycle 66	Dav Cvcle67	Davi
0515 - 0630	RETEST	-			Uzy
0630 - 0700	Personal Hygiene	Personal Hyoiene	B11		
0700-0750	Breakfast	Breakfast	Production	Personal Hygiene	Personal Hygiene
00750-0830	Value Trng: Respect	Value True: Beenaot	DI VANIASI	Breaklast	Breakfast
	W- Training Team Orientation	TOOLOGY STATE OF	value Irng: Kespect	Value Trng: Respect	Value Trng. Respect
0830 - 0920		EXODUS PREP	EXODUS PREP	EXODUS	EXODUS
	Combat Aidbag	TUONIO PLAN			
0930 - 1020		EAUDUS FREF	EXODUS PREP		
	C191W08				
1030 - 1120	Int. Humanitarian Law and Geneva Conv.	EXODUS PREP	EXODUS PREP		
	Int. Humanitarian Law and Geneva Conv				
1130 - 1220		EXODUS PREP	EXODUS PREP		
1230-1320	Lunch				
	Introduction to Medical Threat	10000	Lunch	Lunch	Lunch
1330 - 1420	LP C191W01	EXODUS PREP	EXODUS PREP		
	Introduction to Medical Threat				
1430 - 1520	LP C191W01	EXODUS PREP	EXODUS PREP		
	Waste Disposal in the Field	EXONIS PREP	FVONIC DDED		
1530 - 1620	LP C191W016		EAUDUS FREF		
	Heat Weather Injuries	EXODUS PREP	FXONIS PDFD		
1630 - 1720	LP C191W056		EXOLUSI NEL		
1800-1850	Dinner	Dinner	Dinner	Direct	2
1900-2200					Dunaer
Week 10	SATURDAY	SUNDAY			
Hours	23 December 2006	24 December 2006			
	Cycle 69	Cycle 70			
0730-0930	Breakfast	Breakfast	I continue Bide 1374 Barrer 137 326		
0930-1020		Spiritual Fitness	COMMON BIR 1314 MIR: 771, 770	240, 241, 303 Bldg 13/3 Kms: 262, 263 & Company	Area
1130-1150		Spiritual Fitness	Location: COMPANY AREA		
1200-1400	Linch	opiritual ritness			
1400-1450		1,40,61			
1500-1550					
1600-1650					
1730-1930	Diner				
1030.2130		Duner			
ACT ALON					

Week 11		MONDAV						
Hours	251	December 2006	26 December 2006		WEDNESDAY 27 December 2006	THURSDAY	FRIDAY	
							29 December 2006	
	Day	Cycle 71	Day Cy	cle 72	Day Cvcle 73	Dav Cvola 74		
0515 - 0630							Day	6/3
0630 - 0700	Personal Hygiene		Derconal Huniano					
0700-0750	Breakfast		Breakfast		ersonal Hygiene	Personal Hygiene	Personal Hygiene	
00750-0830	Value Trng: Selfless :	Service	Value True Selfless Service		John Trans Californi Carrier	Breakfast	Breakfast	
	PVONTO				and THIR OCHICSS SELVICE	Value Trng: Selficss Service	Value Trng: Selfless Service	
0830 - 0920	EAUDUS		EXODUS		EXODUS	EXODUS	EXODUS	
0930 - 1020						_		
1030 - 1120								
1130 - 1220								
1720 1270								_
0761-0671	Lunch		Lunch	1	unch	Lunch	Lunch	
1330 - 1420								
1430 - 1520								
1530 - 1620								
1630 - 1720								
1800-1850	Dinner		Dinner	ä	inner	Dinner	Dime	
1900-2200							Dinner	
Wcek 11	SA	ATURDAY	AFUNIS					
Hours	30 De	Vecember 2006						
		Cycle 76	Cvc	e 77				
0730-0930	Breakfast		Breakfast					
0930-1020			Spiritual Fitness	Γ	Location: CUMPANY A	KEA		
1130-1150			Spiritual Fitness					
1200-1400	Lunch		Lunch	T				
1400-1450								
1500-1550			_					
1700-1720								
1730-1930	Dinner		Dinner	T				
1930-2130								

	MONDAY January 2007	TUESDAY	WEDNESDAY	THURSDAY	FRIDAY
		2 January 2007	3 January 2007	4 January 2007	5 January 2007
	ay Cycle 78	Day Cycle 79	Day Cycle 80	Day 44 Cycle 81	Dav 45 Cont. 65
					Daylas Cycle 82
sonal H	fygiene	Personal Hygiene	Descend the second		
akfast		Breakfast	Breakfast	Personal Hygiene	Personal Hygiene
ie Trns	s Honor	Value Trng: Honor	Value Trace House	Dreaktast	Breakfast
XO	DUS	EXODUS RECOVER	RY EXODUS RECOVERY	Value Trng: Honor Cold Weather Injuries	Value Trug: Honor Written Exam - Force Health Protection
		EXODUS RECOVER	V EXODIS RECOVEDV	BITES & STINGS LP C191W057	Eram Review - Force Health Protocica
		EXODUS RECOVER	Y EXODUS RECOVERY	DEPRESSION & SUICIDE PREVENTION	LP C191W074 Infection, Asepsis & Sterile Technique
		EXODUS RECOVER	V EXODIS BECOVEBY	LP C191W071 Stress Management	Starile Pronavation
UCU		Lunch	Lunch	Linch Linch	LP C191W052
		EXODUS RECOVER	Y EXODUS RECOVERY	Stress Management	Luncn Assemble a Needle & Syringe & Draw Medication (LE)
		EXODUS RECOVER	Y EXODUS RECOVERY	Force Health Exam Review LP C191W049	LP C191W009 Administer IM, SQ, ID Injections
		EXODUS RECOVER	Y EXODUS RECOVERY	Force Health Exam Study Hall	LP C191E008 Assemble a Needle & Injection (PE)
		EXODUS RECOVER	Y EXODUS RECOVERY	Force Health Exam Study Hall	LP C191W008/009 Assemble a Needle & Injection (PE)
Ŀ		Dinner	Dinner		LP C191W008/009
					Dinner RETEACH
					LP C191W074
	6 January 2007	SUNDAY 7 Journal 7	ſ		
	Cycle 83	Cvele 84	T		
fast		Breakfast	Ţ		
		Spiritual Fitness	Location: Company Area &	Rido 1374 Rms: 327 228 340 241 350	151 3/7 0 CT / TC
		Spiritual Fitness		100 11+0 '0+0' 0+0' 0+0' 0+0' 0+0' 0+0'	201, 202 & 51 0 1 rng Area
_		Juneh	1		
			Т		
Week 13 Hours	MONDAY	TUESDAY	WEDNESDAY	THURSDAY	FRIDAV
------------------	---	---	--	---	--------------------------------
	s January 2007	9 January 2007	10 January 2007	11 January 2007	12 January 2007
	Day 46 Cycle 85	Day 47 Cvcle 86	Dav 48 Curla 87	D 40	
0515 - 0630	RETEST	DAPFT # 3	1st Road March	5th Combativas	Dayl50 Cyclel89 P.F.T.F.C.T
0630 - 0700	Personal Hygiene	Derconal Humana	e	JUL COMPANYES	VELES I
0700-0750	Breakfast	Resultant	Personal Hygiene	Personal Hygiene	Personal Hygiene
00750-0830	Value Trng: Integrity	Victor Transition	Breaklast	Breakfast	Breakfast
	Basic Math & Calculate an IV Flour Date		Value Trng: Integrity	Value Trng. Integrity	Value Trng. Integrity
0830 - 0920	Paste viant a Calculate all 1 V FIOW Kate	I actical Combat Casualty Care	Invasive Core Skills Round Robin	Written Exam - Invasive Procedure/ Core Skills	Cougar War Day
	LP C191M	050 LP C191W14	C191W003/004/029/052/055/065	LP C191W029	
	unuate of Manage Intravenous Infusion	Tactical Combat Casualty Care	Invasive Core Skills Round Robin	Post Exam Review - Invasive Procedure/Core	
0930 - 1020	LP C191W	055 T D CTOTIVIA		Skills	
	Initiate & Manage Intravenous Infusion	Tactical Combat Casualty Care	Invasive Core Skills Round Rohin	Introduction to Bullistics	
1030 - 1120	TP C191W	550 S20			
	Initiate & Manage Intravenous Infusion	Control Bleeding and Hypovolemic Shock	197 Core Skills Round Robin Invasive Core Skills Round Robin	Access & Initial Manual City Transformer	
1130 - 1220	(19 C191W	500 E E E E E E E E E E E E E E E E E E		reserves on titititat intanage of the Leauma Pt	
1230-1320	Lunch	Linch Lr C191 W002	C191W003/004/029/052/055/065	LP C191W001	
	Airway Management	Control Bleeding and Hynovolemic Shock	Investive Core Shills Damed Dati-	Lunch	Lunch
1330 - 1420	I P C IOIW		INVASIVE CORE SMILLS KOUND KOON	Assess & Initial Manage of the Trauma Pt	
	Airway Management	Treat for Shock	C191W003/004/029/052/055/065 Invasive Core Skills Round Bokin	LP C191W01	
1430 - 1520	W1612 41			Datuencial Medications	
	Airway Management (PE)	Treat for Shock	C191W003/004/029/052/055/065	LP C191W082	
1530 - 1620		Treation Strock	Invasive Core Skills Round Robin	Head Injuries	
	Airway Management (PE)	Obtain and Level Blood & LP C191 W004	C191W003/004/029/052/055/065	LP C191W007	
1630 - 1720		Optain and Label Blood Specimen	Review - Invasive Procedure/Core Skills	Ocular Injuries	
1800-1850	Dinner	002 LP C191W065	LP C191W029	LP C191W006	
		Dinner	Dinner	Dinner	Dianer
1900-2200				RETEACH	
Week 13	SATIBBAV				
Hours		SUNDAY			
	15 January 2007	14 January 2007			
0730-0930	Breakfast	Breakfast			
0930-1020		Spiritual Fitness	I contion: Commany A see P. 1	014-1374 D	
1030-1120		Spiritual Fitness		oldg 13/4 KMS: 32/, 328, 340, 341, 350	1, 331, 362 & ST 6 Trng Area
1130-1150		Spiritual Fitness	Location: STX Site		
1200-1400	Lunch	Lunch			
1400-1450					
0001-0001					
1700-1720					

UNIFORM: BDUs (unless otherwise indicated)

1730-1930

Week 14	MC	DNDAY		55DAV				
Hours	15 Jar	nuary 2007	le Jan	uary 2007	I WEDN	ESDAY ary 2007	18 [mmmm 2007	FRIDAY
								19 January 2007
	Day	Cycle 92	Day 51	Cvcle 93	Day 52	Cycle 94	Dav 53 Cvcle 95	Dav 54 Cvelo 96
0515 - 0630	ML	K Day					6th Combatives	RETEST/ Phase V+ Run
0630 - 0700	Personal Hygiene		Personal Hvgiene		Derconal Huniana			
0700-0750	Breakfast		Breakfast		Brankfact		rersonal hygiene	Personal Hygiene
00750-0830	Value Trng. Personal Co.	Jurage	Value Trng: Personal Con	urace	Value Trug. Personal Cour-	000	Breakfast Voloc Too Book I C	Breakfast
			Thoracic Trauma		Introduction to the Medica	Emonstion Contour	With the rersonal Courage	Value I rng: Personal Courage
0630 - 0920					MILLIOUUCIION TO THE MEDICA	I EVACUATION SYSTEM	Written Exam - COMBAT / EVAC	COMBAT TRAUMA ROUND ROBIN TRAINING
0700-0000				LP C191W014	+	LP C191W023	LP C191W	(PE) 1331 C191 W01 9/01 8/001 /01 4/021 /027/01 0/003/033
			Thoracic Trauma		Manual Evacuation		Post Exam Review - COMBAT/ EVAC	COMBAT TRAUMA ROUND ROBIN TRAINING
0930 - 1020				LP C191W014		1 P C 191 WO18		(PE)
		-	Abdominal Trauma		Litter Evacuation		Triage Round Robin (PE)	COMBAT TP 413/01/01/01/02/102/101/009/033
1030 - 1120								(PE)
			Burn Initrac	LF UISI WUIS	1	LP C191W015	LP C191W081/055/001/REVII	EW C191W019/018/001/014/021/027/010/003/033
1130 - 1220			so and the more		Casualty Triage		Triage Round Robin (PE)	COMBAT TRAUMA ROUND ROBIN TRAINING (PE)
1230-1320	Lunch			LP C191W005		LP C191W027	LP C191W081/055/001/REVIE	W C191 W019/018/001/014/021/027/010/003/033
			Lunch		Lunch		Lunch	Lunch
0071-0131			Burn Injures		U.S. Field Medical Card (F	MC)	Triage Round Robin (PE)	COMBAT TRAUMA ROUND ROBIN TRAINING
0-L 000				LP C191W005		LP C191W035	LP C191W081/055/001/REVIE	(PE) WC191W01940184061701440217027701640027022
1430 - 1670			Spinal Trauma		Evacuation Request Proced	ures	Triage Round Robin (PE)	COMBAT TRAUMA ROUND ROBIN TRAINING
0701 - 0661				LP C191W012		LP C191W021	1 P C101W081/055/001/20 EVITE	(PE)
			Musculoskeletal Trauma		Evacuation Platforms		Triage Round Robin (PE)	COMBAT TRAIMA ROLINI NO PORIN TO ANNUAL
1530 - 1620								(PE)
			N	LP CI91W010		LP C191W033	LP C191W081/055/001/REVIE	W C191W019/018/001/014/021/027/010/003/033
1630 - 1720			Musculoskeletal Trauma		Review - COMBAT/ EVA	c	Triage Round Robin (PE)	COMBAT TRAUMA ROUND ROBIN TRAINING
1800-1840				LP C191W010		LP C191W031	LP C191W081/055/001/REVIE	W C191W019/018/001/014/021/027/010/003/033
0.001-0000	Dinner		Dinner		Dinner		Dinner	Dinner
1900-2200							RETEACH	Phase V+ Ceremony
							LP C191W0	31
Week 14	SATI	BDAV	1.15					
Hours	20 Janu	uary 2007	21 Januar	ary 2007				
		Cycle 97		Cvcle 98				
0730-0930	Breakfast		Breakfast			Location 1	TOMPANY A BEA & WHICKEN TH	
0930-1020			Spiritual Fitness			TOCALIO	II. COMFANT AKEA & WHISNEY IK	IG AKEA
1030-1120			Spiritual Fitness	_				
120-1400			Spiritual Fitness					
1200-1400	Lunch		Lunch					
1400-1450								
1600-1650								
1700-1720								
1730-1930	Dinner		Dinner					

UNIFORM: BDUs (unless otherwise indicated)

Week 15	MONDAY	TIDENAN			
Hours	22 January 2007	3 January 2007	WEDNESDAY	THURSDAY	FRIDAY
		/007 Alaning C7	24 January 2007	25 January 2007	26 January 2007
	Day 55 Cycle 99	Day 56 Cycle 100	Day 57 Cycle 10	Day 58 Cycle 102	Dav159 Cvcle 103
0515 - 0630			RAPFT		
0630 - 0700	Personal Hygiene	Personal Hygiene	Personal Hydiene	Descend Harding	
0700-0750	Breakfast	Breakfast	Breakfast	F CISUIAI INGICIIC Reaabfaar	Personal Hygiene
00750-0830	Value Trng. Loyalty	Value Trng. Lovalty	Value True: Lovalty	Victor Trace Londay	Dreaktast
	COMBAT TRAUMA ROUND ROBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PE)	Trauma Clinic # 2 Surgical Cric/ 2 Pt lanes	PE) COMBAT TRAIMA I ANES TRAINING A	Value Irng: Loyalty
0830 - 0920	(PE) C101 W01 0/01 8/001 /01 4 /001 /007 /002 /002 /002			LP C191W001/002/003	LP C191W001/002/003
	COMB AT TP A1 M A DOT NO PODN TP 1 M 00	LP C191W003/001/055/144	LP C191W144/001/00	/002/010 005/010/014/018/019/021/023/055/144	005/010/014/018/019/021/023/055/144
	COMPAN LAAUMA KUUND KUBIN I KAININU	I rauma Clinic #1 Bleeding & IV (PE)	Trauma Clinic # 2 Surgical Cric/ 2 Pt lanes (PE) COMBAT TRAUMA LANES TRAINING (I	E) COMBAT TRAUMA LANES TRAINING (PE)
0930 - 1020	C191W019/018/001/014/021/027/010/003/033	1 P C191W003/001/055/144		LP C191W001/002/003	LP C191W001/002/003
	COMBAT TRAUMA ROUND ROBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PF)	Traima Clinic # 2 Surgical Cric/ 2 Bt Ionac /	002/010 005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/027/033/055/144
1030 - 1120	(PE)			LP C191W001/002/003	T. CUMBAL I KAUMA LANES TKAINING (PE) [LP C191W001/002/003
	COMP AT TB A114 A DOUDIN DOUDIN 0003/033	LP C191W003/001/055/144	LP C191W144/001/003	/002/010 005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/0227/033/055/144
	COMBALLIKAUMA KUUND KOBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PE)	Trauma Clinic # 2 Surgical Cric/ 2 Pt lanes (PE) COMBAT TRAUMA LANES TRAINING (F	E) COMBAT TRAUMA LANES TRAINING (PE)
1130 - 1220	(FE) C191W019/018/001/014/021/027/010/003/033	1 D C1011000 200111010		LP C191W001/002/003	LP C191W001/002/003
1230-1320	Lunch	Linch	LF C191W144/001/003	/002/010 005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/023/055/144
	COMBAT TRAUMA ROUND ROBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PE)	Traima Clinic # 2 Survival Cric/ 2 Bt Ionar /	LUNCH AT TRAIN AND AT AND COMPANY	Lunch
1330 - 1420	(PE)			LP CI91W001/002/003	E) COMBAT TRAUMA LANES TRAINING (PE) LP C191W001/002/003
	COMB AT TB A11/14 DCI DID DOD03/033	LP C191W003/001/055/144	LP C191W144/001/003	/002/010 005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/027/033/055/144
	COMPAL TRAUMA KUUND KUBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PE)	Trauma Clinic # 2 Surgical Cric/ 2 Pt lanes (PE) COMBAT TRAUMA LANES TRAINING (P	E) COMBAT TRAUMA LANES TRAINING (PE)
1430 - 1520	C191W019/018/001/014/021/027/010/003/033	LP C191W003/001/055/144	T B C101101144/001/002	LP C191W001/002/003	LP C191W001/002/003
	COMBAT TRAUMA ROUND ROBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PE)	Trainma Clinic # 2 Survival Cris(2 B. Januar /	002/010 002/010/014/018/019/021/02/023/044	005/010/014/018/019/021/027/033/055/144
1530 - 1620	(PE)		Tradina Chilly # 2 Jugical City & FU Jarics (LP C191W001/002/003	E) COMBAT TRAUMA LANES TRAINING (PE)
	C191W019/018/001/014/021/027/010/003/033	LP C191W003/001/055/144	LP C191W144/001/003	002/010 005/010/014/018/019/021/023/033/055/144	LF C191 W001/002/003 005/010/014/018/019/021/027/033/055/144
	COMBAT TRAUMA ROUND ROBIN TRAINING	Trauma Clinic #1 Bleeding & IV (PE)	Trauma Clinic # 2 Surgical Cric/ 2 Pt lanes (PE) COMBAT TRAUMA LANES TRAINING (P	E) COMBAT TRAUMA LANES TRAINING (PE)
1630 - 1720	C191W019/018/001/014/021/027/010/003/033	T P C191W003/001/055/144		LP C191W001/002/003	LP C191W001/002/003
1800-1850	Dinner	Dinner	Dinner	002/010 002/010/014/018/019/021/02//033/053/144	005/010/014/018/019/021/027/033/055/144
				Dinner	Dinner
1900-2200					
Week 15	SATURDAV	A FUNIS			
Hours	27 January 2007				
	Cool 104	20 January 2007			
0630-0730	-2014104	Cycle 105			
0730-030	Rreakfast	Burdlers	Location: WH	ISKEY IKNG AKEA	
0930-1020		Dicatuasi Sniritual Fitness	-		
1030-1120		Spiritual Fitnace			
1130-1150		Spiritual Fitness			
1200-1400	Lunch	Lunch			
1400-1450					
1500-1550		_			
0001-0001					
1730-1930	Dinner				
		Dunct			

209

Week 16	MONDAV	THE OWNER OF THE		100L	
Hours	29 January 2007	1 UESDAY	WEDNESDAY	THURSDAY	FRIDAY
		20 January 2007	31 January 2007	1 Febrauary 2007	2 Febrauary 2007
	Day 60 Cycle 106	Day 61 Cycle 107	Day 62 Cycle 108	Davl63 Cvcla109	David 4
0515 - 0630					
0630 - 0700	Personal Hygiene	Personal Hyviene	Durrowal Utrainer		
0700-0750	Breakfast	Breakfast	Preskfast	Personal Hygiene	Personal Hygiene
00750-0830	Value Trng: Duty	Value Trne: Lovalty	Value True: Dute	Dreaklast	Breakfast
	COMBAT TRAUMA LANES TRAINING	COMBAT TRAUMA LANES TRAINING	Scenario Training / Clinical Potation Clinical 1 Dr	Value Iring: Duty	Value Trng. Duty
0830 - 0920	(GPE) LP C191W001/002/003 005/010/014/018/010/0021/022/022/022/022	(GPE) LP C191W001/002/003	CI91W130	C191W130	Scenario Training / Clinical Rotation Clinical LP's C191W130
	COMBAT TRAITMA I ANIES TD AIMING	005/010/014/018/019/021/027/033/055/144	STX LP's C191W114,115,126,127,2	24 STX LP's C191W114,115,126,127,2	24 STX LP's C191W114.115.126.127.224
0001 - 0200	(GPE) LP C191W001/002/003	COMBAL TRAUMA LANES TRAINING	Scenario Training / Clinical Rotation Clinical LPs	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
0701 - 0560	005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/033/055/144	CI91W130	C191W130	C191W130
	COMBAT TRAUMA LANES TRAINING	COMBAT TRAUMA LANES TRAINING	StALFS C191 W114,115,126,127, Scenario Training / Clinical Boration, Clinical 1 Ph	24 STX LP's C191W114,115,126,127,2	24 STX LP's C191W114,115,126,127,224
1030 - 1120	(GPE) LP C191W001/002/003	(GPE) LP C191W001/002/003	CI91W130	Crothering / Clinical Kotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
	005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/027/033/055/144	STX LP's C191W114,115,126,127.2	24 STX LPs C191W114 115 126 127 2	CI9IWI30 STVI ProcisiWill HE INCLUSE
	COMBAT IRAUMA LANES TRAINING	COMBAT TRAUMA LANES TRAINING	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical Di-
1130 - 1220	005/010/014/018/019/021/022/03	(GPE) LP C191W001/002/003 005/010/014/018/010/02/003	CI91W130	C191W130	C191W130
1230-1320	Lunch	1	STX LP's C191W114,115,126,127,2	24 STX LP's C191W114,115,126,127,2	24 STX LP's C191W114,115,126,127,224
	COMBAT TRAUMA LANES TRAINING	COMPATING LANDS TO AND TO AND A	Lunch	Lunch	Lunch
1330 - 1420	(GPE) LP C191W001/002/003	COMBAL TRAUMA LANES TRAINING	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
	005/010/014/018/019/021/023/055/144	005/010/014/018/019/021/027/033/055/144	C191W130 STX LPs C191W114 115 126 127 2	C191W130 STV1 By C101W114 115 105 107 20	C191W130
	COMBAT TRAUMA LANES TRAINING	COMBAT TRAUMA LANES TRAINING	Scenario Training / Chical Potetion Clinical I Di-	24 DIALPSCI9IWI14,[15,126,127,2	24 STX LP's C191W114,115,126,127,224
1430 - 1520	(GPE) LP C191W001/002/003 005/010/014/00 8/016/003 0022/003 0056/14/00	(GPE) LP C191W001/002/003	CI91W130	Scenario Training / Clinical Rotation Clinical LP's C191W130	Scenario Training / Clinical Rotation Clinical LP's C191W130
	COMBAT TRAIMA LANFS TRAINING	COMB AT TB A1144 1 ANT 6 TO 27/033/055/144	STX LP's C191W114,115,126,127,2	24 STX LP's C191W114,115,126,127,22	5TX LP's C191W114.115.126.127.224
1530 - 1620	(GPE) LP C191W001/002/003	COMBAL LKAUMA LANES TKAINING (GPE) LP C191W001/002/003	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
	005/010/014/018/019/021/027/033/055/144	005/010/014/018/019/021/027/033/055/144	STX LP's C191W114-115-126-127-2	CI91W130 STV1P% C191W114 115 126 127 22	C191W130
	COMBAT TRAUMA LANES TRAINING	COMBAT TRAUMA LANES TRAINING	Scenario Training / Clinical Rotation Clinical I De	Connero Training (Clinical D. 441 - 01) 120,127,24	A 51X LPS CI91W114,115,126,127,224
1630 - 1720	(GPE) LP C191W001/002/003 005/010/014/018/019/021/027/0333/055/144	(GPE) LP C191W001/002/003	CI91W130	Occuration training / Climical Kotation Climical LPS C191W130	Scenario Training / Clinical Rotation Clinical LP's C191W130
1800-1850	Dinner	441/CCU/260/120/120/210/010/010/010/000	STX LPs C191W114,115,126,127,2	24 STX LP's C191W114,115,126,127,22	4 STX LP's C191W114,115,126,127,224
			Dinner	Dinner	Dinner
1900-2200					
West 16	A A MARK AN A A A A				
TT	SALUKDAY	SUNDAY			
Hours	3 Febrauary 2007	4 Febrauary 2007			
	Cycle 111	Cycle 112			
0/30-0930	Breakfast	Breakfast	Location: WHISk	TEV & STX TRNG AREA	
100011020		Spiritual Fitness			
1130-1150		Spiritual Fitness			
1200-1400	Lunch	Spurtual Funess			
1400-1450		74070			
1500-1550					
1600-1650					

UNIFORM: BDUs (unless otherwise indicated)

Dinner

1730-1

Week 17	MONDAY				
Hours	5 February 2007	TUESDAY	WEDNESDAY	THURSDAY	FRIDAV
	1007 A 1000 00 0 0	0 rebrauary 2007	7 Febrauary 2007	8 Febrauary 2007	9 Febrauary 2007
	Day 65 Cycle 113	Day 66 Cycle 114	Dav 67 Custoff 116	P 60	
0515 - 0630				Dayl68 Cycle116	Day 69 Cycle 117
0630 - 0700	Personal Hyoiene	B			
0700-0750	Breakfast	Personal Hygiene			
00750-0830	Value Trne: Duty	Dreakiast Volue Terre Dere	Breakfast	Breakfast	Breakfast
	Scenario Trainine / Clinical Potation Clinical 1	Pro Company Lary	Value Trng, Respect	Value Trng: Respect	Value Trno: Resnect
0830 - 0020	C191W130	C Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical I Pre
0700 - 0.000	STX LP's C191W114,115,126,12	7.224 STX LP's C191W114 115 126 127 224	CI91W130	C191W130	C191W130
	Scenario Training / Clinical Rotation Clinical L	P's Scenario Training / Clinical Rotation Clinical I Die	Someric T-mining / CI 91 W 14, 115, 126, 127, 22	4 STX LP's C191W114,115,126,127,224	4 STX LP's C191W114,115,126,127,224
0930 - 1020	C191W130	C191W130	COMMUNICATION CONTRACTION CONTRACTOR LPS	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
	STX LP's C191W114,115,126,12	7.224 STX LP's C191W114,115,126,127,22	34 STX LP's C191W114.115.126 127 224	4 STVI De CIGIWILL DE 132 132	C191W130
	Section 1 raining / Clinical Rotation Clinical L.	P's Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Trainino / Christel Botetica, 15,126,127,224	STX LP's C191W114,115,126,127,224
1030 - 1120	CI9IWI30 STVIP-CI9IWI14 115 127 125	CI91W130	CI91W130	C191W130	Scenario I raining / Clinical Rotation Clinical LP's
	Scenario Training / Clinical Potnical Octavia	V.224 STX LP's C191W114,115,126,127,22	34 STX LP's C191W114,115,126,127,224	4 STX LP's C191W114.115.126.127.224	STVI Pic Cigiwiti iti iti si se instanti
1130 - 1220	C191W130	F S SCENATIO LEAINING / Clinical Rotation Clinical LP's ICTOLWI30	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LPs
0771 - 0.011	STX LP's C191W114,115,126,121	224 STX LP's C191W114 115 126 127 224	C191W130	C191W130	C191W130
1230-1320	Lunch	I unch	21/21/21/12/12/22/	4 STX LP's C191W114,115,126,127,224	STX LP's C191W114,115,126,127,224
	Scenario Training / Clinical Rotation Clinical Li	vs Scenario Training / Clinical Rotation Clinical I Die	Connecto Training / Ali 1 B	Lunch	Lunch
1330 - 1420	C191W130	C191W130	Scenario I faining / Clinical Kotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
	STX LP's C191W114,115,126,127	224 STX LP's C191W114 115 126 127 224	CUSTW130 4 STV1 Pre C101W114 115 155 157 554	CI91W130	C191W130
	Scenario Training / Clinical Rotation Clinical Li	vs Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Pointion Clinical 1 Pt	5 X LPS C191W114,115,126,127,224	STX LP's C191W114,115,126,127,224
1430 - 1520	C191W130	C191W130	C191W130	Cronwright Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
	Same T. T. LP's C191 W114, 115, 126, 127	.224 STX LP's C191W114,115,126,127,224	4 STX LP's C191W114,115,126,127,224	STX LP's C191W114 115 126 127 224	CI91W130
	Scenario Training / Unnical Kotation Chnical Li	vs Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical I Pic	State of Clinical Dataion. Clinical Dataion. Clinical 1 No.
1530 - 1620	STX LP's C191W114 115 126 127	C191W130 STV 1 B4 C101111 11 11 11 11 11 11 11 11 11 11 11	C191W130	C191W130	COMPANY 1140000 CURICAL NOTATION CURICAL LPS
	Scenario Training / Clinical Rotation Clinical LP	's Scenario Training / Clinical Detation Clinical Ltv.	STX LP's C191W114,115,126,127,224	STX LP's C191W114,115,126,127,224	STX LP's C191W114.115.126.127.224
1630 - 1720	CI91W130	C191W130	Scenario Training/ Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's	Scenario Training / Clinical Rotation Clinical LP's
1800 1850	STX LP's C191W114,115,126,127	224 STX LP's C191W114,115,126,127,224,	STX LPis C191W1151 STX 1. Strain 11 Strain 12	CI91W130	C191W130
0021-0021	Dinner	Dinner	Dinner	Dimensional Science 127,224	STX LP's C191W114,115,126,127,224
1900-2200				Unaer	Dinner
10 T T					
W cck 1/	SATURDAY	SUNDAY			
Hours	10 February 2007	11 February 2007	_		
0730.0030	Provident Cycle 118	Cycle 119			
0000 0000	Dreaklast	Breakfast	Location-STY TDN	VC ADFA	
0201-0560		Spiritual Fitness	NI VIGUINING		
1130-1150		Spiritual Fitness			
1200-1400	Lunch	Spiritual Fitness			
1400-1450		Lunco			
1500-1550					
1600-1650					
1700-1720		-			

				_	-	_	-	-	_	-	_	_			-	_	_	_	_																						
	FRIDAY 16 Eabruary 2007		ay 74 Cycle 124				Respect	Indext.		C191W120/121/122/225		C191W120/121/122/225		C191W120/121/122/225		C191W120/121/122/225	C77/771/171/071 M1610			C191W120/121/122/225		C191W120/121/122/225		C191W120/121/122/225		C191W120/121/122/225															
	HURSDAY	-	Cycle 123 Da			Breakfast	Value Trug	FTX	200/201/101/001/M101/J	CT27/271/171/071 M1210		C191W120/121/122/225	LIX	C191W120/121/122/225	FTX	C191W120/121/122/225	Lunch	FTX		C191W120/121/122/225		C191W120/121/122/225	FTX	C191W120/121/122/225	FTX	C191W120/121/122/225	Dinner						S								
			Day 73			Breakfast	Value Trng: Respect	FTX		FTX		FTX		24.2	×1.7		Lunch	FTX		FTX		ETV	×13		FLX		Dinner						NG AREA & SMT								
NESDAV	Tuary 2007		Cycle[122					PREP		PREP		DDED	rker		PREP			PREP		PREP			PREP		PREP								Location:STX TR								
WED	14 Fah	Parent P	Day 1/2			Breaktast	Value Trng: Respect	FTX		FTX		ETV	V11		XLI		Lunch	FTX		FTV			XII		FTX		Unner														
TUESDAY	13 Fehruary 2007	11 Cvola 131	171 51315		010		ury contraction of the contracti	ing / Clinical Kotation Clinical LP's	STX LPs C191W114,115,126,127,224	ing / Clinical Rotation Clinical LP's	STX 1.Ps C191W114 115 126 127 224	ing / Clinical Rotation Clinical LP's	kee een sei stit kiriwiigt Die 1.X.TX	ing / Clinical Rotation Clinical LP's		5LX LPs C191W114,115,126,127,224		ing / Clinical Kotation Clinical LPs	5TX LP's C191W114,115,126,127,224	ng / Clinical Rotation Clinical LP's	TX LP's C191W114 115 136 137 334	ng / Clinical Rotation Clinical LP's	TV 1 Pie C1010/11/11/11/11/2012	ng / Clinical Rotation Clinical I Pe		1 V UF S C 191 W 114, 115, 126, 127, 224			ATTEND OF THE OWNER	ANDAY	IX February 2007	C)CIE 170	C191E120/121/122/225	C191E120/121/122/225	C191E120/121/122/225		C191E120/121/122/225	C191E120/121/122/225	C191E120/121/122/25	C77/771/171/071711/10	
		Dav		Darronal Lini	Breakfact	Value Terrer D	Connerio Tenia	C191W130	24	Scenario Train	24	Scenario Train	CI91W130	Scenario Train	C191W130	I mak	Scenario Train	C191W130	4	Scenario Traini	4 CI71W130	Scenario Traini	CI91W130	Scenario Traini	C191W130	Dinner						Breakfact	FTX	FTX	FTX	Lunch	FTX	FTX FFTV	IFTX	Dinner	
MONDAY	12 Fehriary 2007	Cycle 120					Clinical Rotation Clinical 1 Die		LFS C191 W 114, 115, 126, 127, 2	Clinical Kotation Clinical LP's	LP's C191W114,115,126,127,2	Clinical Rotation Clinical LP's	LP's C191W114,115,126,127,2	Clinical Rotation Clinical LP's	T D's C1011/11/11/11/1010	2,121,021,011,411,01121,02121,2	Clinical Rotation Clinical I Pic		LP's C191W114,115,126,127,22	Clinical Rotation Clinical LP's	LP's C191W114,115,126,127,22	Clinical Rotation Clinical LP's	LP's C191W114.115.126.127.22	Clinical Rotation Clinical LP's	CFS C191W114 115 126 127 22				SATURDAV	February 2007	Cycle 125		C191E120/121/122/225	C191E120/121/122/225	C191E120/121/122/225		CI91E120/121/122/225	C191E120/121/122/225 C191E120/121/122/225	C191E120/121/122/225		
		Day 70		Personal Hygiene	Breakfast	Value Trng: Duty	Scenario Training	C191W130	Connerio Training	C191W130	STX	Scenario Training /	STX	Scenario Training/	C191W130 STX	Lunch	Scenario Training /	C191W130	STX	C191W130 C191W130	STX	Scenario Training /	CI9IW130 STX1	Scenario Training /	C191W130 STX1	Dinner				17		Breakfast	FTX	FTX	XL	Lunch	71X	FTX	FTX	Dinner	
I WEEK I	51001		0515 - 0630	0630 - 0700	0700-0750	00750-0830		0830 - 0920		0930 - 1020			1030 - 1120		1130 - 1220	1230-1320		1330 - 1420		0031 0011	0701-0011		1530 - 1620		1630 - 1720	1800-1850	1000 0001	0077-0061	Week 17	Hours		0730-0930	0930-1020	1030-1120	0011-0011	1200-1400	1400-1450	1600-1650	1700-1720	1730-1930	1930-2130

1001	THIBSDAV	2) Fehruary 2007	A FEBRUARY 2007	Day 78 Cycle 130 Day 79 Cycle 131			Breakfast	Value True Desset	CTV Value Ting, Kespect	EOC Counseling / Critique #3	25	FTX FTX W12012111221225	NKEM1 Recertification Requirements	25 C191W120/121/122/225	FTX AT/FP Brief (OCONUS) / Classroom Clean Up		FTX CI91W120/121/122/255		C191W120/121/122/225	Lunch	FTX Student Assess Questionaire (100)		PTX C191W120/121/122/225 SFC Minard	GRADUATION PRACTICE	C191W120/121/122/225	FTX GRADUATION PRACTICE	C191W120/121/122/225	FTX GRADUATION PRACTICE		Dinner (191W120/121/122/25	Duner					č Company Area								
TTALION, CENTER AND SCHO	WEDNESDAY	21 February 2007	David?	CALIFICATION CONTRACTOR			cfast	Trng Respect			C191W120/121/122/225			C191W120/121/122/225		C191W120/121/122055	C 27/27 1 1 2 1 /02 1 41 1 4 1 7 1 0		C191W120/121/122/225			355/261/161/061/M101/J			C191W120/121/122/225	1	C191W120/121/122/225	<u><u> </u></u>	C191W120/0C19182							Location: SMTS & (
23ZD MEDICAL BA	TUESDAY	20 Fehruary 2007	Day 76 Cvelo 128			ersonal Hygiene	Brea	Value Trng. Duty	FTX		C191W120/121/122/225	FTX		C191W120/121/122/225	X14	C191W120/121/122/225	FTX	C191W101/061/161/061	unch 1.1.2.1.1.2.1.2.1.2.1.2.1.2.1.2.1.2.2.2.2.2	TX Date of the second s	F1X	C191W120/121/122/225	FTX		X C191W120/121/122/225	L1A	X C191W120/121/122/225	FIX	C191W120/121/122/225	aner Dinner		AT MALIS		75 Fehruary 2007	akfast Cycle 133	nual Fitness	ritual Fitness	ritual Fitness		ICD	e	6	5	5
	MONDAY		Day 75 Cycle 127		Personal Hvaiene	Breakfast	Value Trne: Duty	FTX			FTX C191 W120/121/122/225	4	C191W120/121/122/235	FTX		ETV C191W120/121/122/225		C191W120/121/122/225	Lunch	FTX		C191W120/121/122/225	FT	C191W120/121/12025	FTX		TX C191W120/121/122/255		C191W120/121/122/225	Din		SATURDAY	24 February 2007	Cycle 132	reakfast	Spiri	Spiri	Spiri	unch					
West 17	Hours			0515 - 0630	0630 - 0700	0700-0750	00750-0830			0830 - 0920		0620 - 10200	0701 - 0660		1030 - 1120		1130 - 1220		1230-1320		1330 - 1420		0001 0001	0761 - 0641		1530 - 1620	4	1630 - 1720	1800-1850		1900-2200	Week 17	Hours		0730-0930 B	0930-1020	1030-1120	0011-0011	I Z00-1400 IL		1400-1450	1400-1450 1500-1550 1600-1650	1400-1450 1500-1550 1600-1650 1700-1720	1400-1450 1500-1550 1600-1650 1700-1720 1730-1930 Di

UNIFORM: BDUs (unless otherwise indicated)

Week 17	NOM	VDAY	TUES	DAV	WEDA	VESDAV	THU	SHAV		EDIDAV
Hours	76 Fehru	TIND VINT	27 Februar	200c A	28 Eabr	menu 2007	1 Marco	TOUR SE		March 2007
										March /007
	Day 80	Cycle 134	Day 81	Cycle 135	Dav	Cycle	Dav	Cycle	Dav	Cvele
0515 - 0630									1	
0630 - 0700	Personal Hygione		Personal Hygiene							
0700-0750	Breakfast		Breakfast		Breakfast		Breakfast		Rreakfact	
00750-0830	Value Time: Duty		Value Trace Date		Value True Demont		Volue Turn Decouct		Meluo Tener Docusot	
0260 - 0530	CIFT	urn-In	Graduation (0900)		111111		A ALAC A LINK. NEW JOINT			
0501 - 0560	CIFTA	ու-ուղ	Graduation (0900)							
011-0201	CFT	brn-In	Graduation (0900)							
0711 - 0601										
1130 - 1220	CIFT	urn-In	Graduation (0900)		а - - -					
1230-1320	Lunch		11 mets		Innah		Turnet		+1	
	GRADUATIO:	N PRACTICE	(1330) Cadre Sensing Sessi	uo			TUNU		LUDCD	
1330 - 1420				BN CDR / Ceurse Dir						
	GRADUATION	N PRACTICE								
1430 - 1520										
	GRADUATION	N PRACTICE								
1530 - 1620										
	GRADUATION	N PRACTICE								
1630 - 1720									-1	
1800-1850	Dinner		Dinner		Dinter		Dinner		Dinner	
1900-2200	Graduation D	Jinner (1800)								
11 July 10										
Hours	SAIUS	AMAY	SUND							
		Cycle		Cvele						
0730-0930	Breakfast		Breakfast			Location' Compa	IV Area			
0930-1020			Spiritual Fitness			induity of the second				
1030-1120			Spiritual Fitness							
1200-1400	Innch		Spintual Fitness							
1400-1450			1300							
1500-1550				_						
1600-1650										
1730-1030	Dinner		Distant							
1020 2150										

215

VITA

Name:	Yvette Woods	
Address:	264 th Medical Battalion, Company C, U.S. Army Medical Department Center and School, Ft. Sam Houston, TX 78234	1
Email Address:	yvette.woods@us.army.mil	
Education:	 Ph.D., Educational Human Resource Development, Texas A. University, 2007 M.S., Administration, Central Michigan University, 1993 B.S., Occupational Therapy, Chicago State University, 1991 	&M
Area of Clinical Specialty:	Occupational Therapist Neuromusculoskeletal Evaluator Army Medical Department Instructor Certified Strength Training Specialist	
Experiences:	Staff Officer, Army Medical Specialist Corps, Ft. Sam House TX	ton,
	Assistant to the Chief, Army Medical Specialist Corps, Ft. Sa Houston, TX	am
	Chief, Occupational Therapy, Evans Army Community Hosp Ft. Carson, CO	pital
	Assistant Chief, Occupational Therapy, Brooke Army Medic Center, Ft. Sam Houston, TX	al
	Chief, Occupational Therapy, 121 General Hospital Yongsan, Korea	
	Staff, Occupational Therapist, Walter Reed Army Medical C Washington, DC	lenter
Professional Affiliation:	Kappa Delta Pi International Honor Society in Education 2	2005
	38th Parallel Society1American Occupational Therapy Association1	1994 1991